FORMATION AND CHARACTERIZATION OF RARE EARTH OXIDES AS HIGH-K GATE DIELECTRICS ON GERMANIUM SUBSTRATE

TAHSIN AHMED MOZAFFOR ONIK

FACULTY OF ENGINEERING UNIVERSITY OF MALAYA KUALA LUMPUR

FORMATION AND CHARACTERIZATION OF RARE EARTH OXIDES AS HIGH-K GATE DIELECTRICS ON GERMANIUM SUBSTRATE

TAHSIN AHMED MOZAFFOR ONIK

THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

FACULTY OF ENGINEERING UNIVERSITY OF MALAYA KUALA LUMPUR

UNIVERSITY OF MALAYA ORIGINAL LITERARY WORK DECLARATION

Name of Candidate: Tahsin Ahmed Mozaffor Onik

Matric No: 17050736/1 | KVA170057

Name of Degree: Doctor of Philosophy

Title of Project Paper/Research Report/Dissertation/Thesis ("this Work"):

FORMATION AND CHARACTERIZATION OF RARE EARTH OXIDES AS HIGH-K GATE DIELECTRICS ON GERMANIUM SUBSTRATE

Field of Study: Advance Materials / Nanomaterials (Materials Engineering)

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every rights in the copyright to this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature

Date: 08 Sept 2022

Subscribed and solemnly declared before,

Witness's Signature

Date: 8 Sept 2022

Name:

Designation:

FORMATION AND CHARACTERIZATION OF RARE EARTH OXIDES AS HIGH-K GATE DIELECTRICS ON GERMANIUM SUBSTRATE ABSTRACT

In this study, Sm₂O₃/Ge and Ho₂O₃/Ge stack based metal oxide semiconductor (MOS) capacitors were prepared from radio frequency (RF) sputtered metallic Sm on Ge substrate and metallic Ho on Ge substrate followed by thermal oxidation/nitridation in N₂O ambient. The effects of several oxidation/nitridation temperatures, i.e., 300 °C - 600 °C on the sputtered Sm/Ge and influence of various oxidation/nitridation durations, i.e., 5 -20 min on both sputtered Sm/Ge and Ho/Ge system have been comprehensively investigated. The film crystallinity and phase identification, chemical composition, interface chemical bonding states stability and structural morphology were characterized from Xray diffraction (XRD), Raman spectroscopy, X-ray photoelectron spectroscopy (XPS) and High-resolution transmission electron microscopy (HRTEM). Formation of stoichiometric trigonal-Sm₂O₃ dielectric interface has been verified for thermally oxidized/nitrided sputtered Sm/Ge system. In contrary, sub-stoichiometric cubic-Ho₂O₃ has been identified for thermally oxidized/nitrided sputtered Ho/Ge system. Both the stacking structure produced sandwiched interfacial layer containing asymmetrically distributed Ge-O, and Ge-N contents while oxidized/nitrided Sm/Ge stack produced an additional stable interfacial germanate (Sm-O-Ge). Suppression of GeO (g) volatilization was evident from the XPS analysis due to GeO (g) trapping mechanism instigated by interfacial germanate (Sm-O-Ge) formation for oxidized/nitrided Sm/Ge while IL rich with Ge₃N₄ contributed to inhibit the volatilization of GeO (g) for oxidized/nitrided Ho/Ge system. The HRTEM analysis also validated double stack amorphous interfaces obtaining physical oxide thickness, $t_{ox} \sim 4.25 - 6.91$ nm for oxidized/nitrided Sm/Ge and $t_{ox} \sim 8.22 - 6.91$ 9.78 nm were obtained for oxidized/nitrided Ho/Ge system. Thermal oxidation/nitridation

model related to the growth mechanism of Sm₂O₃/IL and Ho₂O₃/IL as high-k gate dielectrics have been suggested after considering the findings of XRD, XPS and HRTEM. Band alignment mapping and electrical measurements revealed that oxidized/nitrided Sm/Ge at 400 °C for 15 min exhibited symmetrical band offsets distribution comprising highest conduction band offset (CBO), ΔE_c of 2.60 eV and valence band offset (VBO), ΔE_v of 2.98 eV leading to lowest leakage current density $J_g \sim 8.38 \times 10^{-6}$ A cm⁻² at enhanced breakdown field, E_{BD} of 13.31 MV cm⁻¹. As for Ho/Ge system the sample oxidized/nitrided at 400°C for 10 min exhibited the maximum $\Delta E_c \sim 2.47$ eV and $\Delta E_v \sim 4.67$ eV inducing lowest $J_g \sim 10^{-5}$ A cm⁻² at highest $E_{BD} \sim 8.59$ MV cm⁻¹. The electrical results of these samples also revealed higher dielectric constant, k of 31.19 and 13.60 for Sm₂O₃/IL/Ge and Ho₂O₃/IL/Ge gate stacks, respectively. However, both the gate stacks expensed high average interface trap density, $D_{it} \sim 10^{13} \text{ eV}^{-1} \text{cm}^{-2}$. Additionally, effective oxide charge, slow trap density, hysteresis was also estimated from C-V curve which eventually ascribed the variance of electrical breakdown field for the investigated gate stacks. Comparing the physical and electrical properties of oxidized/nitrided sputtered Sm/Ge and Ho/Ge system it has been anticipated Sm₂O₃ could be more convenient choice for high-k gate dielectric oxide for Ge based MOS devices.

Keywords: rare-earth, high-k, germanium, thermal oxidation/nitridation, MOS capacitor.

PEMBENTUKAN DAN PENCIRIAN OKSIDA NADIR BUMI SEBAGAI DIEL-EKTRIK GET-K TINGGI PADA SUBSTRAT GERMANIUM ABSTRAK

Dalam kajian ini, kapasitor semikonduktor oksida logam (MOS) berasaskan struktur Sm₂O₃/Ge dan Ho₂O₃/Ge telah disediakan menggunakan teknik percitan frekuensi radio (RF) Sm di atas substrat Ge dan Ho di atas substrat Ge, diikuti dengan pengoksidaan/penitridaan terma dalam ambien N₂O. Kesan suhu pengoksidaan/penitridaan, iaitu, 300 °C – 600 °C pada Sm/Ge dan pengaruh pelbagai tempoh pengoksidaan/nitridasi, iaitu 5 - 20 min pada kedua-dua sistem Sm/Ge dan Ho/Ge dikaji secara menyeluruh. Penghabluran filem dan pengenalpastian fasa, komposisi kimia, ikatan kimia antaramuka (IL) yang menyatakan kestabilan dan morfologi struktur telah dikaji dan dicirikan menggunakan pembelauan sinar-X (XRD), spektroskopi Raman, spektroskopi fotoelektron sinar-X (XPS) dan mikroskop elektron penghantaran resolusi tinggi (HRTEM). Pembentukan antaramuka dielektrik trigonal-Sm₂O₃ stoikiometri telah disahkan dalam sistem struktur Sm/Ge teroksida/ternitrida. Sebaliknya, sub-stoikiometrik kubik-Ho₂O₃ telah dikenal pasti untuk sistem struktur Ho/Ge teroksida/ternitrida. Kedua-dua struktur menghasilkan lapisan antaramuka yang mengandungi kandungan Ge-O dan Ge-N yang teragih tidak simetri manakala struktur Sm/Ge teroksida/ternitrida menghasilkan percambahan antaramuka yang stabil (Sm-O-Ge). Penindasan pemeruapan GeO (g) terbukti daripada analisis XPS disebabkan oleh mekanisme perangkap GeO (g) yang dicetuskan oleh pembentukan percambahan antaramuka (Sm-O-Ge) untuk Sm/Ge teroksida/ternitrida manakala antaramuka yang kaya dengan Ge₃N₄ menyumbang untuk menghalang pemeruapan. daripada GeO (g) untuk sistem Ho/Ge teroksida/nitrid. Analisis HRTEM juga mengesahkan antaramuka bersifat amorfus mempunyai ketebalan oksida fizikal, tox $\sim 4.25-6.91$ nm bagi struktur Sm/Ge teroksida/ternitrida dan $t_{ox} \sim 8.22-9.78$ nm bagi struktur Ho/Ge teroksida/ternitrida. Model pengoksidaan/nitridasi terma yang

berkaitan dengan mekanisme pertumbuhan Sm₂O₃/IL dan Ho₂O₃/IL sebagai dielektrik get-k tinggi telah dicadangkan selepas mempertimbangkan analisis XRD, XPS dan HRTEM. Pemetaan penjajaran jalur dan pengukuran elektrik mendedahkan bahawa Sm/Ge pengoksidaan/nitridasi pada 400 °C selama 15 minit mempamerkan taburan offset jalur simetri yang terdiri daripada offset jalur konduksi tertinggi (CBO), ΔE_c daripada 2.60 eV dan offset jalur valens (VBO), ΔE_{ν} sebanyak 2.98 eV membawa kepada ketumpatan arus bocor terendah $J_g \sim 8.38 \times 10^{-6}$ A cm⁻² dan medan pecah tebat dipertingkatkan, E_{BD} sebanyak 13.31 MV cm⁻¹. Bagi struktur Ho/Ge teroksida/ternitrida pada 400 °C selama 10 minit menunjukkan maksimum $\Delta E_c \sim 2.47$ eV dan $\Delta E_v \sim 4.67$ eV mendorong $J_g \sim 10^{-5}$ A cm⁻² terendah pada $E_{BD} \sim 8.59$ MV cm⁻¹ yang tertinggi. Keputusan elektrik sampel ini juga menunjukkan pemalar dielektrik yang lebih tinggi, k sebanyak 31.19 dan 13.60 untuk struktur Sm2O3/IL/Ge dan Ho2O3/IL/Ge, masing-masing. Walaubagaimanapun, kedua-dua struktur mempamerkan ketumpatan perangkap antara muka yang tinggi, $D_{it} \sim 10^{13} \text{ eV}^{-1} \text{cm}^{-2}$. Selain itu, caj oksida, ketumpatan perangkap perlahan, histerisis juga dianggarkan daripada lengkung C-V dan dihubungkait dengan varians medan pecah tebat elektrik. Perbandingan sifat fizikal dan elektrik di antara sistem Sm/Ge dan Ho/Ge teroksida/ternitrida telah dilakukan. Sm₂O₃ boleh menjadi pilihan yang lebih baik untuk oksida dielektrik get tinggi untuk peranti MOS berasaskan Ge.

Kata kunci: nadir bumi, high-*k*, germanium, pengoksidaan/penitridaan terma, kapasitor MOS

ACKNOWLEDGEMENTS

Firstly, I would thank to Almighty Allah SWT for providing me the strength, potency, and health to carry out my research works. Then, I would like to convey my deepest gratitude to my respectful chief supervisor Assoc. Prof. Ir. Wong Yew Hoong for his continuous motivation, encouragement, and guidance throughout my research journey. He has been very gracious and enthusiastic in providing useful comments and advises while constantly tracking of my progress to ensure my graduation on time.

My grateful thanks also go to my respectful co-supervisors Assoc. Prof. Ir. Dr. Mohd Faizul Bin Mohd Sabri and Ir. Dr. Huzein Fahmi bin Hawari for their valuable assistance and support on time. Special thanks to Mrs. Hartini Bahrun and all the academic and administrative staffs from faculty of engineering for their assistances. Besides, I would like to acknowledge the contribution of all the staffs and technicians associated in this research. Special acknowledgement for the involved technicians from Malaysian Institute of Microelectronic System (MIMOS) for providing smooth services during physical characterizations. Furthermore, I would thank all my fellow colleagues specially Alex, Kazeem and Zhen Ce who always helped me in various prospects.

I am deeply indebted to my parents (Mr. Mohammed Sagir Hossain Chowdhury and Mrs. Sayeda Rabeya Begum) and my brothers (Afridul Ifaj and Tasbiul Akik) for their love, prayer, inspiration, moral support to shape me into a better person. A great and special appreciation to my beloved wife Tasin Parven Tasika for her understanding and patience. She always stayed next to me in every thick and thin, while sharing my sleepless nights and all hard times throughout my PhD journey during stay in Malaysia.

Last but not least I would like to acknowledge the financial support by University of Malaya via Faculty Research Grant (GPF017A-2018) and Southeast Asia – Taiwan Universities (SATU) Joint Research Scheme (ST016-2020).

TABLE OF CONTENTS

Abs	tract	ii	i
Abst	rak		1
Ackı	nowledg	ementsvi	i
Tabl	e of Cor	ntentsvii	i
List	of Figur	esxi	i
List	of Table	esxvii	i
List	of Symb	ools and Abbreviationsxix	ζ
CHA	APTER	1: INTRODUCTION1	l
1.1	Backg	round of Study1	l
1.2	Proble	m Statement	3
1.3	Object	ive of the Research	3
1.4	Scope	of Study9)
1.5	Structu	are of Thesis)
CHA	APTER	2: LITERATURE REVIEW10)
2.1	Introdu	action)
2.2	Moore	's Law and MOS Scaling Trend11	l
2.3	Basic I	Principle of MOS and MOSFET13	3
2.4	Limita	tions of SiO ₂ /Si Scaling15	5
2.5	Requir	rements for Implementing High-k Gate Oxide16	5
2.6	Ge as A	Alternative Substrate Material21	l
	2.6.1	Issues in growth of Native GeO ₂ 22	2
	2.6.2	Thermal Issues of GeO ₂ /Ge Stack	1
	2.6.3	Electrical Characteristics of GeO ₂	5

2.7	Comm	oon High- <i>k</i> Oxides on Ge2	.8
2.8	Rare e	arth oxide as Alternative High-k	0
2.9	Comm	non Deposition Methods of REOs	6
	2.9.1	Chemical Vapor Deposition (CVD)	6
	2.9.2	Physical Vapor Deposition (PVD)	8
2.1	0 Perfor	mance of Commonly Used REOs/Ge Stack4	1
	2.10.1	La ₂ O ₃ /Ge Gate Stack	-2
	2.10.2	Y ₂ O ₃ /Ge Gate Stack	.3
	2.10.3	CeO ₂ /Ge Gate Stack4	4
2.1	$1 \text{ Sm}_2\text{O}_3$	and Ho ₂ O ₃ as Alternative Dielectric4	.5
2.1	2 Role c	of Nitrogen (N) for Ge passivation5	1
СН	APTER	3: RESEARCH METHODOLOGY5	3
3.1	Resear	rch Overview5	3
3.2	Requi	red Materials5	6
	3.2.1	Dimension of Substrate Material5	6
	3.2.2	Chemical Required for Substrate and Quartz Tube Cleaning	6
	3.2.3	Chemical, Gas and Materials for Sputtering Process	6
	3.2.4	Required Gases for Sputtering Process5	7
3.3	Experi	imental Procedure5	8
	3.3.1	Ge substrate Dip cleaning5	8
	3.3.2	Sputtering of Rare Earth Metal on Ge substrate5	8
	3.3.3	Thermal Oxidation and Nitridation Process5	9
	3.3.4	MOS Capacitor Test Structure	1
3.4	Chara	cterization Techniques6	2
	3.4.1	Xray Diffraction (XRD) Analysis6	2

3.4.2 X-ray Photoelectron Spectrometer (XPS) Analysis65

	3.4.3	RAMAN Analysis	66
	3.4.4	HRTEM Analysis	67
	3.4.5	I-V Measurement	68
	3.4.6	C-V Measurement	68
СН	APTER	4: RESULTS AND DISCUSSION	69
4.1	Introdu	uction	69
4.2	Effect	of Various Thermal Oxidation/Nitridation Temperatures on	Sputtered
	Sm/Ge	e in N ₂ O Ambient	70
	4.2.1	XRD Analysis	70
	4.2.2	J-E Characteristics	76
	4.2.3	XPS Analysis	78
	4.2.4	Band Alignment	
	4.2.5	HRTEM Analysis	
	4.2.6	Oxidation and Nitridation Mechanism Model	90
	4.2.7	Capacitance-Voltage (C-V) Characteristics	93
4.3	Influer	nce of Various Thermal Oxidation/nitridation Durations on Sputter	ed Sm/Ge
	in N ₂ C	D ambient	101
	4.3.1	XRD Analysis	101
	4.3.2	Raman Analysis	104
	4.3.3	XPS Analysis	
	4.3.4	Band Alignment	115
	4.3.5	HRTEM Analysis	118
	4.3.6	C-V Characteristics	121
	4.3.7	J-E Characteristics	132
4.4	Influer	nce of Various Thermal Oxidation/Nitridation Durations on Sputter	ed Ho/Ge
	in N ₂ O	O Ambient	135

	4.4.1	XRD Analysis135
	4.4.2	J-E Characteristics
	4.4.3	XPS Analysis142
	4.4.4	Band Alignment
	4.4.5	HRTEM Analysis
	4.4.6	Oxidation/Nitridation Mechanism Model157
	4.4.7	C-V Characteristics
4.5	Compa	rison Between Oxidized/Nitrided Sputtered Sm/Ge and Ho/Ge Stack 168
	4.5.1	Comparison of Physical Properties168
	4.5.2	Comparison of Electrical Properties171
CHA	APTER	5: CONCLUSION AND RECOMMENDATIONS
5.1	Novelt	y and Significance of the Findings173
	5.1.1	Effect of Various Thermal Oxidation/Nitridation Temperature on Sputtered
		Sm Thin Film Based on Ge Substrate175
	5.1.2	Effect of Various Thermal Oxidation/Nitridation Durations on Sputtered
		Sm Film Based on Ge Substrate176
	5.1.3	Effect of Various Thermal Oxidation/Nitridation Durations on Sputtered
		Ho Film Based on Ge Substrate
	5.1.4	Comparison Between Thermally Oxidized/Nitrided Sputtered Sm/Ge and
		Ho/Ge
5.2	Recom	mendation for Future Research179
Refe	erences	
List	of Public	cations and Papers Presented202

LIST OF FIGURES

Figure 2.1: Evolution of transistor gate length from 1970 to 2030 (Radamson, H. & Thylen, 2014)12
Figure 2.2: (a) Typical MOS capacitor stacking arrangement (b) Conventional structure of a MOSFET
Figure 2.3: TEM image of SiO ₂ /Si and HfO ₂ /Si gate stack thickness (Robertson & Wallace, 2015)
Figure 2.4: Advantage of using high- <i>k</i> as gate material to reduce leakage current18
Figure 2.5: Trend of bandgap of high- <i>k</i> dielectrics with respect to dielectric constant (Wilk et al., 2001)
Figure 2.6: Comparison of native oxide Growth rate for Si and Ge substrate (Sahari et al., 2013)
Figure 2.7: Schematic of GeO desorption mechanism from GeO ₂ /Ge interface (S. K. Wang et al., 2010)25
Figure 2.8: Bidirectional $C-V$ curve at (1 MHz) of sputtered GeO ₂ on Ge, Si, and thermally oxidized Si substrates followed by post deposition annealing 400 °C (Kita et al., 2008)
Figure 2.9: Band alignment schematic of GeO ₂ /Ge interface and comparison with the band offset of SiO ₂ /Si interface. (L. Lin et al., 2010)27
Figure 2.10: Candidates for metal oxide high-k gate insulator (Iwai et al., 2002)
Figure 2.11: Dielectric constants of rare-earth oxides (Chin & Cheong, 2011)
Figure 2.12: Band gaps of rare-earth oxide (Goh et al., 2017a)
Figure 2.13: Conduction band offsets (CBO) of various REOs on Si as a function dielectric constant (Chin & Cheong, 2011)
Figure 2.14: The electronegativity and lattice energy of lanthanide REOs (Goh et al., 2017a)
Figure 2.15: Effect of surface roughness of rare-earth oxides on leakage current (Kakushima et al., 2006)
Figure 2.16: Comparison of interface trap densities among transitional metal and rare earth metal deposited on GeO ₂ /Ge interface. (Lu et al., 2014)42

Figure 2.17: Review of leakage current density against electrical breakdown voltage for Sm ₂ O ₃ /Si gate stack for different deposition techniques (Goh et al., 2017a)
Figure 2.18: <i>C-V</i> curve for Ho ₂ O ₃ /Si gate stack for diiferent annealed temperatures (Pan, Chang, et al., 2010)
Figure 2.19: Leakage current density $(J-V)$ with respect to gate voltage for Ho ₂ O ₃ / p type Si gate stack annealed at different RTA temperatures (Pan, Chang, et al., 2010)51
Figure 3.1: Experimental design and variables for investigation
Figure 3.2: Sequential Steps for Sample Preparation
Figure 3.3: Overview of characterization techniques
Figure 3.4: Schematic of PVD RF sputtering system
Figure 3.5: Schematic of horizontal furnace tube setup for thermal oxidation and nitridation
Figure 3.6: Al Gate electrode deposited in top of REOs/Ge
Figure 4.1: XRD patterns of the Sm based gate stack at various oxidation/nitridation temperatures of 300 °C, 400 °C, 500 °C and 600 °C73
Figure 4.2: Intensities of Sm_2O_3 , GeO_2 and Ge_3N_4 at various oxidation/nitridation temperatures (300 – 600) °C73
Figure 4.3: W–H linear regression plot of a. Sm_2O_3 , b. GeO_2 at regression co-efficient $r^2 = 1$ for various oxidation/nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C).
Figure 4.4: Micro-strain and crystallites of Sm ₂ O ₃ and GeO ₂ at various oxidation and nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C)76
Figure 4.5: <i>J-E</i> characteristics of Sm ₂ O ₃ /Ge gate stack at different oxidation and nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C)77
Figure 4.6: Sm 3d narrow scan deconvoluted spectra for various oxidation and nitridation temperatures (300 °C, 400 °C, and 500 °C)
Figure 4.7: XPS Sm 3d narrow scan peak position of various investigated temperatures (300 °C, 400 °C and 500 °C)
Figure 4.8: Deconvoluted Ge 3d narrow scan spectra of oxidized/nitrided sputtered

Sm/Ge for various oxidation / nitridation temperatures (300 °C, 400 °C, and 500 °C). 83

Figure 4.15: Slow trap density and hysteresis voltage as function for $Sm_2O_3/IL/Ge$ gate stack at different oxidation/nitridation temperature (300 – 500) °C......96

Figure	4.23:	Deconvoluted	core	level	narrow	scan	spectrums	of	Ols	for	different
oxidati	on/nitr	idation duration	ns (5 -	- 20) n	nin			••••			114

Figure 4.24: Degree of oxidation/nitridation level as function of different durations..114

Figure 4.25: Estimation of valance band offset (ΔE_v) and energy bandgap (E_g) for Sm ₂ O ₃ /IL/Ge structure at various oxidation/nitridation durations (5 – 20) min (a) Valance band edge, E_v and VBO extraction from XPS valence band spectra. (b) Energy band gap deduction from oxygen plasmon loss XPS spectra
Figure 4.26: Band Alignment mapping of $(Sm_2O_3 + IL)$ with respect to Ge substrate for various oxidation/nitridation durations $(5-20)$ min
Figure 4.27: HRTEM cross-sectional morphology for thermally oxidized/nitrided 15 and 20 min samples
Figure 4.28: Linear regression plot of total oxide thickness of $Sm_2O_3/IL/Ge$ as function of various oxidation/nitridation durations (5 – 20) min
Figure 4.29: <i>C-V</i> features recorded at high frequency 1Mhz for $Sm_2O_3/IL/Ge$ based MOS capacitors oxidized/nitrided at different durations (5 – 20) min
Figure 4.30: Effective dielectric constant, k_{eff} and equivalent oxide thickness obtained for Sm ₂ O ₃ /IL/Ge stack with respect to different oxidation/nitridation durations (5 – 20) min
Figure 4.31: Effective oxide charge, Q_{eff} and slow trap density, Q_{it} acquired from <i>C-V</i> plot of Sm ₂ O ₃ /IL/Ge gate stack at different oxidation/nitridation durations (5 – 20) min. 129
Figure 4.32: Hysteresis voltage variations of $Sm_2O_3/IL/Ge$ gate stack for various thermal oxidation/nitridation durations (5 – 20) min
Figure 4.33: Average interface trap density D_{it} approximated with respect to trap energy level ($E_C - E$) for Sm ₂ O ₃ /IL/Ge gate stack developed under different oxidation/nitridation (5 – 20) min. Inset image presenting total interface density as function of different durations
Figure 4.34: <i>J-E</i> features for $Sm_2O_3/IL/Ge$ based MOS capacitors at different oxidation/nitridation durations (5 – 20) min

Figure 4.36: Cubic, c-Ho ₂ O ₃ intensities at various oxidation/nitridation durations (5 min, 10 min, 15 min, and 20 min)
Figure 4.37: Williamson–Hall (W–H) plot of c-Ho ₂ O ₃ for investigated oxidized/nitrided samples at various durations $(5 - 20)$ min
Figure 4.38: Crystallite sizes (triangle symbol, left axis) and micro-strains (circle symbol, right axis) of Ho ₂ O ₃ at various oxidation/nitridation durations (5 min, 10 min, 15 min, and 20 min).
Figure 4.39: <i>J-E</i> curve of various oxidized/nitrided sample at durations (5 min, 10 min, 15 min, and 20 min)
Figure 4.40: Ho 4d narrow scan spectra measured for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations $(5 - 10)$ min
Figure 4.41: Deconvolution of O1s narrow band spectra for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations (5 – 10) min
Figure 4.42: Deconvolution of O1s narrow band spectra for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations $(5 - 10)$ min149
Figure 4.43: Peak position of Ho 4d narrow scan spectra as function of various oxidation/nitridation durations estimating the degree of oxidation level
Figure 4.44: Valance band offset (ΔE_v) and energy bandgap (E_g) Ho ₂ O ₃ /IL/Ge structure at various oxidation/nitridation durations (5 – 20) min (a) Extraction of valance band edge, E_v and VBO estimation from valance band spectra (b) Energy bandgap extraction from O1s plasmon loss XPS spectrum
Figure 4.45: Band alignment schematic $Ho_2O_3/IL/Ge$ stack for various oxidation and nitridation durations (5 – 20) min
Figure 4.46: Structural cross-sectional morphology obtained from HRTEM for oxidized/nitrided sputtered Ho/Ge samples at 10 min and 20 min durations. Inset images shows FFT diffraction
Figure 4.47: Determination of $Ho_2O_3/IL/Ge$ stack physical thickness for investigated samples at various oxidation/nitridation durations (5 – 10) min157
Figure 4.48: Oxidation/nitridation mechanism model for sputtered Ho/Ge after subsequent oxidation/nitridation at different duration 10 and 20 min. Dotted circle represents oxygen vacancies, blue filled circle denotes in-diffusing oxygen and IL is interfacial layer

Figure 4.49: (a) C-V characteristics of Ho₂O₃/IL/Ge MOS capacitors measured at 1 MHz for bidirectionally swept gate bias for various oxidation/ nitridation durations (5 min, 10 min, 15 min, and 20 min). The inset displays effective dielectric constant and accumulation capacitance and effective dielectric constant as function oxidation and nitridation durations.

Figure 4.50: Amount of Effective oxide charge, Q_{eff} and slow trap density, Q_{it} existing in the Ho₂O₃/IL/Ge MOS capacitors for oxidized/nitrided samples at durations (5 – 20) min. 164

Figure 4.52: Average interface trap density D_{it} and total interface density contained in Ho₂O₃/IL/Ge MOS capacitors at different oxidation/nitridation durations (5 – 20) min.

LIST OF TABLES

Table 2.1: CMOS technology node evolution (Sicard & Aziz, 2011). 12
Table 2.2: Electron mobility, μ_e and hole mobility, μ_h of different semiconductor materials (Kamata, 2008)
Table 2.3: Leakage current density and dielectric constant of Sm ₂ O ₃ /Si gate stack for different deposition techniques (S. Jeon et al., 2001; Paivasaari et al., 2005; D. Yang et al., 2003)
Table 3.1: Parameter details of chemical utilized for Ge substrate and quartz tube cleaning
Table 3.2: Parameter details of chemical, gas, materials used during sputtering process.
Table 3.3: Detail specifications for gases used during thermal oxidation/nitridation 57
Table 4.1: Optimum thermal oxidation/nitridation durations for sputtered Sm/Ge and Ho/Ge at constant 400 °C
Table 4.2: Comparison of physical properties between thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge 170
Table 4.3: Comparison of electrical features between thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge. 172

LIST OF SYMBOLS AND ABBREVIATIONS

Permittivity of free space (Fm⁻¹) : \mathcal{E}_{O} Micro-strain ε : Diffraction angle (°) θ : Wavelength (cm^{-1}) λ : Carrier mobility (cm²V⁻¹s⁻¹) : μ Electron mobility (cm²V⁻¹s⁻¹) : μ_e Hole mobility $(cm^2V^{-1}s^{-1})$: μ_h Surface potential (eV) : φ_s Α Gate electrode area (cm²) : Oxide capacitance (µF) C_{ox} : D Crystallite size (nm) : Interface trap density (ev⁻¹ cm⁻¹) D_{it} : Ε Electrical field (MV cm⁻¹) : Electrical breakdown field (MV cm⁻¹) E_{BD} : E_g Energy Band Gap (eV) : Valance band edge (eV) E_{v} : Conduction band offset (CBO) (eV) ΔE_c : Valance band offset (VBO) (eV) ΔE_{v} : Ι Current (A) : Drive current (A) I_D : Leakage current density (A cm⁻¹) J_g : Dielectric constant k : L Channel length (cm) :

- : Oxide thickness (nm) t_{ox} V_{g} Gate voltage (V) : Flatband voltage (V) V_{FB} : W : Channel Width Atomic layer deposition ALD : CVD Chemical vapor deposition : CMOS Complementary metal oxide semiconductor : CET Capacitance equivalent thickness : EOTEquivalent oxide thickness : HRTEM High resolution transmission electron microscopy : IL Interfacial layer : MBD Molecular beam deposition : Metal oxide semiconductor MOS : Metal Oxide semiconductor field effect transistor MOSFET : PLD Pulse Laser Deposition : PVD Physical vapor deposition : Post deposition annealing PDA : RTA Rapid thermal annealing Rare earth oxides REOs ÷ STD : Slow trap density TMOs Transitional metal oxide :
 - XRD : X-Ray diffraction
 - XPS : X-Ray photoelectron spectroscopy

CHAPTER 1: INTRODUCTION

1.1 Background of Study

Complementary Metal oxide Semiconductor (CMOS) technology are progressing very rapidly to generate high performance and more speedy powerful electronic devices including transistors and capacitors. Aggressive development of Silicon (Si)-based complementary metal oxide semiconductor (CMOS) technology has immensely contributed to the semiconductor industry for decades. The advancement has been made according to the Gordon Moore's prediction of scaling where the number of components in an integrated circuit doubles every two years (Mack, 2011; Thompson & Parthasarathy, 2006). This substantial expansion of metal oxide semiconductor (MOS) technology entails ultrathin gate oxide (~ 1.2 nm or 4-5 atomic layers) (Houssa et al., 2006; Robertson & Wallace, 2015). Hence a superior MOS with an ultrathin gate oxide functionality has become the potential demand. Till now SiO₂/Si gate stack has been serving the semiconductor industry due to its excellent insulating properties of native SiO₂ (Frank & Taur, 2002; Park & Kim, 2005; Wilk, Wallace, & Anthony, 2001). However, due to drastic downscaling of SiO₂/Si interface it has reached the potential limit, further downscaling led to large leakage current and degradation of interface quality. The increment of leakage current at an undesirable scale due to extreme reduction of oxide thickness (~ 1 nm) triggered substantial concern regarding reliability and static power dissipation which obstructs the demand of low power consumption devices. (Gordon, Becker, Hausmann, & Suh, 2001; He, Zhu, Sun, Wan, & Zhang, 2011; Robertson & Wallace, 2015). Hence, a physically thicker oxides with similar equivalent oxide thickness (EOT) are required to replace conventional SiO₂. Since for MOS devices the capacitance is directly proportion to dielectric constant, k and inversely proportional to gate oxide thickness, t_{ox} (Goley & Hudait, 2014; Wong & Cheong, 2010). This led the interest toward high-k materials that can provide physically thicker oxide but provides same equivalent thickness as SiO₂. For the last 20 years, an immense progress has taken place on the selection criterion of highk gate dielectrics on Si substrate while exploring their physical and electrical properties for integration of high performance CMOS devices (Chew et al., 2016; Cui, Tuokedaerhan, Cai, & Lu, 2022; Juan. Gao et al., 2016; He et al., 2011; Kaya & Yilmaz, 2015, 2019; J. S. Lee et al., 2014; Robertson, 2004; Shao et al., 2003; Wilk et al., 2001). However, deposition of high-k gate oxides on Si semiconductor are governed by several limitation which includes growth of thick interfacial layer, increment of fixed oxide charges, excessive charge trapping, large interface trap density and degradation of electron mobility. It is well known that for high-k gate dielectric tunneling current increases exponentially with reduction barrier height on the other hand the bandgap maintains an inversely proportional relationship proportional to dielectric constant which limits the choice of high-k material due to low bandgap issue (Wilk et al., 2001). Since, drive current, I_D is proportional to the carrier mobility μ and so superior mobility substrate is required (He et al., 2011). For consideration of high speed operation low hole carrier mobility Si limits device performance and so replacing Si with higher mobility semiconductor is an alternative to deliver high speed MOS applications. Accordingly, Germanium (Ge) has gained widespread interest to replace Si substrate due to its higher carrier mobility than Silicon (Kamata, 2008). Moreover, ease of integration process also added additional advantage towards Germanium. Thus, finding suitable combination of high-k gate oxide on Ge for the improvement of MOS device performance is one of the major scopes of this research.

1.2 Problem Statement

For past few generations, scaling of oxide thickness has been the key route to cater the demand of low power consumption and high-speed devices. In accordance with the current CMOS scaling, it is required to maintain higher capacitance between the integrated gate dielectric and substrate while reducing the oxide film thickness with a smaller gate area. Currently, CMOS technology reached the scaling limit in order of sub-100 nm regime and beyond (Wilk et al., 2001). Rapid downscaling of SiO₂/Si structure has pressed the SiO₂ thickness below 1.2 nm for sub-100 nm (Kurniawan, Cheong, Razak, Lockman, & Ahmad, 2011a; Wilk et al., 2001; Wong & Cheong, 2010). The physical thickness of SiO₂ has become thinner below the threshold of electron tunneling limit (\sim 2 nm) which causes unacceptable gate leakage current density, J_g (exceeding 1 A cm⁻² at $V_g = 1$ V) due to oxide tunneling effect, limiting the low power application (Kamata, 2008; Robertson & Wallace, 2015). In addition, relatively low carrier hole mobility, (μ_h) in Si is another significant drawback for high-speed CMOS device operation (Kamata, 2008). To address this issue beside shrinking approach numerous alternative high mobility substrate materials from group IV and III-V with superior mobility have been extensively studied for replacement of SiO₂ (Del Alamo, 2011; Juan. Gao et al., 2017; He, Chen, & Sun, 2013). Germanium, (Ge) gained widespread interest due to having higher hole mobility, (μ_h) in contrast to other group IV and III-V which is (1900 cm²V⁻¹s⁻¹) four times higher than Si. Moreover, the small bandgap of 0.67 eV enables formation of low resistance metal contacts which is advantageous for further voltage scaling in MOSFET devices (Kamata, 2008; Lei, Goh, Zainal Abidin, & Wong, 2017; Saraswat, Chui, Krishnamohan, Nayfeh, & McIntyre, 2005). It has been reported, current MOS scaling requires $EOT \sim 0.68$ nm which needs a $t_{ox} \sim 4.36$ nm for Ge substrate interfaces (Goley & Hudait, 2014).

Unfortunately, the interface quality of native oxide GeO₂/Ge is highly unstable in counter to SiO₂/Si interface within the CMOS scaling limit. It has been identified that the quality of Ge MOS interface degrades mainly due to hygroscopic nature of GeO₂ and deteriorating tendency of GeO₂ into detrimental Ge–O species during thermal processing even below 420 °C thus degrading the electrical properties (Kita et al., 2008; Prabhakaran, Maeda, Watanabe, & Ogino, 2000). In addition, GeO₂ possess a low conduction band offset of 1 eV or less with a small dielectric constant ($k \sim 5-7$) which is a prominent obstacle in scaling for the expected CMOS devices (Chroneos, Schwingenschlögl, & Dimoulas, 2012). Therefore, the major issue regarding Ge is to maintain the integrity of GeO₂, while having ultra-thin GeO₂ interface without degradation of electrical properties.

Fortunately, the continuous development of high-k gate dielectrics has brought the opportunity to utilize both geometry of high-k and Ge together. Deposition of high-k materials on Ge substrate was expected to improve the imperfection of native oxide as well as would comply with the scaling limit. Several high-k materials have been extensively on Ge including transitional metal oxides (TMOs) hafnium oxide, HfO2 (Oh et al., 2015; Venkata Rao et al., 2018), Aluminum oxide, Al₂O₃ (Botzakaki et al., 2018; X. Yang et al., 2014) Zirconium oxide, ZrO₂ (Lei et al., 2017; Lei, Zainal Abidin, & Wong, 2018) and rare earth oxides (REOs) such as Lanthanum oxide, La₂O₃ (Mavrou et al., 2007; J. Song et al., 2009), cerium oxide, CeO₂ (Brunco et al., 2007), Gadolinium oxide Gd₂O₃, (Evangelou, Rahman, & Dimoulas, 2009), Yttrium oxide, Y₂O₃ (Nishimura et al., 2011; Seo et al., 2017; Zimmermann, Bethge, Winkler, Lutzer, & Bertagnolli, 2016), Thulium oxide, tm₂O₃ (Zurauskaitea et al., 2018), Samarium oxide, Sm₂O₃ (C. C. Lin, Wu, Wu, & Lee, 2014) has been widely studied to improve the MOS device. It has been reported that REOs were found to be more Ge friendly in improving electrical properties and thermodynamic stability in compare to transitional metal oxides (Chroneos et al., 2012; Houssa, Pourtois, Caymax, Meuris, & Heyns, 2008; Lu et al., 2014). The key reason for such

improvement has been pointed due to-spontaneous growth of rare earth germanate (RE–O–Ge) interfacial layer (IL) caused by catalytic oxidation of Ge atoms through the vigorous reaction between REOs and Ge interface (Chroneos et al., 2012; Zhao, Liu, Wang, Wang, & Wang, 2018). However, a complete MOS device based on REOs/Ge which can compete the superior features of SiO₂/Si is still an open question. Although REOs have emerged as potential contenders for high-k material but yet but many of highk metal oxides (both TMOs and REOs) in Germanium still suffers from high thermodynamic instability, large leakage current density J_a , poor C-V characteristics, low recrystallization temperature and high interface trap density, D_{it} in compare to conventional SiO₂ (Dimoulas et al., 2007; Evangelou, Mavrou, Dimoulas, & Konofaos, 2007; Goley & Hudait, 2014; Kamata, 2008; Venkata Rao et al., 2018). For instance, La₂O₃ exhibits low $D_{it} \sim 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$ and well-shaped C-V (capacitance – voltage) curve, but with an expense of large gate leakage $J_g \sim 10^{-1}$ A cm⁻² with lower $k \sim 9$ which conflicts with the requisite of scaling (Mavrou, Galata, et al., 2008). Moreover, another major concern for REOs is their hygroscopic characteristic, which influences adversely on MOS scaling by causing increment of physical thickness and reduction of dielectric constant (Chin, Cheong, & Hassan, 2010). Thus, selection of suitable high-k/Ge gate stack with coordination of good interfacial and electrical properties is still a key challenge.

Among these oxides, Sm_2O_3 can be a potential choice of high-*k* gate oxide material to be integrated on Ge substrate due to its relatively high dielectric constant, *k* value (7 - 18), which is the second highest among the REO below La₂O₃, large energy band gap, $E_g \sim (4.33 \text{ eV})$, high electrical breakdown field, E_{BD} (5 MV cm⁻¹ to 7 MV cm⁻¹) and conduction band offset, CBO (ΔE_c) ~ 2.7 eV with Si (Goh, Haseeb, & Wong, 2016b; C. C. Lin et al., 2014). Moreover, Sm₂O₃ possesses less hygroscopic nature because of its less electro-positivity due to small ionic radius (Goh, Haseeb, & Wong, 2017a). Numerous investigations of Sm₂O₃ have revealed well behaved electrical characteristics and thermodynamic stability when it is employed on Si substrate (Chin & Cheong, 2011; Goh et al., 2016b; Pan & Huang, 2010). Conversely, there are very limited investigation of Sm_2O_3 on Ge substrate. C.C Lin et al., (2014) reported the growth and electrical properties of Sm_2O_3 on Ge substrate, deposited via e-beam evaporator followed by furnace annealing. The investigation has demonstrated that post-annealing of Sm_2O_3 on Ge reduces average interface trap density D_{it} in the order of 10^{11} eV⁻¹ cm⁻², increase gate capacitance with less frequency dispersion and exhibits significantly low equivalent oxide thickness, $EOT \sim 0.61$ nm. The improvement was reasoned due to formation samarium germanate (SmGeO_x) interfacial layer (IL). Moreover, it was predicted that the fourfold coordinated configuration of Samarium (Sm) in the GeO₂ network was advantageous in passivating Ge dangling bonds to improve the electrical characteristics. (C. C. Lin et al., 2014). Since there is very limited study on Sm_2O_3/Ge gate stack and the reported investigation provides partial insight on Sm_2O_3/Ge , henceforth this material is considered for further investigation.

Another alternative REO is Holmium oxide (Ho₂O₃) can also be a potential candidate for transforming into MOS gate stack, since this material entails a moderate dielectric constant $k \sim 13$ and bandgap energy, E_g (5.3 eV), conduction band offset, CBO (ΔE_c) ~ (2 eV) with respect to Si substrate (Pan, Chang, & Chiu, 2010). Ho₂O₃ indicates least hygroscopic nature among (REOs) due to owing higher electronegativity and lattice energy. Despite of having such potential characteristics, the investigation on Ho₂O₃ based MOS devices are very scarce. Pan, Chang, et al., (2010) has demonstrated inspiring outcome in terms of electrical and thermal properties for Ho₂O₃/Si gate stack MOS device. However, there is no pertinent investigation of Ho₂O₃ as dielectric material for Ge based MOS devices. Therefore, beside Sm₂O₃ the investigation will also consider Ho₂O₃ for realization of High-*k*/Ge gate stack.

So far, several studies have demonstrated that incorporation of nitrogen (N) is an efficient technique to improve chemical and thermodynamic stability of oxide/Ge interface (Dushaq, Nayfeh, & Rasras, 2018; Fukuda, Okamoto, Iwasaki, Otani, & Ono, 2011; Maeda et al., 2004, 2006; Yi, Chang-Liao, Hsu, & Huang, 2018). It was found that incorporation of nitrogen effectively passivates Ge dangling bond and suppresses oxygen (O) diffusion by forming Ge-N /Ge-O-N interfacial layer which blocks the formation unstable Ge-O species in some extent and thus reducing leakage current density increasing capacitance and the interface quality (Kutsuki, Okamoto, Hosoi, Shimura, & Watanabe, 2009; Maeda, Nishizawa, Morita, & Takagi, 2007; Otani et al., 2007). N₂O and NO are commonly used gases for CMOS fabrication using thermal oxidation and nitridation in semiconductor industry. N₂O was selected for this study because it has less toxic nature higher activation energy than NO (Lei et al., 2018). Besides, a sputtering process with pure metal followed by thermal oxidation and nitridation in N₂O ambient has shown promising outcome forming stoichiometric metal oxide and improving electrical characteristics for both Ge and Si substrate (Lei et al., 2018; Wong & Cheong, 2011b, 2011c). However, there is no substantial investigation on thermal oxidation and nitridation using N₂O for sputtered Samarium (Sm) and Holmium (Ho) into Ge substrate. Nonetheless, oxide film and interfacial layer quality as well as the electrical properties can be significantly influenced by the oxidation ambient, deposition temperature, deposition duration and type of substrate (Goh, Haseeb, & Wong, 2016a; Hetherin, Ramesh, & Wong, 2017a; Lei et al., 2017; Wong & Cheong, 2012a). Therefore, it is imperative to investigate the effect of oxidation/nitridation (N₂O ambient) while varying deposition temperatures and durations of sputtered Samarium on Germanium (Sm/Ge) and Holmium on Germanium (Ho/Ge) system.

1.3 Objective of the Research

The main purpose of this study is to develop Samarium oxide (Sm₂O₃)/Ge and Holmium oxide (Ho₂O₃)/Ge based gate stack using metal sputtering followed by thermal oxidation/nitridation (oxy-nitridation) within current MOS scaling trend corresponding to high electrical performance and greater thermodynamic stability. The following goals are to be attained for this research.

- To investigate the effect of various temperatures and durations on the chemical, structural, and electrical properties of sputtered pure Samarium (Sm) metal on Ge substrate following thermal oxidation/nitridation in N₂O ambient.
- To investigate the influence of various thermal oxidation/nitridation durations on the chemical, structural, and electrical properties of pure Holmium (Ho) metal on substrate Ge in N₂O ambient.
- To determine the band alignment mapping of both thermally oxidized/nitrided Sm₂O₃/Ge and Ho₂O₃/Ge gate stack.
- 4. To develop a possible mechanism model of thermal oxidation/nitridation to demonstrate the growth of the thermally oxidized/nitrided sputtered pure Sm thin film on Ge substrate as well as for the thermally oxidized/nitrided sputtered pure Ho thin film sputtered on Ge substrate.
- 5. To compare physical and electrical properties between thermally oxidized/nitrided sputtered pure Sm/Ge and Ho/Ge.

1.4 Scope of Study

This investigation aims on the deposition of sputtered Sm/Ge and Ho/Ge using radio frequency (RF) Physical Vapor Deposition (PVD). Sm₂O₃ and Ho₂O₃ gate dielectric layer is expected to form after thermal oxidation/nitridation of sputtered Sm/Ge and Ho/Ge. Several variables will be investigated in this study which includes (i) effect of thermal oxidation/nitridation in N₂O gas ambient for several temperatures of sputtered Sm/Ge (ii) Influence of different thermal oxidation/nitridation durations on sputtered Sm/Ge in N₂O gas ambient (iii) Influence of different thermal oxidation/nitridation durations in N₂O gas ambient for sputtered Ho/Ge. Chemical and structural properties will be analyzed using X-Ray Diffraction (XRD), X-Ray Photoelectron Spectroscopy (XPS), RAMAN Spectroscopy, High Resolution Transmission Electron Microscopy (HRTEM). Meanwhile the electrical properties will be determined from Semiconductor Characterization System (SCS).

1.5 Structure of Thesis

This thesis is segmented into five chapters. Chapter 1 provides a brief introduction based on the research gap and main objectives of this research. Chapter 2 discusses the relevant theory and previous works associated to the development of MOS technology. The experimental design and process flow, required equipment, experimental procedures, and characterization techniques has been discussed in chapter 3. The chapter 4 discusses experimental findings which incorporates detail analysis and comparison with previous reported works. Chapter 5 concludes and summarizes the investigations of this study along with recommendation for future research.

CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

Due to aggressive downscaling of SiO₂/Si based MOS technology SiO₂ has reached its potential limit and further downscaling led to degradation of thermal and electrical properties of the downscaled devices. Accordingly, improvement in the structure of transistor and incorporation new materials has become the key route for increasing MOS technology and field-effect transistors (FETs) performance for last few generations. Immense progress has been made by the introduction of high-k materials and by re-implementing Germanium (Ge) based gate stacks in recent years. Nevertheless, numerous challenges have been introduced for high-k/Ge interface which includes suitability of dielectrics on Ge, complexities in deposition methods and incompatibility of metal gate and high-k interface. Based on current progress trends and fundamental considerations, the literature review chapter includes brief discussion on current MOS scaling trend, limitation of Si based MOS technology, requirements for high-k material. This chapter also reviews superior properties of Germanium as alternative substrate material and concerns regarding native GEO₂. Following that commonly used transitional metal oxides (TMOs) as high-k dielectrics on Ge has been documented. As rare earth oxides (REOs) as high-k has been the key route of this study, so this chapter also describes selection criteria of rare earth oxides (REOs), common deposition methods of REOs and subsequently, performance of several materials from lanthanide group has been reviewed for integration into REOs/Ge MOS application. Since Sm₂O₃ and Ho₂O₃ was selected for this study and so this chapter pays particular consideration in reviewing the physical characteristics, electrical features, and various deposition techniques as high-k oxides. Lastly, role of incorporating nitrogen (N) during the fabrication process of gate oxides into Ge has been described.

2.2 Moore's Law and MOS Scaling Trend

The device downscaling of Si based metal-oxide field effect transistor (MOSFET) has become inevitable to cater the increasing demand of low power consumption, high speed operation, high performance device reliability, and indeed lowering the cost of MOSFET devices (Thompson & Parthasarathy, 2006; Wilk et al., 2001). This advancement has been made according to the Gordon Moore's prediction of scaling. In 1965, Gordon Moore the founder of intel predicted that the quantity of transistors in an integrated circuits (ICs) will be double every two years (Mack, 2011). For past few decades the components of an integrated circuited was truly doubled about every 18 months. Corresponding to this prediction for every technology generation the channel length should reduce 0.7 times. In 1970s, the channel length was downscaled to 8 µm. Following the downscaling trend in year 2003 the channel length reached nano-meter scale (Radamson, H. & Thylen, 2014). As the density of transistor increased exponentially the quantity eventually transistor per chip also experienced an exponential growth. In 1975 quantity of transistors per chip was 16000 which now turned in a range above a billion (Mack, 2011). The evolution of transistor gate length from 1970 to 2030 is shown in Fig. 2.1. Since planar Si MOS have confronted its scaling limit under gate length of 20 nm and breakthrough is essential for further expansion. In the modern CMOS scaling the technology node has reached sub-100 nm and beyond (Chin et al., 2010; He et al., 2013; Ieong, Doris, Kedzierski, Rim, & Yang, 2004; Kurniawan et al., 2011a). It has been reported that according to the current MOS scaling trend the gate oxide should be scaled down at a physical gate oxide thickness of 1.5 nm at maximum gate leakage allowance

of 1.5×10^{-2} A cm⁻² a gate voltage, V_g of 1V without sacrificing the breakdown voltage (Ieong et al., 2004; Muller et al., 1999; Robertson & Wallace, 2015). Table 2.1 displays the downscaling evolution where the native SiO₂ could serve as gate insulation oxide up to it reached 90 nm. Afterwards SION substituted the pure native SiO₂. In future it is expected that new high-*k* element will swap SiON. Nevertheless, the physical oxide thickness remained at 5 atomic layers in order to inhibit the QMDT current conduction (Sicard & Aziz, 2011).



Figure 2.1: Evolution of transistor gate length from 1970 to 2030 (Radamson, H. & Thylen, 2014).

Technology node	0.18 μm	0.13 μm	90 nm	65 nm	45 nm	32 nm	22 nm
Gate length	130	70	50	35	25	17	12
Gate atoms	10	8	5	5	5-10	5-10	5-10
Gate material	Poly SiO ₂	Poly SiO ₂	Poly SiO ₂	Poly SION	Metal High- <i>k</i>	Metal High- <i>k</i>	Metal High- <i>k</i>
Manufacture year expected	1999	2001	2003	2005	2007	2008	2011

Table 2.1: CMOS technology node evolution (Sicard & Aziz, 2011).

2.3 Basic Principle of MOS and MOSFET

In order to understand the fundamentals of MOS scaling it is vital to recall the basic structure of MOS capacitor and a MOSFET. A typical MOS structure arrangement with stacking layer is shown in Fig. 2.2a where gate oxide layer is inserted between gate electrode and substrate. I_{in} denotes the input current while I_{out} indicates the output current. The functionality of a capacitor basically based on capacitances induced due to insulation properties of gate oxide. On the other hand, typically, a MOSFET consists of three building blocks which includes gate (G), source (S) and drain (D) as shown in Fig. 2.2b. In operation of MOSFET applied voltage in the gate capacitor junction controls the device performance. Depending on the applied gate voltage, a positive or negative charges are introduced in the channel region underneath the gate capacitor. According to the polarity of carriers induced in the channel region, the nodes of drain and source either connects or isolates (He et al., 2011; Taur, 2002). The extent of induced charge (Q) induced across the channel region is a combination of gate capacitance (C_{ox}) and applied gate voltage (V) which can be expressed by Eq. 2.1 (Goley & Hudait, 2014)

$$\boldsymbol{Q} = \boldsymbol{C}_{ox}\boldsymbol{V}$$
 (Equation. 2.1)

The gate capacitance, (C_{ox}) of a parallel plate MOS capacitor can be expressed by Eq. 2.2 (Goley & Hudait, 2014; Houssa et al., 2006; Robertson & Wallace, 2015).

$$\boldsymbol{C_{SiO_2}} = \frac{k_{SiO_2} \,\varepsilon_o \,A}{t_{SiO_2}} \tag{Equation. 2.2}$$

Where, k_{siO_2} is the relative dielectric constant of SiO₂, ε_0 is the permittivity of free space, t_{siO_2} is the physical oxide thickness of SiO₂ and A is the area of gate electrode. Another important constraint of MOSFET device operation is to maintain high drive current between drain-source nodes along the channel region. The relationship of drive current, I_D and gate capacitance can be given by Eq. 2.3 (Taur, 2002).

$$I_D = \frac{W}{L} \mu C_{ox} (V_G - V_T)^2 \qquad (\text{Equation. 2.3})$$

Where, I_D = Drive current, W = Channel width, L = Channel Length, C_{ox} = Gate capacitance, V_T = Threshold voltage, V_G = Gate voltage, μ = Carrier mobility. From Eq. 2.2 and 2.3 it is apparent that, reduction of channel length and physical oxide thickness of the MOSFET dimensions enables to increase the drive current between drain and source region. Instinctually, shrinkage of t_{ox} increases C_{ox} and hence the of amount induced charged in the channel region is increased, while the reduction of L reduces the travelling distance of the induced charges to initiate current flow. Reduction of the gate oxide thickness along with the channel length also facilitates to retain the gate electrode's control on the short channel effects. Based on Eq. 2.2, the gate electrode area (A) is constant and so reduction of gate oxide thickness is the most preferrable way to increase gate capacitance which ultimately attributes to maintain high drive current in the channel region. Accordingly, shrinkage of oxide thickness technology has been considered as the key route for device downscaling for last few decades. Tremendous advancement of planer Si-CMOS has been driven by the capability of continual shrinkage of these critical dimensions due to excellent insulating properties of native SiO₂ (Hirose et al., 2000). It is obvious that the potentiality of most reliable SiO₂/Si interface that has been used in today's semiconductor industry will sooner reach limits of scaling, thus introducing a major question towards the sustainability for continuity of device miniaturization.



Figure 2.2: (a) Typical MOS capacitor stacking arrangement (b) Conventional structure of a MOSFET.

2.4 Limitations of SiO₂/Si Scaling

It has been reported that SiO₂/Si exhibits low interface trap density, $D_{it} \sim 10^{11}$ eV⁻¹ cm⁻², leakage current density, $J_g \sim 1 \times 10^{-12}$ A cm⁻² and electrical breakdown field, $E_{BD} \sim 15 \text{ MV cm}^{-1}$ with high thermodynamic stability. (Houssa et al., 2006; Park & Kim, 2005; Wilk et al., 2001). Despite of owing such outstanding properties further downscaling of SiO₂ is confronting significant challenges to comply with the current scaling trend. The main issue of SiO₂ based metal oxide semiconductor (MOS) downscaling is associated with SiO₂ layer thickness and electron tunneling mechanism which exponentially increases leakage current density with the decrease of SiO_2 thickness (Cui et al., 2022; D. Wang, He, Fang, et al., 2019). It has been reported that, aggressive downscaling of SiO₂/Si interface below 1.2 nm causes direct tunneling effect resulting in an unacceptably higher gate leakage current density (> 1 A cm⁻² at 1 V) (He et al., 2011; Kurniawan et al., 2011a; Robertson, 2004; D. Wang, He, Hao, Gao, & Zhang, 2019). This has been reasoned that due to drastic downscaling the oxide thickness has become so thin that, it has reached the electrons tunneling limit of ~ 2 nm which allows electron to penetrate through SiO₂ layer without passing through the conduction band (Goley & Hudait, 2014; Kamata, 2008). It is notable that for SiO₂/Si interface $J_g \sim 1 \times 10^{-12}$ A cm⁻² for 3 nm
oxide thickness which increased to unacceptably higher of $J_g \sim 1 \text{ A cm}^{-2}$ (twelve order magnitude higher) when thickness reduced to 1.5 nm (Lo, Buchanan, Taur, & Wang, 1997; Robertson, 2004). Over and above the drastic downscaling of SiO_2 also springs up difficulty in manufacturing such thin oxide layer while maintain thickness reliability issue. When the oxide layer reaches such thin thickness critical defect density at the SiO₂/Si interface has been introduced by the flow of excessive charge carriers during MOSFET operation leading to the breakdown of SiO₂/Si gate unit (He et al., 2011). Consequently, the increment of leakage current expands the thermal load of the MOS device which brings severe challenges to the aim of developing low power consumption energy efficient devices. Thus, the relationship established in Eq. 2.2 motivates to increase gate capacitance, C_{ox} by increasing the dielectric constant, k value while the oxide thickness cannot be reduced further, and the gate area remains constant. Hence, alternative material with higher permittivity known as high-k is required for replacement of SiO₂ (H. D. Kim & Roh, 2006; Wilk et al., 2001; Wong & Cheong, 2010). Additionally, the low hole mobility (μ_h) of Si is another potential drawback on the way to high-speed CMOS devices (Frank & Taur, 2002; Kamata, 2008). From Eq. 2.3, it is obvious that high carrier mobility, μ is a potential factor on the way to high-speed device operation. Accordingly, replacement of Si substrate with alternative higher carrier mobility substrate and high-k dielectric material is a potential solution on way to realization of low power consumption and high-speed devices within the scaling trend.

2.5 Requirements for Implementing High-k Gate Oxide

The crucial purpose of high-k oxides is to obtain larger physical oxide thickness while obtaining similar capacitance as of SiO₂ by increasing the dielectric permittivity kvalue of new dielectric material following Eq. 2.2. The high-k gate stack provides benefit to obtain physically thicker oxide layer equivalent to SiO₂ thickness as shown in Fig.2.3. It has been reported that, using high-k dielectric material reduces gate leakage between dielectric oxide layer and the substrate (Liang et al., 2018; D. Wang, He, Fang, et al., 2019). This is because high-k leads to physically thicker oxide material which could be barrier for electron tunneling through dielectric layer (Goley & Hudait, 2014). Since further downscaling SiO₂ below 1.2 nm is no longer applicable, the new suitable high-k oxide should maintain a convenient equivalent oxide thickness, (*EOT*) calculated according to Eq. 2.4 (Goley & Hudait, 2014; Robertson & Wallace, 2015).

$$\boldsymbol{t}_{ox} = \boldsymbol{EOT} = \left(\frac{3.9}{k_{high-k}}\right) \boldsymbol{t}_{high-k} \qquad (\text{Equation 2.4})$$

Where, k_{high-k} is the dielectric permittivity of new high-*k* material, t_{high-k} is the physical oxide thickness of new high-*k* dielectric material. For the purposes to retain gate oxide thickness of 1.2 nm or to further downscale the *EOT* the thickness of high-*k* or/and the *k* value could be enhanced (Sharma, Kumar, & Anthony, 2001; Wong & Cheong, 2010). Fig.2.4 illustrates the benefit of employing high-*k* in terms of mitigation of gate leakage between insulator and the substrate layer for MOS devices.



Figure 2.3: TEM image of SiO₂/Si and HfO₂/Si gate stack thickness (Robertson & Wallace, 2015).



Figure 2.4: Advantage of using high-k as gate material to reduce leakage current.

However, reduction of only t_{high-k} using ultra-thin gate high-k oxides could not resolve the problem of gate leakage current owing to direct electron tunneling thus, selection of high-k material with large k value is the key route to mitigate this problem In order to be applicable for dielectric oxide for MOS integration the selection of high-kshould retain k value ranging from 10 - 30. (Engstrom et al., 2007; Gillen & Robertson, 2013; Goh et al., 2017a). However, selection of k value should not be larger than 25 because very large k enhances the thickness of gate oxide beyond the dimension of gate channel length which diminishes the gate control on channel region. Eventually, degrade the FET device performance owing to short channel effect. (He et al., 2011; Houssa et al., 2006). It is well known that dielectric tunneling current is dependent on barrier height of oxide/substrate interface where reduction of barrier height exponentially increases the leakage current. Thus, it is required to have a higher conduction band offset (CBO) > 1eV which could retain a moderate barrier height to impede the direct tunneling of electron into conduction band of high-k material that causes unacceptable leakage of the device. In order to achieve large CBO with the semiconductor substrate it is important to select high-k material with higher bandgap. It has been suggested that, a higher bandgap, $E_g >$ 5 eV is preferably acceptable for high-k gate oxides. (Houssa et al., 2006; Robertson, 2006; Robertson & Wallace, 2015). Fig. 2.5 shows the trend of bandgap of high-k dielectrics with respect to dielectric constant where the bandgap shows an inversely proportional relationship to dielectric constant. Consequently, range of selecting high-k dielectric material is being restricted due to low bandgap dispute which would hamper the aim to retain large CBO requirement.



Figure 2.5: Trend of bandgap of high-*k* dielectrics with respect to dielectric constant (Wilk et al., 2001).

In addition, the high-k should maintain good interface quality and thermodynamic stability with the substrate material, superior film morphology, and compatibility with the current integration in order to be used for current and future CMOS technology (Kamata, 2008; Kaya & Yilmaz, 2019). For Ge based MOS device *EOT* limit has reached 0.68 nm. In order to serve this purpose a high-k material corresponds to physical oxide thickness $t_{ox} \sim 4.36$ nm with dielectric constant $k \sim 20$ is required. The following key requirements should be fulfilled for being applicable for high-k dielectric gate oxide material. (Goley & Hudait, 2014; Robertson & Wallace, 2015).

- 1. Meet the scaling trend set by ITRS of low equivalent oxide thickness.
- 2. The leakage current density must not exceed $J_g \sim 1.5 \times 10^{-2}$ A cm⁻² at $V_g \sim 1$ V.
- Should have high enough conduction band offset and valance band offset (at least 1 eV for both).
- 4. Should exhibit low interface trap density D_{it} (less than $10^{11} \text{ cm}^{-2} \cdot \text{eV}^{-1}$).
- 5. Acceptable high dielectric electric breakdown field, E_{BD}
- 6. Thermodynamic and kinetic stability.

Several high-k gate oxides such as Al_2O_3 , (I. S. Jeon et al., 2003; Lale et al., 2021; Rahman, Kim, Kim, & Kim, 2019; Shao et al., 2003), ZrO₂ (Chew et al., 2016; H. D. Kim & Roh, 2006; Shuan Li et al., 2020; Weinreich et al., 2009; Wong & Cheong, 2011b), HfO₂ (Choi, Shin, Park, & Yoon, 2001; Liang et al., 2018; Park & Kim, 2005; Srivastava, Nahar, & Sarkar, 2011; Tan, 2010; Tirmali, Khairnar, Joshi, Electronics, & 2011, 2011), La₂O₃ (Fei, Liu, Wang, Zhao, & Wang, 2016; Jun, Jun, Wang, Won, & Choi, 2002; Ramana et al., 2011; Schamm et al., 2009; J. Q. Song, Qian, & Lai, 2018), Y₂O₃ (Evangelou, Wiemer, Fanciulli, Sethu, & Cranton, 2003; Ioannou-Sougleridis et al., 2004; Pan & Lee, 2007; Quah & Cheong, 2015), Ta₂O₅ (Rao, Verma, & Singh, 2015; Spassov, Atanassova, & Virovska, 2006), TiO₂ (Gullu & Yildiz, 2021; Nagabharana, Kumaraswamy, Gundanna, Susheel, & Umananda, 2020; Sekhar, Nanda Kumar Reddy, Venkata Rao, Mohan Rao, & Uthanna, 2014) and Gd₂O₃ (S. Li et al., 2019) were widely investigated on Si substrate for replacing SiO₂ in realization of advanced MOS technology. However, high-k/Si are subjected to several limitations (Chin et al., 2010; Leskela, Kukli, & Ritala, 2006; Wong & Cheong, 2010). Previous researchers has classified some major limitations which includes (i) growth of interfacial layer during fabrication period and afterwards at the point of post deposition annealing (Kurniawan, Cheong, Razak, Lockman, & Ahmad, 2011b; Ma & Zhang, 2008; Wong & Cheong, 2010) (ii) lateral oxidation takes place at the edge of oxide interface and development of micro crystal

during heat treatment (Iwai et al., 2002) (iii) dielectric defects caused by the precursors used during chemical vapor deposition (Choi et al., 2001) (iv) existence of fixed charges which shifts the flatband voltage and large interface trap density (Juan. Gao et al., 2016; Iwai et al., 2002; Loo et al., 2005; Naumann, Otto, Wehrspohn, Werner, & Hagendorf, 2012; Wong & Cheong, 2011b). (v) lowering of conduction band offset with respect Si (Alers et al., 1998; Robertson & Wallace, 2015; Shao et al., 2003). Moreover, low-k value and increment of overall dielectric thickness by the expansion of additional interfacial thickness made it challenging to utilize these metal oxides with Si interface under the current MOS scaling trend below 1.2 nm (He et al., 2011; Houssa et al., 2006). Therefore, combination of high-k with alternative substrate material intended to replace SiO₂ and Si substrate is the potential aspect for this research.

2.6 Ge as Alternative Substrate Material

The continuous progress of integrated circuit (IC) efficiency through device miniaturization has been the key route for the rapid success of semiconductor micro-electronics industry. Currently, several materials such as Germanium (Ge), strained $Si_{1-x}Ge_x$ and III–V group compounds have been widely investigated to replace Si in order to continue the miniaturization of CMOS device (Del Alamo, 2011; Kamata, 2008; Simoen et al., 2012; J. Q. Song et al., 2018; Yao et al., 2021). It is well known that CMOS technology is a combination of NMOS and PMOS components, and so both high electron mobility μ_e and high hole mobility μ_h is required. Due to low μ_h of Si, integration PMOS devices is bottle-neck for further development. In this regard mobility enhancement through strained technology has been subjected to limitations due to lattice mismatch, threading dislocations, thermal budget mismatch and abrupt change in conduction band. Shifting on the III–V compounds NMOS device has been successfully integrated due to its high electron mobility μ_e while integration of PMOS still remains a key challenge. This fact is reasoned due to large discrepancy between μ_e and μ_h in III–V compounds where the μ_e several times larger than μ_h (Del Alamo, 2011; Juan. Gao et al., 2016; Goley & Hudait, 2014; He et al., 2013; Simoen et al., 2012). According to Table 2.2, among all the alternative material Ge has gained widespread due to its higher electron mobility which is $2 \times \mu_e$ (Si) and hole mobility $4 \times \mu_h$ (Si) (Kamata, 2008). Ge also maintains smaller band gap ~ 0.67eV which will be advantageous for scaling supply voltage, V_T (Saraswat et al., 2005). It is expected that replacing Si with Ge would reduce the amount of power consumption and enhance the high-speed operation for PMOS device. Based on these backgrounds, it can be accepted that Ge would be the best choice to replace Si for further miniaturization of CMOS devices.

Semiconductor Material	μ _e (cm ² /Vs)	$\mu_h \mathrm{cm}^2/\mathrm{Vs}$	
Si	1400	450	
Ge	3900	1900	
GaAs (III–V)	8500	400	
InP (III-V)	4600	650	

Table 2.2: Electron mobility, μ_e and hole mobility, μ_h of different semiconductor materials (Kamata, 2008).

2.6.1 Issues in growth of Native GeO₂

Although Ge has attracted more attention due to its superior carrier mobility, however, it has turned challenging to realize in practical applications. This is because the native, GeO_2 is not that thermodynamically stable as SiO_2 . While comparing Si and Ge as substrate material it was found that the native oxide GeO_2 of Ge grows faster than that of SiO_2 when exposed to air (Kita et al., 2008; Sahari et al., 2013). It was pointed that the thickness of native GeO₂ increases 0.4 nm with 10 min after exposure to air at 24 °C. The thickness of GeO₂ increases between growth rate of 0.2 nm - 0.3 nm with an interval of 10 minute per step as shown in Fig. 2.6. This variation has been attributed to the layered growth phenomenon of GeO₂ and the bond length of Ge-O which was about 0.2 nm. It was also observed that the growth rate of n-Ge (100) orientation is faster than the p-Ge (100) which indicates a more stable surface for p-Ge (100) (Sahari et al., 2013). Since the native oxide of Ge deteriorates the electrical performance of MOS devices hence it is required to clean the wafer surface for eliminating the detrimental native GeO₂ prior to deposition of dielectric film into Ge. For the cleaning of bare Ge surface all halogen acids which includes, hydrochloric acid (HCL), hydrofluoric acid (HF) and hydroiodic acid (HI) could be utilized to remove the native GeO₂ and other impurities of Ge (Kamata, 2008).



Figure 2.6: Comparison of native oxide Growth rate for Si and Ge substrate (Sahari et al., 2013).

2.6.2 Thermal Issues of GeO₂/Ge Stack

Since the native GeO₂ growth increases rapidly during air exposure and so elevation of temperature is an effective method for hindering the growth of native GeO₂ prior to the deposition of dielectric film into Ge substrate. However, it has been pointed that the desorption of GeO₂ occurs at GeO₂/Ge while such desorption was not found in GeO₂/Si interface (Kita et al., 2008). It is believed that the origin of GeO₂ degradation is caused by the interfacial redox reaction between GeO₂ and Ge which initiates volatile GeO desorption according to Eq. 2.5 (Prabhakaran et al., 2000).

$$GeO_2 + Ge \rightarrow 2GeO(s) \text{ or } 2GeO(g) \text{ at } 400 \,^{\circ}C$$
 (Equation 2.5)

S. K. Wang et al., (2010) have demonstrated that oxygen vacancies that are being generated from Eq. 2.5 diffuses through the GeO₂ network leading to GeO (g) desorption as shown in Fig. 2.7. It is also pointed that exchange of in-diffusing O with GeO₂/Ge interface also reinforces the oxygen vacancy production during thermal oxidation (S. K. Wang et al., 2010). It has been reported that the diffusivity of oxygen is much higher than Ge as a result the prominent transportation of oxygen through the native GeO₂ occurs at low temperature in contrast to SiO₂/Si stack processing temperature (500 – 600) °C. In fact, the transportation of oxygen into GeO₂/Ge interface has been observed below 500 °C temperature which was mediated by generation of mobile oxygen vacancies (Da Silva, Rolim, Soares, Baumvol, & Krug, 2012; X. Wang, Nishimura, Yajima, & Toriumi, 2017). Moreover, the diffusion mechanism of oxygen into GeO₂/Ge structure and production of oxygen vacancies from GeO₂ network also influences the thermal processing of metal/dielectric oxide/semiconductor substrate stack structure. This has been evidenced from comparison of thermally annealed Pt/HfO₂ gate stack based on both Si and Ge substrate at O₂ ambient. It was found that Ge substrate introduces higher quantity of O diffusion

into the dielectric/substrate interface at processing temperature (450 - 500) °C in compared to Si substrate due to defective character of native GeO₂ than SiO₂ (Rolim, Gobbi, Soares, & Radtke, 2015). This stimulates higher oxidation of the Ge substrate which creates major challenges in terms of thermodynamic stability, scalability and gate capacitance of dielectric oxide since oxygen diffusion promotes higher growth rate of interfacial GeO₂ interface (Etcheverry, Boudinov, & Soares, 2019). Beside GeO desorption, water solubility of GeO₂ is another major concern in the way to integration Ge based MOS devices (Kamata, 2008; Watanabe et al., 2012).



Figure 2.7: Schematic of GeO desorption mechanism from GeO₂/Ge interface (S. K. Wang et al., 2010).

2.6.3 Electrical Characteristics of GeO₂

Several reports evidenced that, desorption of large amount of GeO (g) ultimately deteriorates the electrical performance by generation additional fixed charges, inevitable hysteresis effect, large flatband voltage shift and large interface trap density (Bellenger et al., 2008; Jung et al., 2012). The investigation of Kita et al., (2008) recorded a superior *C-V* feature for GeO_2 when deposited on Si while a huge hysteresis observed for GeO_2/Ge as shown in Fig. 2.8. This suggests that the instability of GeO₂ is not an intrinsic property rather incorporation of Ge into GeO₂ network during thermal processing is responsible for the instability of GeO₂/Ge stack (Kita et al., 2008). In case of GeO₂/Ge stack, the interface trap density, D_{it} was found to be scattered in range of $10^{11} - 10^{12}$ eV⁻¹cm⁻² which is significantly higher with respect to SiO₂/Si interface $D_{it} \sim 10^{10} \text{ eV}^{-1} \text{cm}^{-2}$ (Hosoi et al., 2009). Matsubara et al., (2008) reported that GeO₂/Ge formed by standard dry oxidation at 500 – 600 °C in O₂ ambient can achieve $D_{it} \sim 10^{11} \text{ eV}^{-1} \text{cm}^{-2}$ although the hysteresis was around >100 mv and physical oxide thickness 30.9 - 30.7 nm (Matsubara, Sasada, Takenaka, & Takagi, 2008). The dielectric constant k of GeO_2/Ge has been estimated around 5.5 - 5.9 which is another prominent obstacle for scaling that increases the equivalent oxide thickness (EOT). Furthermore, the conduction band offset of GeO_2/Ge has been reported less than < 1eV as shown in Fig. 2.9 which increases the leakage current density (Chroneos et al., 2012; Dimoulas et al., 2007; L. Lin, Xiong, & Robertson, 2010; Matsubara et al., 2008). To mitigate these issues several high-k materials with different deposition techniques have been coupled as gate dielectric for Ge-based MOS application (Goley & Hudait, 2014; Kamata, 2008).



Figure 2.8: Bidirectional C-V curve at (1 MHz) of sputtered GeO₂ on Ge, Si, and thermally oxidized Si substrates followed by post deposition annealing 400 °C (Kita et al., 2008).



Figure 2.9: Band alignment schematic of GeO₂/Ge interface and comparison with the band offset of SiO₂/Si interface. (L. Lin et al., 2010).

2.7 Common High-*k* Oxides on Ge

Since, native GeO_2 of Ge is thermally unstable, integration of high-k material into Ge has created the exciting prospect of using Ge as substrate material for MOS gate stack. Various transitional metal oxides which include HfO₂ (Arora et al., 2009; Chandra, Jeong, Park, Yoon, & Choi, 2011; Jung et al., 2012; Kita, Kyuno, & Toriumi, 2004; Misra, Garg, Srinivasan, Rahim, & Chowdhury, 2006; Oh et al., 2015; Venkata Rao et al., 2018), ZrO₂ (C. O. Chui, Ramanathan, Triplett, McIntyre, & Saraswat, 2002; Kamata, Kamimuta, Ino, & Nishiyama, 2005; H. Kim, Chui, Saraswat, & McIntyre, 2003; Lei et al., 2017, 2018; Wong, Lei, & Abidin, 2021) and Al₂O₃ (Bom et al., 2012; Botzakaki et al., 2018; Ke, Takenaka, & Takagi, 2018; Xiang et al., 2020; X. Yang et al., 2014; Yoshida et al., 2014; R. Zhang, Iwasaki, Taoka, Takenaka, & Takagi, 2011) has been intensively studied for integration of high-k/Ge gate stack. These materials were widely investigated because of their relatively high band offsets and large dielectric constant as shown earlier in Fig. 2.5. Although HfO₂ has shown compatibility with Si substrate, but this material introduces large leakage current $J_g \sim 10^{-3}$ A cm⁻² when it is directly deposited on Ge (Kamata et al., 2005). An improved electrical property for HfO₂/Ge gate stack comprising $J_g \sim 10^{-9}$ A cm⁻² at $V_g = 1$ V and $t_{ox} \sim 6.4$ nm stack has been achieved by inserting an additional interfacial layer of Ge₃N₄ at annealing temperature of 400 °C in O₂ and forming gas ambient (Venkata Rao et al., 2018). Nevertheless, this investigation is still limited to poor C-V characteristics, scaling issues and D_{it} has not been studied. Meanwhile, ZrO₂/Ge gate stack without an interfacial layer free interface inducing $J_q \sim$ 10^{-7} A cm⁻² at enhanced breakdown field E_{BD} of 16.6 MV cm⁻¹ stack has been reported using thermally oxidation (500 °C in O₂ ambient) (Lei et al., 2017). Although ZrO₂ could be formed without an interfacial layer but is limited to large hysteresis > 200 mv and large interface charge traps of D_{it} of 7×10^{12} cm⁻² (H. Kim et al., 2003). In regard to MOS fabrication process a thermal budget of 400 °C for 30 min has been acknowledged which

requires the gate dielectric layer should not recrystallize at this temperature rather be amorphous in order to maintain the high-k requirements (C. Chui, Kim, & Chi, 2002). One of the major concerns for HfO₂/Ge and ZrO₂/Ge gate stack is that Ge could diffuse toward HfO₂ and ZrO₂ dielectric layer even at low temperature less than 300 °C. Consequently, interdiffusion of Ge instigates electrically active state within the interstitial or substitutional sites of these oxides' leading to higher interface trap density (Caymax et al., 2008; Zhu, Tamagawa, M., Furukawa, & Ma, 2002). The recrystallization of HfO2 occurs at 400 °C which is also a critical issue that effects the thermodynamic stability of HfO₂/Ge stack. Considering Al₂O₃ based Ge MOS device X. Yang et al., (2014) developed Al₂O₃/GeO_x/Ge gate stack showing superior electrical properties with a low $D_{it} \sim$ 1.9×10^{11} eV⁻¹ cm⁻², $J_g \sim 10^{-7}$ A cm⁻² at $V_g = 1$ eV and small hysteresis, better C-V characteristic but with an expense of $EOT \sim 2.7$ nm. However, TiO₂/Al₂O₃ /Ge gate stack could provide promising EOT as low as 0.6 nm but exhibits poor $J_g \sim 10^{-2}$ A cm⁻² at $V_g = -1$ eV which has been reasoned due to near zero CBO of TiO₂ on Ge (L. Zhang, Thombare, & McIntyre, 2013). Another promising development of Gunji, Al₂O₃/GeO_x/Ge was made by using plasma post oxidation demonstrating $EOT \sim 1$ nm and $D_{it} \sim 1 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$. It was pointed that controlling GeO_x thickness is the critical issue for reduction of interfacial layer (IL) which results sharp increase of D_{it} while the J_g also becomes very high exceeding 1 A cm⁻² (R. Zhang, Iwasaki, Taoka, Takenaka, & Takagi, 2012). Since direct deposition of transitional metal oxides could not comply with all the requirements needed for high-k based gate stack it has turned interest in employing rare earth oxides (REOs). In addition, the EOT expenses including an IL layer has also triggered the motivation towards REOs.

2.8 Rare earth oxide as Alternative High-*k*

Recently, lanthanide based rare earth oxides have gained widespread interest for its superior features which includes high dielectric constant, large band gap and conduction band offset, high chemical, and thermal stability (Chin et al., 2010; Goh et al., 2017a; Leskela et al., 2006; Shuan Li, Wang, Lin, Wang, & Li, 2022). Fig. 2.10 displays partly section of the periodic table for rare-earth (lanthanide) elements. Among the 15 materials of lanthanides, Pm has been eliminated, since it is considered as unstable radioactive material (Iwai et al., 2002). The lanthanide REOs can appear at different oxidation states including (+2, +3, and +4) states which differentiates the oxygen composition of lanthanide oxides LnO_x. Thus, various stoichiometries of lanthanide oxides could be in form of Ln₂O₃, LnO₂ and LnO depending on the oxidation state. Among the lanthanide oxides, the stoichiometry of Ln₂O₃ has been focused for utilization into high-*k* MOS application. Since, the stoichiometry of LnO₂ is steady at +3 oxidation state while stoichiometry of LnO lacks insulating properties (Chin & Cheong, 2011; Osten et al., 2008; Scarel, Svane, & Fanciulli, 2006). Therefore, lanthanide based rare earth oxides of Ln₂O₃ have gained widespread interest for its superior features.



Figure 2.10: Candidates for metal oxide high-k gate insulator (Iwai et al., 2002).

As mentioned in earlier section the potential range of dielectric constant, k should be between 10 - 30 (Chin & Cheong, 2011; Houssa et al., 2006). REOs are well known for owing high dielectric constant, k values ranging between 7 - 30 as shown in Fig. 2.11 The high dielectric constant values of REOs should grant them to lower the EOT which is desirable for fulfilling the downscaling trend (Gillen & Robertson, 2013; Goh et al., 2017a). The dielectric constant values of REOs are associated with the crystalline composition of oxide components. Those REOs which own higher dielectric constant, k values are recommended to form stable hexagonal configurations which are known as light REOs consisting of $La_2O_3 - Pr_2O_3$. Meanwhile, the comparatively lower k value REOs are suggested to form stable cubic and monoclinic configurations for $(Sm_2O_3 - Lu_2O_3)$ (Adachi & Imanaka, 1998; Iwai et al., 2002). Although, very limited evidence is found to verify this phenomenon. On the other hand, the oxide deposition technique, its formation defect and thickness are likely to be more influential on varying the dielectric constant values (Scarel et al., 2006). For instance, $k \sim 30$ was approximated for Pr₂O₃/Si interface which was deposited employing MBE method. On the contrary, $k \sim 15$ was approximated for the same oxide when it was deposited using Atomic layer deposition (ALD) with a film thickness ranging between 80 nm to 100 nm (Fissel, Osten, & Bugiel, 2003; Kukli et al., 2004).

In order to replace SiO₂ it is essential that, the REOs should owe a high energy bandgap, E_g above 5 eV which could act as potential barrier for electron tunneling to maintain better insulation (He, Liu, et al., 2014; Houssa et al., 2006). Fig. 2.12 shows the bandgap of REOs which indicates that among the REOs the dielectric oxide of La₂O₃, Sm₂O₃, Gd₂O₃, Ho₂O₃, Er₂O₃, Tm₂O₃, and Lu₂O₃ could be serve as replacement of SiO₂ due to their high bandgap > 5eV (Goh et al., 2017a; Ohmi et al., 2001). Therefore, these REOs has been the primary consideration from lanthanide list while some of them are being widely investigated in the last few years for integration of Ge based MOS devices.



Figure 2.11: Dielectric constants of rare-earth oxides (Chin & Cheong, 2011).



Figure 2.12: Band gaps of rare-earth oxide (Goh et al., 2017a).

For consideration of high-*k* dielectric in replacement of SiO₂ another important criterion is that the band offset should be above 1 eV. In the work of Robertson, (2002) it has been suggested that, the stochiometric ratio between metal and oxygen could effectively influence the value of band offset. The greater ratio value between metal: oxygen should impose higher value of band offset. According to this investigation the CBO of HfO₂ was around 1.5 eV while La₂O₃ exhibited a CBO of 2.3 eV with respect Si interface. This has been reasoned that La₂O₃ considers larger stoichiometric ratio of metal and oxygen and so, a larger CBO was obtained in counter to HfO₂ (Robertson, 2002). This outcome also put up REOs to be a superior alternative. The comparison of CBO of several REOs as function of dielectric has been portrayed in Fig. 2.13. It is found that all the estimated CBO of REOs is above 1 eV which makes them potential candidate to be investigated for high-*k* dielectrics.



Figure 2.13: Conduction band offsets (CBO) of various REOs on Si as a function dielectric constant (Chin & Cheong, 2011).

The hygroscopic nature of rare earth material is one of the major difficulties during deposition of REOs. Due to moisture absorption the hydride Ln₂O₃-H₂O is formed in REOs which ultimately initiate the growth of hydroxide [Ln(OH)₃]. The formation of such hydroxide would eventually enhance the thickness of deposited dielectric REOs film and as well as drops the k value of the film (Engstrom et al., 2007). The reaction rate of forming hydroxide (OH⁻) in REOs due to moisture absorption increases when the ionic radius of the rare earth component is decreases (S. Jeon & Hwang, 2003). In addition, the moisture reactivity of REOs is also influenced by the positive ion electronegativity and lattice energy of REOs. The positive ion of REOs having lower electronegativity is found be highly reactive with negative ion of (OH⁻) while increasing the moisture reactivity. Moreover, with the increment of lattice energy of REOs decreases reactivity to water which declines the hygroscopic behaviour of REOs (Chin & Cheong, 2011; Goh et al., 2017a; Iwai et al., 2002). Fig. 2.14 shows the electronegativity and lattice energy values of various REOs. According to Fig. 2.14, the Lu₂O₃ is the least hygroscopic REOs as this oxide possess highest electronegativity and lattice energy while La₂O₃ is the most hygroscopic REOs for having lowest electronegativity and lattice energy. In order to mitigate the moisture absorption difficulties in REOs a typical gate electrode of Al can be deposited on the top of the dielectric REOs. It has been suggested that deposition of a gate electrode could decrease the moisture reactivity which effectively reduce the formation of (OH⁻) in REOs and so the resulting in lowering of EOT (Iwai et al., 2002). Furthermore, the hygroscopic behaviour of REOs also effects adversely on surface roughness of the dielectric interface which eventually increases the extent of leakage current density. Fig. 2.15 displays the variation of leakage current density due to surface roughening which was caused after the moisture absorption of REOs (Kakushima et al., 2006; J. Song et al., 2009). Therefore, rare earth element imposing smaller ionic radius and less hygroscopic nature are preferably considered for integration into future High-k based MOS devices.



Figure 2.14: The electronegativity and lattice energy of lanthanide REOs (Goh et al., 2017a).



Figure 2.15: Effect of surface roughness of rare-earth oxides on leakage current (Kakushima et al., 2006).

2.9 Common Deposition Methods of REOs

Several deposition methods have been investigated for the development of REOs as high-*k* dielectric. The deposition method of REOs can be segmented into two major categories which includes chemical vapor deposition (CVD) and physical vapor deposition (PVD). The CVD has been classified into two major techniques which includes mainly atomic layer deposition (ALD) and metal organic chemical vapor deposition (MOCVD). Meanwhile, the PVD includes several deposition techniques which includes magnetron sputtering, electron beam evaporation, thermal evaporation and pulsed laser deposition (Chin et al., 2010; Goh et al., 2017a; He, Liu, et al., 2014; Houssa et al., 2006; Wong & Cheong, 2010).

2.9.1 Chemical Vapor Deposition (CVD)

The deposition of thin film using ALD basically utilizes chemical gas phase of two or more precursors. These precursors are deposited into the substrate surface through saturation reaction deposition between precursors substrate material by utilizing alternative oxidation in a subsequent process causing different monolayers of the reactant precursors within the reaction chamber. Subsequently, deposition of dielectric oxide occurs by the regulation of surface controlled growth cycles. Primarily, an excessive amount of complementary gaseous precursors is exposed to the substrate surface which introduces complementary monolayer of reactant. A controlled temperature and gas flow is regulated in such way until the reactant precursors monolayer could chemically absorb the maximum substrate surface so that there is no further expansion of the complementary precursors until the anticipation of the second precursor. Before the exposure of second precursors the residual of complementary precursors gas is taken out by purging an inert gas into reaction chamber. In the last stage second precursor is purged into the chamber which

reacts with first reactant monolayer and chemically saturates the reactant to form the desired dielectric thin film. Subsequently, again the second precursor is brought out from the reaction chamber by purging an inert gas which completes the ALD process (Goh et al., 2017a; Jo, Ha, Park, Kang, & Kim, 2006; W. H. Kim, Maeng, Moon, Myoung, & Kim, 2010; Paivasaari et al., 2006). The ALD method is also defined in various names such atomic layer epitaxy (ALE) and atomic layer chemical vapor deposition (ALCVD) (Wong & Cheong, 2010). Regarding ALD process the precursor should be non-toxic and volatile, maintain high-level of purity, certainly not participate in reactions at gas phase and the oxide should be free of any etching and self-decomposition (Leskela & Ritala, 2003). Utilization of thermal and plasma enhanced ALD could produce much better quality of thin film with less impurities, good interface and electrical properties (Khosla, Schwarz, Funk, Guguieva, & Schulze, 2021; W. H. Kim et al., 2010; Zhao et al., 2018). The MOCVD deposition method is another technique under CVD for deposition of thin film on substrate. This process utilizes both fundamental principles of ALD and CVD. The primary distinction between ALD and MOCVD is that during MOCVD process metalorganic compounds with hydrate or alkyl group are used as precursor into the reactor chamber. In this method usage of liquid precursor is beneficial for avoiding unwanted particle formation where the molecules of decomposed liquid precursors are to be transferred into heated substrate by utilizing (H₂ or N₂) as gas carrier and afterwards surface reaction takes place to form thin films (Chin & Cheong, 2011; Houssa et al., 2006). Both the ALD and MOCVD are widely used since these methods are advantageous to for uniform thin film while enables to control the oxide film thickness accurately. These methods also hiders the growth of complex atomic configuration because the precursors could control self-constraining growth process. Furthermore, these methods also allow to grow higher-level film density covering a wide region of depositions with higher deposition rates and control the film composition (Goh et al., 2017a; He et al., 2011; Jo et al., 2006; Mallem et al., 2019).

Nevertheless, these methods are subjected to some limitations which includes availability, stability, and volatility of molecular precursor, For the decomposition of metal organic precursor relatively higher temperature and a post deposition annealing are essential. Accordingly, the electrical characteristics are hampered due to the presence of large volume of detrimental carbon (Goh et al., 2017a; He, Liu, et al., 2014; Kosola, Päiväsaari, Putkonen, & Niinistö, 2005). Wong and Cheong, (2010) encapsulated three major concerns of ALD (1) it's complicated and challenging to transfer particles from source to the deposited film since the dimension of particles is very delicate (2) poor nucleation on hydrogen terminated silicon, and (3) residual defects at the film edge (Wong & Cheong, 2010). For instance, Nd₂O₃ dielectric film was left over with a substantial volume of residual carbon in its film region during ALD procedure (Kosola et al., 2005).

2.9.2 Physical Vapor Deposition (PVD)

The pulsed laser deposition is a PVD technique using laser ablation to deposit thin films on the substrate. In this method laser beam with high energy density irradiates target source which ablates away some particles of the material in vaporized form that ultimately transported towards the substrate surface. The radioactivity of repetitive laser pulse subsequently generates reoccurring vapor clouds which causes to form a new material on the substrate surface (Goh et al., 2017a; Wong & Cheong, 2010). However, the kinetic energy and specific ratio of the ablated vaporized particles depends on wavelength, intensity, and pulse width of the laser. During the PLD process formation of thin film includes three phases which are (i) interaction of laser radiation through the target source (ii) dynamic aspects of excision materials (iii) sublimation of extracted source material into substrate and formation of the desired thin film (Chin & Cheong, 2011). PLD is a simple technique to form thin oxide film which could serve variety of film compositions and structures. However, PLD deposited REOs tends to increase surface roughness and crystallization occurs with increment of substrate temperature (Balakrishnan et al., 2013; Constantinescu, Ion, Galca, & Dinescu, 2012). Electron beam evaporation is one more common deposition techniques that has been widely utilized to grow thin solid film on substrate surface. In this method an electron beam is employed as an energy source at high-level vacuum ambient which heats up the target material for producing vaporized form of the target material and then condensed on the surface of substrate to grow thin solid film (Chin & Cheong, 2011; C. Yang, Fan, Qiu, Xi, & Fu, 2009). It has been reported that this method could form smooth thin oxide films which reveals promising electrical properties. Electron beam evaporation can produce ultrathin and smooth films with excellent electrical properties (Sen et al., 2007). This method also allows to form compact and densified oxides with low stress (Goh et al., 2017a; C. Yang et al., 2009). Nevertheless, this method is not compatible for conventional manufacturing procedure.

Sputtering is the simplest and frequently used procedure of depositing thin film under PVD. This is a deposition procedure where the solid source materials known as target source discharge plasma particles towards the solid substrate material positioned inside a vacuum chamber. At the beginning of this process the target source is applied to a negative bias which discharge plasma particles. Accordingly, positively charge gas ion is being produced across the plasma domain. Consequently, the negatively biased target material source attracts the positive gas ions incredibly at high velocity from the plasma domain. Therefore, ejection of target material source occurs due to an exchange of momentum amongst the target source particles caused by the surface bombardment through energetic particles. (Chin & Cheong, 2011; Goh et al., 2017a; Wong & Cheong, 2010). The magnetron sputtering basically involves radio frequency (RF) sputtering and direct current (DC) sputtering. The type to be used depends on the electrical conductivity of target material. The target source material with low electrical conductivity utilizes RF sputtering while the high electrical conductivity target source material employs DC sputtering (Goh et al., 2017a; Robertson, 2004). RF magnetron sputtering is a widely used straightforward method for high-k deposition which is broadly available to deliver high growth rate during deposition, deposit smooth surface good quality adhesion among the oxide film and substrate material (Jagadeesh Chandra, Choi, Uthanna, & Mohan Rao, 2010; Kaya, Yilmaz, Karacali, Cetinkaya, & Aktag, 2015). Although deposition of thin film using the sputtering from target material (such as metal or metal oxide) is quite easier and straightforward than other methods, but the problem is oxygen and argon flow is used for the formation oxide in this method could possibly oxidize substrate material leading thicker interfacial layer of native oxides (Pampillon et al., 2011; Pan & Huang, 2010). Apart from that sputtering does not provide good quality of coverage on complex structure since this process only deposits line up of spots (Robertson, 2004). Several studies reported that using pure metal target source during RF sputtering at an argon ambient subsequently following thermal oxidation could be a superior choice for thin film formation into MOS gate stack (Lei et al., 2017; Wong & Cheong, 2012a; Wong et al., 2021). This is because combination of pure metal RF sputtering with subsequent thermal oxidation process could provide formation of stochiometric metal oxide while controlling unwanted interfacial layer.

2.10 Performance of Commonly Used REOs/Ge Stack

Rare earth oxides (REOs) have been found to be thermodynamically stable with Ge due to vigorous reaction between REOs with the Ge interface. The catalytic oxidation of Ge atom with REOs promotes the spontaneous growth of rare earth germanate (RE -O – Ge) leading to low interface trap densities, $D_{it} \sim 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$ (Mitrovic et al., 2014; Zhao et al., 2018). Moreover, rare earth oxides do not involve any pre-deposited interfacial layer between REOs and Ge since they form stable germanate layer during deposition or annealing process (Dimoulas et al., 2007; W. Liu et al., 2022). The most used high-k transition metal oxides, HfO₂, ZrO₂ and Al₂O₃ reveals drawbacks in electrical properties and thermodynamic stability of the gate oxide (Kamata et al., 2005; H. D. Kim & Roh, 2006; X. Yang et al., 2014). Hence, REOs have drawn major attention in the highk dielectric researchers. Lu et al. (2014) has been demonstrated that REOs provide more attractive properties for M-GeO₂/Ge (i.e., M= Al, Sc, Y, La, and Hf) gate stack in terms of aggressive scaling, electrical properties, and thermodynamic stability. Fig. 2.16 shows that rare earth metal-based compound Y–GeO₂ reveals the lowest $D_{it} \sim 1 \times 10^{11} \text{ eV}^{-1}$ cm⁻² while Hf– GeO₂ yielded the highest, $D_{it} \sim 10^{12} \text{ eV}^{-1} \text{ cm}^{-2}$ (Lu et al., 2014). The interface trap density found for Y-GeO₂ is quite close according to the state of art GeO₂/Ge interface formulated by high pressure oxidation (C. H. Lee, Nishimura, Nagashio, Kita, & Toriumi, 2011). The most used rare earth oxides as high-k dielectric on Ge substrate are La₂O₃ (Dimoulas et al., 2010; Kanashima et al., 2015; Q. Liu, Fang, Liu, Tan, & Chen, 2014; Mavrou et al., 2007; J. Song et al., 2009; Zhao et al., 2018; Zhao, Liu, Wang, Wang, & Wang, 2019), Y₂O₃ (Bethge et al., 2014; D. G. Kim et al., 2021; Mitrovic et al., 2014; Nishimura et al., 2011; Seo et al., 2017; Zimmermann et al., 2016), CeO₂ (Brunco et al., 2007; Dimoulas et al., 2007), Gd₂O₃, (Evangelou et al., 2009), and Tm₂O₃ (Žurauskaitėa et al., 2018).



Figure 2.16: Comparison of interface trap densities among transitional metal and rare earth metal deposited on GeO₂/Ge interface. (Lu et al., 2014).

2.10.1 La2O3/Ge Gate Stack

In work of J. Song et al., (2007) La₂O₃/Ge gate stack developed by electron beam evaporation with post deposition annealing (PDA) in 5% (O₂+N₂) ambient at (500 – 400) °C. A gate leakage current density, $J_g \sim 10^{-4}$ A cm⁻² by introducing an interfacial La–O–Ge. Although formation of additional GeO_x during the annealing process induces higher D_{it} in order of magnitude 10^{12} eV⁻¹ cm⁻² which impacted to higher hysteresis. In work of Mavrou, Galata, et al., (2008) a promising *C-V* characteristic with negligible hysteresis and reduced accumulation frequency dispersion with lower D_{it} in order of magnitude 10^{11} eV⁻¹ cm⁻² but with high *EOT* ~ 5.7 nm has been achieved while utilizing La₂O₃ deposited through MBD at temperature above 200 °C and subsequent PDA. However, it was found that deposition temperature below 200 °C degrades the electrical and interface properties. It is suggested that electrical properties can be improved using high deposition temperature or by annealing. The improvement of La₂O₃ based Ge stack has

been attributed due to four fold configuration of La within GeOx which changes the surface chemistry through saturation of Ge dangling bonds by creation of interfacial LaGeO_x. However, formation of LaGeO_x exhibited low- k values of 9, and high J_g of 10^{-1} A cm⁻ ² at $V_g = +1$ V which hinder the MOS scaling (Dimoulas et al., 2010; Mavrou, Galata, et al., 2008). The reduction of k value has also been predicted due to the hygroscopic nature of La₂O₃. In the same study using combination HfO₂ with 1 nm La₂O₃ improved gate leakage current density up to 10^{-8} A cm⁻² and k value up to 17 but expenses high interface trap density. Combination of ZrO_2 and La_2O_3 through MBE provides higher k of around 40, leading to EOT as low as 1.2 nm but increases the interface trap density as well (Mavrou, Tsipas, et al., 2008). Q. Liu et al., (2014) deposited 10-nm La₂O₃/Ge by RF magnetron sputtering following rapid thermal annealing (RTA) in O₂ ambient at 200 °C. In this work the conduction band offset, $\Delta E_c \sim 1.75$ eV, valance band offset, $\Delta E_v \sim$ 2.76 eV and energy band gap $E_g \sim 5.18$ eV was determined using krauts method and Xray photoelectron spectroscopy. An improved $\Delta E_c \sim 2$ eV and $E_g \sim 5.18$ eV has been determined for La₂O₃/Ge at a deposition temperature of 400 °C by MBE (Mitrovic et al., 2014). However, MBE is not a conventional thin deposition method for MOS fabrication.

2.10.2 Y2O3/Ge Gate Stack

Mitrovic et al., (2011) demonstrated that Y₂O₃ is more Ge friendly than La₂O₃ which exhibits a band gap of $E_g \sim 5.7$ eV and CBO, $\Delta E_c \sim 2.3$ eV with the Ge interface and reduced the leakage current below 10⁻⁶ A cm⁻² using MBE deposition method at temperature 400 °C (Mitrovic et al., 2014). A high electron mobility, μ_e of 1480 cm²V⁻¹s ⁻¹ and $D_{it} \sim 10^{11}$ eV⁻¹ cm⁻² has been attained recently by employing high pressure oxygen annealing (Nishimura et al., 2011).

In work of Bethge et al., (2014), Y₂O₃/Ge gate stack was developed by employing ALD with PDA exhibiting D_{it} as low as 10¹⁰ eV⁻¹ cm⁻² with very low $J_g \sim 5.7 \times 10^{-4}$ A cm⁻² at relatively higher thermal budget of 600 °C but with the price of higher capacitance equivalent thickness (CET) of 10 nm and hysteresis of 140 mv. The improvement in interface trap density and gate leakage current density for Y₂O₃/Ge has been attributed to penetration of Y atom into oxygen deficient GeO₂ causing reduction of GeO_x defects by the growth interfacial YGeO_x (Bethge et al., 2014; Nishimura et al., 2011). Meanwhile, YGeO_x/Ge/Si stacked has shown improved performance at high temperature of 750 °C through thermal oxidation which induced small hysteresis < 50 mv but with relatively low-*k* of 10.8 that impeded the scaling of the device (Seo et al., 2017). Typically, the dielectric constant, *k* of Y₂O₃/Ge has been reported ranging between 11–18 (Mitrovic et al., 2014; Nishimura et al., 2011).

2.10.3 CeO₂/Ge Gate Stack

Dimoulas et al., (2007) demonstrated CeO₂ gate stack by Molecular Beam Deposition (MBD) which revealed moderate-surface passivation with the reaction of Ge substrate inducing relatively higher $D_{it} \sim 10^{11} - 10^{12}$ eV⁻¹ cm⁻². However, this CeO₂/Ge gate stack showed small but visibly noticeable frequency dispersion and hysteresis in the *C-V* curve and revealed high leakage current density (> 0.1 A cm⁻² at $V_g = +1$ V). The high leakage current density could be reduced to $J_g \sim 10^{-7}$ order magnitude along with higher dielectric constant $k \sim 20.5$ when rare earth metal Ce was doped with HfO₂ using optimum mixing condition of Ce: HfO₂ = 1:4 for CeHfO₂/Ge gate stack developed by ALD with subsequent PDA process. However, the defect density and hysteresis were quite high where $D_{it} \sim 10^{13}$ eV⁻¹ cm⁻² which requires a controllable doping ratio (Maeng et al., 2014).

2.11 Sm₂O₃ and Ho₂O₃ as Alternative Dielectric

Although high-k materials of (TMOs and REOs) emerged to be potential contender for upcoming MOS technology but each of TMOs and REOs are still subjected to their own deficiencies in terms of gate leakage, interface trap densities, dielectric constant, EOT and thermodynamic stability (Goley & Hudait, 2014; Kamata, 2008; Mavrou, Galata, et al., 2008; Oh et al., 2015; Venkata Rao et al., 2018). Additionally, hygroscopic nature of REOs is an additional limiting factor on the way of MOS scaling which lowers dielectric constant and enhances equivalent oxide thickness (Chin et al., 2010; Goh et al., 2017a). Accordingly, it is imperative to seek for an appropriate high-k gate oxide material for Ge based gate stack which can compete the superior properties of conventional SiO₂/Si. Amongst the REOs Sm₂O₃ can be a prospective contender for MOS application, since this high-k material possesses larger $k \sim 7 - 15$, higher energy bandgap $E_g \sim 4.33$ eV, CBO ~ 2.7 eV with an electrical breakdown field $E_{Bd} \sim 5 - 7$ MV cm⁻¹, low frequency dispersion, low leakage current, with high gate capacitance with Si substrate (Chen et al., 2013; Chin & Cheong, 2011; Dakhel, 2004; Goh et al., 2016a, 2017a). In addition, Sm_2O_3 possess the second highest position of k value below La_2O_3 and less hygroscopic nature due to high electronegativity and large lattice energy in counter to other lanthanide oxide REOs as shown in earlier Fig. 2.14. So far, numerous studies were conducted on Sm₂O₃/Si using several deposition methods of CVD and PVD which revealed encouraging electrical and thermodynamic features (Chin & Cheong, 2011; Chin et al., 2010; Constantinescu et al., 2012; Goh et al., 2017a; Kaya et al., 2015; Paivasaari, Putkonen, & Niinisto, 2005; Pan, Huang, You, & Yeh, 2008). Table 2.3 demonstrates the dielectric constant and leakage current density for Sm2O3/Si gate stack different deposition methods (S. Jeon et al., 2001; Paivasaari et al., 2005; Pan & Huang, 2010; D. Yang, Xue, & Devine, 2003).

In works of Paivasaari et al., (2005) 50-nm Sm₂O₃ has been deposited on p-type Si using ALD that revealed low leakage current density, $J_g \sim 1.1 \times 10^{-8}$ A cm⁻² at electrical breakdown field $E_{Bd} \sim 0.32$ MV cm⁻¹ and $k \sim 10$. The improvement of leakage current density has been attributed to the smooth surface and to the growth of polycrystalline cubic phase of Sm₂O₃. The surface roughness was measured to be around root mean square, RMS ~ 1.2 nm. A smooth and amorphous surface of Sm₂O₃ was also developed using both PLD and RF sputtering at room temperature where a low surface roughness around RMS ~ 0.31 nm for sputtering was recorded (Constantinescu et al., 2012; Pan et al., 2008). However, an improved k value can be achieved from PLD , and electron beam evaporation as shown in Table 2.3 (S. Jeon et al., 2001; D. Yang et al., 2003).

Several researchers have deposited Sm₂O₃ thin film using RF magnetron from Sm₂O₃, or pure metal target source followed by thermal oxidation or post deposition annealing (PDA) to improve the physical and electrical properties Sm₂O₃/Si gate stack (Chen et al., 2013; Chin & Cheong, 2011; Goh et al., 2016a, 2016b; Goh, Haseeb, & Wong, 2017b; Kaya et al., 2015; Pan & Huang, 2010; Pan et al., 2008). In works of (Pan et al., (2008) Sm₂O₃/p-type Si gate stack combination of RF magnetron and RTA in O₂ ambient at 700 °C induced a low *EOT* ~ 2.75 nm and *k* ~ 9.93 at low *J*_g of 1.1 × 10⁻⁸ A cm⁻² at *V*_g = -1V, and average interface trap density $D_{it} \sim 1.68 \times 10^{11}$ eV⁻¹ cm⁻² at mid-gap. A well-shaped *C-V* curve with high capacitance and low hysteresis was also achieved. The improvement has been attributed to the formation of uniform cubic polycrystalline Sm₂O₃/Si film using reactive sputtering of pure Sm in different Ar/O mixture following with subsequent PDA at various temperatures in O₂ ambient. The sample developed using O/Ar ratio 15/10 at 700 °C improved the dielectric constant up to *k*

~ 14.3 with $J_g \sim 10^{-8}$ A cm⁻² at $V_g = -1$ V. It was found that formation of more uniform structure of Sm₂O₃ and improved recrystallization at this deposition condition enables to increase dielectric constant and reduced trapped charges within the dielectric surface. Goh et al., (2017a) summarized the electrical features of the leakage current density with respect to electrical breakdown field (*J-E*) for Sm₂O₃/Si films for several film deposition techniques as shown Fig. 2.17 (Goh et al., 2017a). According to Fig. 2.17, Sm₂O₃/Si developed using RF magnetron sputtering shows superior leakage current and higher electrical breakdown field.

However, investigation relating Sm_2O_3 /Ge stack for MOS application is very limited and scarce in counter to Sm_2O_3 /Si gate stack. In work of C. C. Lin et al., (2014), Sm_2O_3 /Ge stack has been investigated by combining e-beam evaporation with subsequent furnace annealing process in O₂ ambient at 400 °C, which corresponded promising outcome of high dielectric constant $k \sim 17.1$ inducing very low $EOT \sim 0.68$ nm and low interface trap density of $D_{it} \sim 5.10 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$. The reduction of interface traps has been attributed to the four fold configuration of Sm atoms which saturates the vacant locations of Ge dangling bonds inside the GeO₂ network by composing Sm–O–Ge interfacial layer. Although, the report outlined some potential findings, but this gate stack suffers from high gate leakage current, and several other aspects of Sm₂O₃/Ge interface are still unexplored which requires further investigation. Therefore, Sm₂O₃ on Ge substrate has been selected for this investigation for the realization of high-k/Ge MOS capacitor.

Deposition pro-	Thickness	Leakage current	Dielectric
cess	(nm)	Density A/cm ²	Constant k
ALD	50	1.11×10^{-8}	10
Electron Beam Evaporation	25	9.5×10^{-5}	11.4
PLD	111	1.63×10^{-6}	12.8
RF Magnetron Sputtering	7-8	10 ⁻⁷	14.3

Table 2.3: Leakage current density and dielectric constant of Sm₂O₃/Si gate stack for different deposition techniques (S. Jeon et al., 2001; Paivasaari et al., 2005; D. Yang et al., 2003).



Figure 2.17: Review of leakage current density against electrical breakdown voltage for Sm₂O₃/Si gate stack for different deposition techniques (Goh et al., 2017a).

Among the REOs, Ho₂O₃ could be another prospective contender for realization of High-k/Ge because it befalls on the selection criterion of high-k dielectric. Ho₂O₃ owns a dielectric constant value $k \sim 13.1$, large enough bandgap energy, $E_g \sim (5.3 \text{ eV})$ and high enough conduction band offset (CBO) of (2 eV) with respect to Si substrate (Paivasaari et al., 2005; Pan, Chang, et al., 2010; Pan & Huang, 2011). In addition, Ho₂O₃ possesses the highest electronegativity and the highest lattice energy among all other eligible REOs as shown in Fig. 2.14 indicating the least hygroscopic nature of Ho₂O₃ which would be favourable for implementing Ho₂O₃/Ge gate stack within the expected MOS scaling limit (Goh et al., 2017a; Iwai et al., 2002). Based on superior properties of Ho₂O₃ has been integrated as dielectric insulator material in sensor application for electrolyte-insulatorsemiconductor (EIS) device yielding higher sensitivity and pH sensing efficiency (Pan, Huang, Lin, & Wu, 2010). Despite of such attarctive features the investigation on Ho₂O₃ based MOS is very scarce. In work of Pan, Chang, et al., (2010) a 11nm Ho₂O₃ /p-type Si stack has been developed utilizing RF sputtering with a subsequent RTA in O₂ ambient at temperature range (600 - 800) °C. A well behaved C-V curve with less hysteresis as shown Fig. 2.18 has been obtained which induced $k \sim 10.1$, and low interface trap density of $D_{it} \sim 3.10 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$ at 700 °C. The low interface trap density indicated thermodynamic stability of Ho₂O₃ during deposition process. A smooth surface has been formed with a roughness around RMS ~ 0.86 nm which aided in reducing leakage current density, $J_g \sim 10^{-8}$ A cm⁻² order magnitude at $V_g = -1$ V as in Fig. 2.19. However, one order magnitude higher leakage current density was found at 700 °C in counter to 800 °C RTA samples which has been attributed to the increment of surface roughness due to crystallization or grain decomposition. Paivasaari et al., (2005) showed that, Ho₂O₃/Si film grown using ALD at 300 °C induces dielectric constant of 9.9 with low leakage current density of $J_g \sim 1.2 \times 10^{-8}$ A cm⁻² at $E_{Bd} \sim 2.06$ MV cm⁻¹. However, the surface roughness might be affected by the contaminants (carbon, sodium, and hydrogen) from

reactor and precursor which limits the usage of this procedure. In works of Castan et al., (2015), HoTiO_x metal insulator metal (MIM) was developed through ALD at 300 °C following RTA process in N₂ ambient. It is observed that, HoTiO_x film with higher Ho content possess a poor *C-V* characteristic, but the gate leakage is less. This has been attributed to better re-crystallization of Ho-O content during annealing. However, a well behaved *C-V* has been reported for 16 nm-HoTiO₃/Si gate stack fabricated from reactive sputtering and subsequent annealing process. It was found that, well crystallized HoTiO₃ was formed at 700 °C inducing low leakage current density $J_g \sim 1.2 \times 10^{-8}$ A cm² and hysteresis < 9 mv (Pan, Yen, Hu, & Sheng, 2010). Till to date there has been no pertinent study for integration of Ho₂O₃/Ge based MOS device.



Figure 2.18: *C-V* curve for Ho₂O₃/Si gate stack for diiferent annealed temperatures (Pan, Chang, et al., 2010).



Figure 2.19: Leakage curren density (*J-V*) with respect to gate voltage for Ho₂O₃/ p type Si gate stack annealed at different RTA temperatures (Pan, Chang, et al., 2010).

2.12 Role of Nitrogen (N) for Ge passivation

Surface nitridation procedure using N₂ and/or NH₃ has been comprehensively examined in several studies for the stabilization of high-*k* oxide/Ge (Dushaq, Rasras, & Nayfeh, 2016; Fukuda et al., 2011; Kutsuki et al., 2009; Maeda et al., 2004, 2006; Mallem et al., 2019; Xie et al., 2012). Since, single layer of Ge–N/Ge–O–N possess low dielectric constant and so more investigation was focused on using Ge–N/Ge–O–N as an interfacial layer in conjunction with high-*k* dielectric layer. It was found that incorporation of N atom into GeO₂ forming Ge–O–N/Ge–N interfacial layer improves dielectric constant and thermal stability of the gate stack (Bhatt, Chaudhuri, Kothari, Nainani, & Lodha, 2013; Dushaq et al., 2018; Q. L. Li et al., 2011; Maeda et al., 2007; Otani et al., 2007). Nevertheless, improvement of electrical properties was found for directly grown Ge–N interfacial layer on Ge substrate topped with High-*k* gate stack.
Otani et al., (2007) reported that for Ta₂O₅/Ge₃N₄/Ge gate stack inserting a 2 nm thick Ge-N interfacial layer beneath Ta₂O₅ exhibits low $D_{it} \sim 4.0 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-1}$ at midgap while more improvement of $D_{it} \sim 1.84 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-1}$ at mid-gap was reported with slightly thick 2.4 nm Ge₃N₄ for HfO₂/Ge gate stack (Maeda et al., 2007; Otani et al., 2007). However, the key perspective is that creation of Ge–N/Ge–O–N interfacial layer beneath the dielectric layer with incorporation of N atom improves the thermodynamic stability, dielectric constant, and electrical properties of High-k/Ge interface. It has also been reported that blocking capability of Ge-N/Ge-O-N hinders the inter-diffusion and Ge surface oxidation which is advantageous removing Ge dangling bonds and blocking the formation unstable GeO_x contents in some extent and thus improving the interface quality (Kutsuki et al., 2009; Maeda et al., 2006). Furthermore, incorporation N during REOs has been considered an effective way to passivate GeO_x which improves the thermal stability and electrical properties of the gate stack. It has been reported that, N incorporation into La₂O₃ lessened the formation of GeO_x and germanate as a result thermodynamic stability increased leading a low interface traps up-to $D_{it} \sim 6.34 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-1}$ ¹ with a leakage current density $J_g \sim 10^{-4}$ A/cm² at $V_g = 1$ V (Z. X. Cheng, Liu, Xu, Huang, & Lai, 2016). Typically, N₂O and NO gases are extensively utilized in semiconductor industry during thermal oxidation/nitridation process. N2O was used for this investigation due to its higher activation energy and less toxicity than NO. Several reported literatures (Lei et al., 2017; Wong & Cheong, 2011b, 2012a) evidenced that MOS gate stack established from the combination of metal deposition utilizing radio frequency (RF) magnetron sputtering with subsequential thermal oxidation/nitridation flowing N₂O gas could enhance the dielectric film quality and improve the electrical features.

CHAPTER 3: RESEARCH METHODOLOGY

3.1 Research Overview

This chapter focuses on describing and illustrating the experimental methods and materials applied in this research. Typically, high-*k* gate oxide on Ge should maintain all the six requirements listed in section 2.5 for MOSFET operation enduring thermal budget at least 400 °C for 10 min while using thermal oxidation and nitridation (Goley & Hudait, 2014; Kamata, 2008). Moreover, several previous reports have evidenced that variation in deposition condition in terms of oxidation/nitridation temperatures and durations could significantly impact the interface morphology, electrical and thermodynamic properties of MOS capacitors (Hetherin et al., 2017a; Lei et al., 2018). Hence, the framework for the experimental steps and designated variables were employed for this investigation according to Fig. 3.1. The experimental and sample preparation methodology are segmented into three sections:

- (i) Required Materials.
- (ii) Experimental Methodology.
- (iii) Characterization techniques.

The dimensions of the required materials are segmented into four sub sections as follows:

- (i) Dimensions of Ge substrate material.
- (ii) Chemicals applied for Ge substrate and quartz tube cleaning.
- (iii) Chemicals and gases applied for Sm, Ho and Al sputtering process.
- (iv) Gases applied during thermal oxidation/nitridation process.

The sequential steps required for sample preparation are shown in Fig. 3.2. The experimental method for sample preparation is mainly segmented in four sequential steps (i) Ge substrate cleaning (ii) Sputtering of metal (Sm) and metal (Ho) (iii) thermal oxidation/nitridation and (iv) top electrode and back contact deposition for MOS structure.

The physical and electrical characterization used in this research is shown Fig. 3.3. The physical properties are analysed using four physical characterization techniques which involves (i) X-Ray Diffraction (XRD) (ii) X-Ray Photoelectron Spectroscopy (XPS) (iii) RAMAN Spectroscopy (iv) High Resolution Electron Transmission Microscopy (HRTEM). The electrical characterization consists of two fundamental features which are (i) Current – Voltage (*I-V*) measurements (ii) Capacitance – Voltage (*C-V*) measurements.



Figure 3.1: Experimental design and variables for investigation.



Figure 3.2: Sequential Steps for Sample Preparation.



Figure 3.3: Overview of characterization techniques.

3.2 Required Materials

3.2.1 Dimension of Substrate Material

The substrate material using for this research is Germanium (Ge). The Ge wafers used in this research following the dimension as n-type, antimony (Sb) doped Ge:Sb (100) orientation with resistivity of $(0.005 - 0.02) \Omega$ cm and with a one-sided polished surface. The Ge wafers were supplied by Wafer World, Inc.

3.2.2 Chemical Required for Substrate and Quartz Tube Cleaning

Chemical that has been used for Ge substrate cleaning is hydrofluoric acid (HF). For the dip-cleaning process HF solution was prepared with ratio maintaining (HF: H_2O = 1: 50). On the other hand, Acetone has been utilized as chemical material during the cleaning of quartz tube of the thermal furnace. The details parameter of chemical materials utilized for Ge substrate and quartz tube cleaning is given in Table 3.1.

 Table 3.1: Parameter details of chemical utilized for Ge substrate and quartz tube cleaning.

No	Chemical	Chemical Formula	Assay	Supplier /(CAS)
			(%)	No
1	Hydrofluoric Acid	HF	49%	R & M chemicals
				7664-39-3
2	Acetone	(CH ₃) ₂ CO	>99%	R & M chemicals
				67-64-1

3.2.3 Chemical, Gas and Materials for Sputtering Process

Rare earth metals of pure Samarium (Sm) and pure Holmium (Ho) has been utilized for sputtering process as target source for high-*k* deposition. Aluminum (Al) has been used as target source in the sputtering process during deposition of top electrode and bottom contact. The inert Argon (Ar) gas was utilized as sputtering gas. Acetone has been utilized as chemical agent for cleaning the sputter chamber and substrate holder. The details parameter of chemical, gas and materials utilized during sputtering procedure is given in Table 3.2.

No	Chemical and Mate- rial	Chemical Formula	Specifications	Supplier /(CAS) No
1	Samarium	Sm	Purity: 99.9%, Diame-	
2	Holmium	Но	ter: 101.6 mm, Thick-	Kurt J. Lesker
3	Aluminum	Al		
3	Acetone	(CH ₃) ₂ CO	>99%	R & M chemicals 67-64-1
4	Argon	Ar	Purity: 99.99%	Gaslink 7440-37-1

 Table 3.2: Parameter details of chemical, gas, materials used during sputtering process.

3.2.4 Required Gases for Sputtering Process

Two types of gases have been utilized during the thermal oxidation/nitridation process which are nitrous oxide (N_2O) and inert Argon (Ar). The detail specifications of these gases in terms of chemical formula, purity, and supplier have been given in Table 3.3.

No	Chemical	Chemical Formula	Purity (%)	Supplier /(CAS) No
1	Nitrous Oxide	N ₂ O	99%	Gaslink
				10024-97-2
2	Argon	Ar	99%	Gaslink
				7440-37-1

 Table 3.3: Detail specifications for gases used during thermal oxidation/nitridation.

3.3 Experimental Procedure

3.3.1 Ge substrate Dip cleaning

In first step the Ge wafer is sliced into pieces at a dimension $1 \text{ cm} \times 1 \text{ cm}$ using diamond cutter. In second step each of the sample pieces underwent for wet cleaning treatment. During the wet cleaning process the Ge substrates were dipped into hydrofluoric acid, HF solution (1:50 HF: H₂O) and deionized water DI water for 15s to eradicate native GeO₂ and any other contaminants. Sequentially, in the last step the wet substrates were dried using N₂ air gun. During the wet cleaning process, it is expected to remove native GeO₂, and any other additional impurities present in the substrate.

3.3.2 Sputtering of Rare Earth Metal on Ge substrate

Physical Vapor Deposition (PVD) radio frequency (RF) sputtering system (TF 450) model equipment has been utilized for sputtering process in this research. Before initiating the sputtering process the parts of sputtering system including the sputter chamber, target source holder, substrate holding plate, and shutter were cleansed using acetone to eliminate the possibility of contamination from these parts.

For fabrication of Sm₂O₃ gate stack pure samarium (Sm) metal and for Ho₂O₃ gate stack pure Holmium (Ho) metal was placed into the target holder as source material. Subsequently, Ge substrate sample pieces were laid on the substrate holding plate and then the gate of the sputtering chamber was shutdown. The distance of (Sm or Ho) target source with Ge substrate holder plate was about 0.2 meter. Fig. 3.4 displays the schematic of (RF) sputtering system. In summary, during the metal sputtering stage metallic rare earth metal (Sm) or (Ho) with thickness of 3-nm is being sputtered on the cleaned Ge substrate through radio frequency (RF) sputtering system (TF450) from rare earth metal (Sm or Ho) target source. The radio frequency (RF) sputtering system was regulated at

the RF power of 170 W, base pressure of 3×10^{-5} mbar, sputtering time of 75 s and inert Argon (Ar) gas flow rate of 25 cm³/min during the sputtering process



Figure 3.4: Schematic of PVD RF sputtering system.

3.3.3 Thermal Oxidation and Nitridation Process

In this research Carbolite CTF model no: (12/65/550) tube furnace has been utilized for conducting thermal oxidation and nitridation process. Thermal oxidation /nitridation (oxy-nitridation) processes were carried in N₂O gas ambient, for both sputtered (Sm) and sputtered (Ho). Before initiating the thermal oxidation and oxidation/nitridation processes, quartz tube and sample carrier quartz boat were cleaned applying acetone (CH₃)₂CO. Then, the samples standing in the quartz boat was placed into the middle location of horizontal CTF tube furnace as displayed in Fig.3.5. Subsequently, the desired gas is flowed when the furnace reached set temperature according to experimental variables. The section below demonstrates the design for each oxidation and oxidation/nitridation setup. To explore the impact of various oxidation/nitridation temperatures for sputtered Sm on Ge substrate, thermal oxidation/nitridation were conducted at different temperature 300 °C, 400 °C, 500 °C and 600 °C using CTF tube furnace in N₂O ambient. The samples of sputtered Sm on Ge substrate were placed into the middle of horizontal tube furnace and heated up from room temperature (heating rate of 10 °C/min) to the experimental setting temperature for each sample in an Ar ambient. When the furnace tube has reached the experimental set temperature (300 °C, 400 °C, 500 °C and 600 °C) N₂O gas was purged into the horizontal tube with a gas flow rate of 150 mL/min for 15 min and Ar gas flow is turned off. The furnace tube was cooled down to room temperature and then the samples were taken out.

To explore the influence of various oxidation/nitridation durations for sputtered Sm/Ge and sputtered Ho/Ge, thermal oxidation/nitridation was conducted at different durations of 5 min, 10 min, 15 min and 20 min at 400 °C using CTF tube furnace in N₂O ambient. Samples of sputtered Sm/Ge were placed into the center of furnace tube for fabricating Sm₂O₃ stack. On the other hand, for fabricating Ho₂O₃ gate stack samples of sputtered Into the center of furnace tube. Subsequently, the furnace tube with the samples of (sputtered Sm/Ge or Ho/Ge) was heated up from room temperature to a set temperature of 400 °C (heating rate of 10 °C/min) in an argon (Ar) ambient. When the set temperature was achieved, Ar gas supply has been turned off while N₂O gas purged into the horizontal tube using flow rate of 150 mL/min for various oxidation/nitridation durations (5, 10, 15, and 20 min). When the furnace tube cooled down to room temperature and then the samples are brought out.



Figure 3.5: Schematic of horizontal furnace tube setup for thermal oxidation and nitridation.

3.3.4 MOS Capacitor Test Structure

To examine the electrical properties of dielectric films, MOS capacitors test structures with area 500 μ m × 500 μ m was fabricated. Firstly, 100 nm Aluminum (Al) metal electrode was formed by sputtering Al target on the oxidized/nitrided film through a shadow mask using Physical Vapor Deposition (PVD) radio frequency (RF) sputtering system (TF 450) as presented in Fig. 3.6. Afterwards, 100 nm ohmic contact is formed on the backside of the Ge substrate by sputtering Al without shadow mask. The sputtering process of Al follows the same sequential steps as described for metal (Sm) and (Ho) sputtering procedure shown in earlier Fig. 3.4. For electrode deposition and ohmic contact formation metal Al was placed into the target holder as source material. The radio frequency (RF) sputtering system was regulated at the RF power of 170 W, base pressure of 4.93×10^{-7} mbar, and inert Argon (Ar) gas flow rate of 25 cm³/min during the sputtering process.



Figure 3.6: Al Gate electrode deposited in top of REOs/Ge.

3.4 Characterization Techniques

3.4.1 Xray Diffraction (XRD) Analysis

XRD is widely used versatile characterization technique which could effectively deliver the chemical analysis report including elemental detection of the deported thin film demonstrating the phase identification, and crystallinity information. (e.g. amorphous or polycrystalline). This technique also well recognized for phase identification. In addition, XRD could also provide information regarding chemical composition and crystal structure of the deposited film. This information is gathered by the monochromatic Xray diffraction interference Xray within the crystal atoms at specified diffraction angle according to Eq. 3.1.

$$\lambda = 2dsin\theta \tag{Equation. 3.1}$$

Where,

 λ = Wavelength of Xray radiation beam (nm),

d = Interplanar spacing in a crystal atom (nm)

 2θ = Diffraction angle between the diffracted and transmitted beam (°)

In order to verify crystal phase orientation and crystallinity of the fabricated films X-ray diffraction (XRD) pattern were collected from Rigaku Miniflex Benchtop diffractometer. As X-ray radiation source copper radiation (Cu K α) has been applied corresponding operating power 40 kV with a wavelength $\lambda \sim 0.154$ nm and the measurement were taken at scan range $2\theta = 10^{\circ}$ to 80° at scanning rate 0.02°. The XRD pattern for designated sample can be obtained by plotting the intensity of Xray with respect to the diffraction angle 2θ .

It is important to evaluate the crystallite size and micro-strain in order to get insight on peak broadening due to crystal defect or dislocation. Peak broadening caused by crystallite size occurs due to incoherent dispersion between finite size elements. On the other side the peak broadening due to strain is initiated by the dislocation or non-uniform displacements of atom with the reference to lattice position. The crystallite size can be estimated using both Debye–Scherrer equation and Williamson – Hall (W - H) plot. The crystallite size using Debye–Scherrer approach is calculated using Eq. 3.2 (Goh et al., 2016b; Hetherin, Ramesh, & Wong, 2017b; J. W. Zhang et al., 2014).

$$\boldsymbol{D} = \frac{k\lambda}{\beta_D \cos\theta}$$
(Equation 3.2)

Where, *D* is the crystalline size, *K* is the shape factor which is a constant (0.9), λ is the wavelength of Cu K α (0.154 nm), β_D is the peak width with half maximum intensity

and θ is the peak position. Since the Debye–Scherrer equation only provides lower bound of crystallite size while micro-strain is not considered in this calculation. Thus, Williamson - Hall (W-H) plot was applied for calculating both crystallite size (*D*) and microstrain (ε) at the same time for various diffraction angles (2 θ) from the obtained XRD patterns of the thin film samples. Alongside with crystallite size (*D*), micro-strain (ε) of finite size particles is another important independent factor that can affect the peak broadening of the diffraction pattern. Hence, the total peak broadening of a diffraction pattern is an accumulated impact induced from combination of crystallite size and micro-strain which can be expressed as Eq. 3.3.

$$\boldsymbol{\beta}_{\tau} = \boldsymbol{\beta}_{\boldsymbol{D}} + \boldsymbol{\beta}_{\boldsymbol{\varepsilon}}$$
 (Equation. 3.3)

Where, β_{τ} is the total peak broadening, β_D and β_{ε} are the peak width with half maximum intensity due to crystallite size and induced strain, respectively. The induced strain (ε) due to crystal defect or dislocation of atom is given by Eq. 3.4

$$\boldsymbol{\varepsilon} = \frac{\beta_{\varepsilon}}{4\tan\theta}$$
(Equation. 3.4)

Replacing and rearranging the Eq. 3.2 and Eq. 3.4 into Eq. 3.3 gives the W-H as Eq. 3.4.

$$\beta_{\tau} \cos \theta = \frac{\kappa \lambda}{D} + 4\varepsilon \sin \theta$$
 (Equation. 3.5)

A graph of $\beta_{\tau} \cos \theta$ versus $4\varepsilon \sin \theta$ will be plotted according to Eq. 3.5. The intercept of the graph is equal to $\frac{\kappa\lambda}{D}$ which provides the estimation of crystallite size *D*, whereas the gradient (slope) of linear regression gives the micro-strain, ε value.

3.4.2 X-ray Photoelectron Spectrometer (XPS) Analysis

The interface chemical bonding states and chemical compositions of the oxidized/nitrided sputtered Sm/Ge thin films has been quantified ULVAC-PHI Quantera II, X-ray photoelectron spectrometer (XPS) with an X-ray source of monochromatic Al- k_{α} (hv = 1486.69 eV) was used. The chemical composition was measured from combination of both survey scan and narrow scan. Both survey scan and narrow scan was recorded at X-ray operating power of 50 W, X-ray beam size with an area of 300 μ m × 300 μ m using take off angle 45° with reference to normal surface. For the elemental analysis, survey scan has been recorded using high energy passing energy of 112 eV and energy resolution of 1eV/step. Afterwards, for chemical states quantification narrow scan has been recorded using a low passing energy of 112 eV with energy resolution 0.1 eV/step. In order to establish the band alignment mapping of Sm₂O₃/Ge interface an additional valance band scan was operated using this XPS system applying passing energy ~112 eV and higher energy resolution step size 0.1eV/step at binding energy ranging from -5 to 20 eV.

On the other side X-ray Photoelectron spectroscopy (model: Thermo Scientific K-Alpha) were used to study chemical compositions and chemical bonding states of thermally oxidized/nitrided sputtered Ho/Ge thin films. Both survey scan and narrow scan measurements were conducted at output power 150 W, with area of $400 \times 400 \ \mu\text{m}^2$ (X-ray beam size), take of angle 0° with respect to the normal surface. The survey scan was performed at passing energy of 160 eV and energy resolution of 1eV/step. Subsequently narrow scan measurement was taken at a low passing energy of 20 eV with high resolution step size 0.1 eV/step. Band alignment mapping of Ho₂O₃/Ge structure was also investigated from the XPS measurements. Considering the band alignment analysis, a valance band scan was performed using this XPS system where the binding energy ranging

from -5 to 20 eV with the same passing energy and energy resolution as used in this survey scan.

Deconvolution of the XPS spectrums was performed using CasaXPS software (version 2.3.19). Prior to deconvolution the linear background was corrected on this software. Afterwards, each of the narrow scan spectrums for both thermally oxidized sputtered Sm/Ge and Ho/Ge were calibrated with respect to C1s binding energy (B.E) at 284.80 eV (Lei et al., 2017) . The deconvolution of multiple peaks was processed by applying non-linear Gaussian-Lorentzian function during extraction of integrated area and fitting the line position (Wong & Cheong, 2011c).

3.4.3 RAMAN Analysis

Raman spectroscopy is well known as vibrational spectroscopic system which uses a monochromatic radiation from laser source to identify the existence of various organic and inorganic components residing into the crystals of material (Nickel, Friedrich, Rommelure, & Galtier, 2005). During RAMAN spectroscopy monochromatic radiation of phonons is irradiated on the material surface. As a result, the material surface molecules will be elevated to a higher level of excitement state due to high extent of molecular vibration. Subsequently new phonons will be generated from the surface once the molecular excitement is lowered to the preliminary position. According to the character of newly emitted phonons the species/compounds incorporating within the material has been identified by RAMAN spectroscopy (Larkin, 2017). In order to get further details on chemical composition and bonding stability Raman spectrums of thermally oxidized/nitrided sputtered Sm/Ge were collected from Horiba Xplora One Raman spectrometer using a scan ranging from (0 - 3000) cm⁻¹.

3.4.4 HRTEM Analysis

HRTEM is a kind of imaging microscope where transmission of electron beam at a higher energy level is being accelerated into the thin film sample plane. Due to the interference between sample film plane with the scattered/transmitted wave of electrons will generate an image of the sample plane. Afterwards, the sample plane image is being focused by an imaging device which produces high resolution magnified images. This image processing technique could capture images at atomic scale since the resolution is below 0.14 nm. HRTEM can come up with both real and reciprocal space information at the same time. Since, imaging mode for real space and diffraction mode for reciprocal space can run simultaneously (Kumar, Pavithra, & Naushad, 2019).

In this study, HRTEM was used to examine the cross sectional structural morphology and thickness of the thin film grown after thermal oxidation and nitridation. Before proceeding to HRTEM characterization samples underwent for focus ion beam lamella preparation employing Helios NanoLab 650 for attaining a cross-sectioned thin film. An additional sputtered Pt protective layer was deposited on the top of the sample surface in order to avert the ion bombardment defect induced during the focused ion beam lamella preparation. Afterwards, the cross-sectional morphology of the film was captured form the TECNAI G2 F20 high resolution transmission electron microscope. The HRTEM analysis was operated at enhanced voltage (20 - 200) kV with magnification scale 13.5 x to 134 kx. The pressure of HRTEM chamber was applied at 2.7×10^{-5} Pa in a vacuum ambient. In this study, HRTEM captured images of 1024×1024 pixels. Lastly, ImageJ software was used to extract the thickness from the HRTEM images. Furthermore, to verify the crystallinity, diffraction patterns of deposited thin films was generated employing Fast Fourier Transformation (FFT) process on HRTEM images regulated by ImageJ Software.

3.4.5 I-V Measurement

The current – voltage (*I-V*) characterization was performed from BPW-800 probe station linked with Keithley 4200 SCS semiconductor characterization system. The (*I-V*) measurement was reordered at frequency $f \sim 50$ Hz and sweep range (0 – 30) V with step size 0.05V. The current–voltage (*I-V*) measurement was transformed to leakage current density with respect to electrical field (*J-E*) plot by applying Equation Eq.3.6 (Hetherin et al., 2017a; Lei et al., 2018; Wong & Cheong, 2011b).

$$\mathbf{E} = \frac{V_g - V_{FB}}{t_{ox}}$$
(Equation. 3.6)

Where, V_g denotes gate voltage, V_{FB} is flatband voltage and t_{ox} signifies total oxide thickness including interfacial layer.

3.4.6 C-V Measurement

The capacitance – voltage (*C-V*) characteristics were also gained from mutual arrangement system of BPW-800 8" probe station and Keithley 4200 semiconductor characterization system (SCS). For *C-V* measurement gate voltage was swept bidirectionally using linear DC voltage ramp. For forward bias condition the gate voltage was swept between ± 10 V to ± 10 V to ± 10 V to ± 10 V to ± 10 V. To observe the frequency dispersion behaviour the measurement was taken for various frequencies of 100 kHz, 500kHz and 1MHz using forward bias sweep ± 10 V to ± 10 V. Several important electrical features of MOS capacitors have been retrieved from *C-V* curve which includes dielectric constant *k*, effective fixed oxide charge, Q_{eff} , hysteresis amount, slow trap density (*STD*), Q_{it} and average interface trap density, D_{it} .

CHAPTER 4: RESULTS AND DISCUSSION

4.1 Introduction

According to the objective of this research, the experimental findings obtained from various characterization techniques has been analysed and described in this chapter. The chapter consists of four main sections: (i) effect of various thermal oxidation/nitridation temperatures on RF magnetron sputtered Sm/Ge in N₂O ambient (ii) influence of various thermal oxidation/nitridation durations on RF magnetron sputtered Sm/Ge in N₂O ambient. (iii) influence of various thermal oxidation/nitridation durations on RF magnetron sputtered Ho/Ge in N₂O ambient. (iv) Comparison between thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge stack.

4.2 Effect of Various Thermal Oxidation/Nitridation Temperatures on Sputtered Sm/Ge in N₂O Ambient

4.2.1 XRD Analysis

A detailed investigation of XRD measurements were performed to examine the crystalline structure and crystallite growth of the developed thin film. Fig. 4.1 shows the diffraction spectrum of sputtered Sm on Ge substrate after oxidation/nitridation at different temperatures 300 °C, 400 °C, 500 °Cand 600 °C. Six diffraction peak of Trigonal-H, T-Sm₂O₃ are detected at diffraction angle of 27.16°, 31.15°, 47.28°, 48.26°, 54.04° and 56.36° corresponding to plane orientation (100), (101), (003), (110), (103) and (200) respectively for all the oxidized sample. Addition to these peaks one weaker peak of T-Sm₂O₃ found at 59.82° for 300 °C and a strong peak at 59.08° for 500 °C corresponding to plane (201). Two more T-Sm₂O₃ weak peak for 600 °C were detected at 40.98° and 60.48° corresponding to plane (102) and (004) respectively. These peaks were confirmed by Inorganic Crystal Structure Database (ICSD) reference code no. 647455 and 77685. The diffraction peak belongs to GeO₂ was detected in two different phases namely Tetragonal and Trigonal-H which are denoted as t1-GeO2 and t2-GeO2. Three diffraction peaks of t_1 -GeO₂ were detected at 29.16°, 42.92°, 57.18° corresponding to (110), (111) and (211) plane, respectively for all the oxidized sample. The peak found at 29.16° increased sharply when the temperature raised to 600 °C while the peak broadened with lower intensities for 300 °C – 500 °C samples. The t₂-GeO₂ was detected at 39.22° corresponding to (102) plane for all samples except 600°C. Along with these peaks four more additional t2-GeO2 peaks were found for 300 °C in diffraction angles of 20.52°, 25.96°, 38.10° and 41.78° corresponding to (1 0 0), (101), (012) and (200) plane, respectively. t₁-GeO₂ and t₂-GeO₂ peaks were confirmed by ICSD reference code no. 158593 and 53870. A sharp peak belongs to Ge₃N₄ was found for 500 °C at diffraction angle of 31.72° corresponding plane (211) while at 35.76° a weak peak appeared with corresponding plane (311) for all the oxidation temperatures. These diffraction peaks matched well with the ICSD reference code no. 97568 and 156338 which represents the cubic phase of c-Ge₃N₄. Moreover, for 600 °C orthorhombic o-Sm₄Ge₇ peaks were found at 22.84° and 39.14° with a plane of (101) and (203) respectively, while no such peaks were detected for other temperatures. These peaks were confirmed by the ICSD reference code 237695. However, there was no detectable diffraction peak from the substrate-Ge in the XRD pattern. Similar phenomenon was also observed in the XRD pattern of for ZrO₂/Ge system were prepared via thermal oxidation and nitridation in N₂O ambient (Lei et al., 2018). The XRD pattern confirms the formation of Sm₂O₃ for oxidation/nitridation temperature at (300-600) °C and presence of GeO₂, Ge₃N₄ and Sm₄Ge₇ indicates the formation of interfacial layer (IL) between oxide layer and Ge substrate. The findings also suggest that, the number of Sm₂O₃ peaks increased with the increment of oxidation/nitridation temperature which can be attributed to excessive diffusion of oxygen atoms from the atmosphere towards the high k/IL while increasing temperature. Conversely, the GeO₂ peak number decreases at 600 °C which might be due to in-diffusion of Sm or out-diffusion of Ge towards the high k/IL favouring the formation of Sm₄Ge₇ in place of GeO₂.

Furthermore, the highest intensity peak for Sm_2O_3 belongs to plane orientation of (110) and (201) for 300 °C and 500°C, respectively while (003) plane was highest for 400 °C and 600 °C samples. Meanwhile, the highest intensity peaks of GeO₂ for 300 °C were at t₂GeO₂ (101) plane and t₁GeO₂ (110) plane for rest of the samples. The highest diffraction peak intensity of Ge₃N₄ for 500 °C belongs to (211) plane while (311) plane was highest for 300 °C, 400 °C and 600 °C samples.

To investigate the effect of different oxidation and nitridation temperatures on the crystallinity of oxide and interfacial layer the highest intensities of Sm₂O₃, GeO₂ and Ge₃N₄ are plotted as shown in Fig. 4.2. Since the increase in intensities of diffraction peaks indicates the degree of crystallization. Hence, Fig. 4.2 reveals that GeO₂ has the highest intensities at 600 °C followed by 300 °C the second highest while intensities are almost same for 400 °C and 500 °C. This implies that, crystallinity of GeO₂ increases at higher oxidation temperature above 500°C. However, the peak intensity of GeO₂ is slightly higher for low temperature 300 °C in counter to 400 °C and 500 °C samples which indicates the improvement of GeO₂ crystallinity for 300 °C sample during the oxidation/nitridation process. The improved GeO₂ crystallinity at this oxidation/nitridation condition can be attributed to the self-diffusion of O atom from High-k interface. Consequently, increases the availability of O atom near the Ge substrate and inside the GeO₂ network which strengthens Ge-O inter-atomic bonding causing slightly intensified GeO₂ peak. The release of oxygen atom from Sm₂O₃ for 300 °C was evident from XPS Sm 3d spectra will be discussed in section 4.2.3. Meanwhile, the highest intensities of Sm₂O₃ and Ge₃N₄ are recorded at 500 °C. This observation indicates that, Sm₂O₃ and Ge₃N₄ are in more crystalline form for 500 °C while Sm₂O₃ and Ge₃N₄ appears to be in less crystalline form for rest of the temperatures. Comparing the intensities of Sm₂O₃, GeO₂ and Ge₃N₄ among all the temperatures it can be concluded that, both oxide and interfacial layer are relatively less crystalline for 400 °C sample.



Figure 4.1: XRD patterns of the Sm based gate stack at various oxidation/nitridation temperatures of 300 °C, 400 °C, 500 °C and 600 °C.



Figure 4.2: Intensities of Sm_2O_3 , GeO_2 and Ge_3N_4 at various oxidation/nitridation temperatures (300 – 600) °C.

In order to get insight on peak broadening due to crystal defect or dislocation. crystallite size, (*D*) and micro-strain, (ε) has been estimated using Williamson - Hall (W-H) as described in section 3.4.1. A graph of $\beta_{\tau} \cos \theta$ versus $4\varepsilon \sin \theta$ for Sm₂O₃ and GeO₂ was plotted according to Eq. 3.5 as in Fig. 4.3. A linear fitting was performed with selected point from the distributed values at regression co-efficient r² = 1. According to Eq. 3.5 the crystallite size, *D* has been approximated from the intercept of the graph and micro-strain, ε was approximated from the gradient (slope) of the line. The crystallite sizes of Ge₃N₄ and Sm₄Ge₇ were not calculated because the number of peaks were insufficient for the calculation.

The crystallite size (*D*) and micro-strain (ε) calculated from W-H plot are shown in Fig. 4.4. The crystallite size of Sm₂O₃ for sample 300 °C, 400 °C, 500 °C and 600 °C are 18.39 nm, 32.48 nm, 32.26 nm, and 59.19 nm, respectively, while the crystallite size of GeO₂ increases gradually from 39.40 nm to 68.72 nm with increment of oxidation temperature. It is observed that, Sm₂O₃ crystallite sizes at 400 °C and 500 °C exhibits very close value with smaller differences which might be due to more homogenous distribution of crystallites at these temperatures (Goh et al., 2016b). Meanwhile, all the samples show negative micro-strain values for Sm₂O₃ crystallite except sample oxidized at 300 °C. On the other hand, the micro-strain of GeO₂ crystallite exhibits positive values for all samples except 400 °C oxidized sample. Additionally, the highest micro-strain value for both Sm₂O₃ and GeO₂ crystallites were recorded at 300 °C while the lowest recorded at 400 °C. However, the micro-strain value of Sm₂O₃ crystallite was same for both 500 °Cand 600 °C while the second highest value of GeO₂ crystallite was recorded at 600°C.



Figure 4.3: W–H linear regression plot of a. Sm_2O_3 , b. GeO_2 at regression co-efficient $r^2 = 1$ for various oxidation/nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C).



Figure 4.4: Micro-strain and crystallites of Sm₂O₃ and GeO₂ at various oxidation and nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C).

4.2.2 J-E Characteristics

Fig. 4.5 displays the typical leakage current density Vs electrical breakdown field (*J-E*) for different oxidation/nitridation temperature 300 °C, 400 °C, 500 °C and 600°C. The (*J-E*) plot was converted from the current–voltage (*I-V*) measurement by using following Eq. 3.6. The sample that yields the highest breakdown electrical field, E_{BD} and the lowest leakage current density, J_g is considered as the best electrical properties. Two step breakdowns (E_{SBD} and E_{HBD}) are being observed for all the samples except 600 °C oxidized/nitride sample. The presence of interfacial layer and Sm₂O₃ layer is the reason behind two step breakdowns (Goh et al., 2016a; Wong & Cheong, 2011a). The breakdown at lower electric field along with relatively small instantaneous increment leakage current density is considered as soft breakdown, E_{SBD} . Conversely, the breakdown at higher electric field with large instantaneous increment leakage current density labelled as hard breakdown, E_{HBD} . The recorded electrical breakdown field of E_{HBD} for 300 °C, 400 °C, and 500 °C samples were 9.14 MV cm⁻¹, 13.31 MV cm⁻¹ and 11.25 MV cm⁻¹, respectively.

The leakage current density of E_{HBD} for these samples were in 10⁻⁴, 10⁻⁶, and 10⁻⁵ order of magnitude, respectively. This analysis shows that 400 °C sample yielded the highest electrical breakdown field, E_{BD} of 13.31 MVcm⁻¹ with lowest leakage current density, J_g of 8.38 × 10⁻⁶ A cm⁻². Therefore, 400 °C oxidized/nitrided sample has the best electrical properties for Sm₂O₃ dielectric MOS devices. For 600 °C sample *J*-*E* plot reveals almost a short circuit behaviour. The reason for short circuit nature is due to the formation of Sm–Ge metallic bond. The presence of Sm–Ge for 600 °C sample was confirmed from the XRD pattern. Moreover, 400 °C sample has shown comparatively higher electrical breakdown field and lower leakage current density than the reported literature of ZrO₂/Ge and Sm₂O₃/Si based MOS structure (Goh et al., 2016a; Lei et al., 2018).



Figure 4.5: *J-E* characteristics of Sm₂O₃/Ge gate stack at different oxidation and nitridation temperatures (300 °C, 400 °C, 500 °C, and 600 °C).

4.2.3 XPS Analysis

Based on XPS wide scan Sm 3d, Ge 3d, Ge 2p, Ge 3s and O1s (data not shown) core level spectra were detected. The core level spectra of N1s were untraceable during the wide scan might be due to very low atomic percentage of nitrogen content below the XPS detection limit range (0.1 - 1) at %. Moreover, the deep penetration of N1s content nearby the Ge substrate during thermal oxidation/nitridation in N₂O ambient was evident from reported literature (Lei et al., 2018). Since Ge₃N₄ was observed in the XRD analysis and so it is suspected that N content is deeply seeded near the Ge substrate which was not detected in XPS measurement due to limited surface analysis depth (~5 nm). Narrow scan of core level spectrum Sm 3d and Ge 3d were performed to examine the structural and compositional changes of Sm₂O₃/Ge dielectric for various oxidation/nitridation temperatures (300 - 500) °C as shown in Fig. 4.6 and Fig. 4.8.

Fig. 4.6 shows the deconvoluted peaks of Sm 3d narrow scan for different oxidation/nitridation temperatures of 300 °C, 400 °C and 500 °C. For 400 °C and 500 °C samples the centroid Sm3d peak shifts positively towards higher binding energy by (0.20 – 0.37) eV with respect to reference position of Sm 3d at 1084.0 eV (C. C. Lin et al., 2014). The line position of Sm 3d spectra shifting towards higher binding energy clearly suggests the presence of Sm–O–Ge interfacial layer (IL) between the Sm₂O₃ film and Ge substrate. The growth of Sm–O–Ge is mainly related with the out-diffusion of Ge and Sm–O–Sm suboxide elimination. Since the Ge–Ge covalent bond breaks even at room temperature and so, out diffusion of Ge atom is more prominent at higher oxidation temperature (Mitrovic et al., 2014). It is well known that Ge is more electronegative than Sm (electronegativity Ge = 2.01 and Sm = 1.7 using Pauling's scale) indicating relatively higher electron accepting nature of Ge. Thus, the diffused Ge atoms replaces the Sm atoms from Sm–O–Sm bonding configuration promoting the formation of Sm–O–Ge (C. C. Lin et al., 2014). Based on this assumption, it is suggested that positive shifting in 400 °C and 500 °C samples was due to partial elimination of Sm–O–Sm suboxide caused by out-diffusion of Ge atoms. Meanwhile it is noted that, 300 °C sample shifts slightly toward lower binding energy from the reference line position which indicates the deficiency of Sm–O–Ge as interfacial layer component.



Figure 4.6: Sm 3d narrow scan deconvoluted spectra for various oxidation and nitridation temperatures (300 °C, 400 °C, and 500 °C).

According to the literature peak position of Sm_2O_3 was reported in lower binding energy at 1083.90 eV and Sm-O-Ge in higher binding energy at 1086.2 eV (C. C. Lin et al., 2014). The sample oxidized/nitrided at 300 °C comprises only a single peak in the lower binding energy at 1083.88 eV which confirms the formation of Sm_2O_3 . For, sample oxidized/nitrided at 400 °C and 500 °C was deconvoluted into two peaks where Sm_2O_3 located at lower binding energy site and Sm-O-Ge on higher binding energy site. For 400 °C and 500 °C sample the Sm 3d peak belongs to lower binding energy site are at 1083.93 eV and 1083.97 eV for 400 °C and 500 °C, respectively which corresponds to Sm_2O_3 . Sm 3d peak belongs to higher binding energy lies at 1085.17 eV and 1085.31 eV, respectively for these temperatures which corresponds to Sm-O-Ge.

The effect of different oxidation temperature on the growth of dielectric oxide is analysed by comparing the integrated peak area is of Sm_2O_3 and Sm-O-Ge. The Sm_2O_3 integrated peak area increased at 400 °C and reduced subsequently for 500 °C and 300 °C. On the Contrary, the integrated peak area of Sm-O-Ge increased gradually while increasing oxidation temperature from 400 °C to 500 °C. For 400 °C the increase of Sm_2O_3 and reduction of Sm-O-Ge is due to fewer diffusion of Ge atom into the Sm_2O_3 dielectric. Conversely, for 500 °C reduction of Sm_2O_3 and increase in Sm-O-Ge is mainly due to excessive amount Ge diffusion into Sm_2O_3 causing large amount Sm substitution during Sm-O-Sm suboxide elimination. For 300 °C the reduction of Sm_2O_3 and deficiency of Sm-O-Ge is due to O diffusion from the dielectric Sm_2O_3 itself toward the interface preventing the diffusion of Ge into Sm_2O_3 dielectric.

Fig. 4.7 depicts the oxidation level of 300 °C, 400 °C and 500 °C oxidized samples by plotting the binding energy shift of Sm 3d versus temperature. For 400 °C sample Sm 3d binding energy is closer to full oxidation level in counter to 500 °C indicating certain level of over-oxidation condition for 500 °C sample. As of for 300 °C sample Sm 3d binding energy is below the full oxidation level indicating under-oxidation condition. From this observation it is evident that oxidation rate gradually increases with the increment of oxidation/nitridation temperature. In contrast with the *J-E* characteristics discussed in section 4.2.2 both the under-oxidation and over-oxidation condition influenced adversely on the electrical breakdown field voltage. The 400 °C samples exhibit moderate oxidation rate and the highest electrical breakdown field voltage.



Figure 4.7: XPS Sm 3d narrow scan peak position of various investigated temperatures (300 °C, 400 °C and 500 °C).

Fig. 4.8 demonstrates the Ge 3d spectra at different oxidation and nitridation temperature of 300 °C, 400 °C and 500 °C. Ge 3d was deconvoluted into four peaks with the reference position of Ge $3d^0$ at 29.6 eV (Fu et al., 2014). The four peaks corresponding to Ge¹⁺, Ge²⁺, Ge^{*}, Ge⁴⁺ were well fitted with a core level shift of +0.8 eV, +1.8eV, +2.4 eV and > +3 eV, respectively. It has been reported that, the shift lies at +0.8 eV can be denoted as GeO_x while shift lies at +1.8 eV was denoted as GeO (Mitrovic et al., 2014; J. Song et al., 2007). The Ge⁴⁺ was assigned for native oxide GeO₂ which is expected to appear in between +3 to +3.4 eV (Dimoulas et al., 2010; Mitrovic et al., 2014; Sasada, Nakakita, Takenaka, & Takagi, 2009). The shift lies at +2.4 can be ascribed as Ge* corresponds to Sm–O–Ge, since the metal germanate ($M_xGe_yO_z$) of rare earth material tends to occur in between the Ge²⁺ and Ge³⁺ in range of 1.8 eV to 2.8 eV due to second nearest neighbour effect (Mitrovic et al., 2014). From Fig. 4.8 it is evident that, the content of native oxide GeO₂ formed for all oxidation temperatures. A strong existence of Sm_xGe_yO_z and GeO_x is found for the temperature 400 °C and 500 °C except 300 °C. Conversely, volatile GeO is found at 300 °C and 500 °C except 400 °C.

The effect of various oxidation temperature on interface stability is examined by comparing the integrated area of deconvoluted peaks from Fig. 4.8. The stoichiometric stable phases of GeO₂ with higher binding energy is found more for the sample oxidized at 400 °C and reduces subsequently at 500 °C and 300 °C samples. Consequently, the lower binding energy sub-stoichiometric component Sm_xGe_yO_z, GeO_x and GeO with unstable phase are fewer for 400 °C than those of 300 °C and 500 °C. In addition, 500 °C sample exhibits larger integrated area of stoichiometric GeO₂ and less GeO than 300 °C sample. Since GeO is the most thermally unstable interface component compare to other sub-oxide of Ge (D. Wang, He, Fang, et al., 2019). Hence, most pronounce growth of volatile GeO at 300 °C sample indicates highly defected interface quality which will deteriorate the electrical properties. The reduction of volatile GeO for both 400 °C and 500 °C sample can be attributed to the spontaneous formation of oxygen deficient SmGeO_x. The oxygen deficient SmGeO_x traps the desorbing GeO and concurrently preserves the integrity of GeO₂ (H. Wang, Chroneos, Dimoulas, & Schwingenschlögl, 2012). In addition, the fewer unstable phases at 400 °C sample might be related to the formation of

amorphous Ge₃N₄ due to N⁺ atom diffusion which effectively passivates excessive Ge diffusion during this oxidation-nitridation process (Maeda et al., 2007; Venkata Rao et al., 2018). The detail mechanism of GeO trapping, and Ge passivation will be discussed in section 4.2.6. Based on the above discussion it can be ascribed that, 400 °C oxidized sample exhibits more stable IL layer with high composition of stoichiometric components and fewer traps which will improve the electrical properties of the dielectric.





4.2.4 Band Alignment

The band diagram of Sm₂O₃/IL/Ge was obtained for different oxidation temperature of 300 °C, 400 °C and 500 °C by deducing O1s plasmon loss XPS spectra and valance band spectra as shown in Fig. 4.9. Based on Krauts method the same extraction procedure described in the reported literature by Wong & Cheong, (2011a, 2012a) were employed to determine the valance band edge, (E_v) and bandgaps, (E_g) of Sm₂O₃/IL. Fig. 4.9a illustrates the XPS valance band spectra for sample investigated at 300 °C, 400 °C and 500 °C. A typical small tail is observed in the valence band spectra which is considered as Ge signal. The valence band edge, (E_v) of Ge = 0.10 eV was used as reference from literature (Cao, Wu, Wu, & Li, 2017; S. J. Wang et al., 2004). The valance band edge, (E_v) for sm₂O₃/IL was calculated by extrapolation of the maximum negative slope near the edge to the minimum horizontal Ge signal baseline as illustrated in Fig. 4.9a (Cao et al., 2017; Wong & Cheong, 2011a). The valance band edge, (E_v) of Sm₂O₃/IL with respect to Ge substrate were estimated 3.27, 3.02 and 3.08 eV for 300 °C, 400 °C and 500 °C, respectively. The valence band offset, VBO (ΔE_v) was estimated from the difference between valance band edges, (E_v) of Sm₂O₃/IL and Ge using Eq. 4.1:

$$\Delta E_{\nu} \left(\text{Sm}_2 \text{O}_3 / \text{IL} \right) = \left(E_{\nu} \right) \left(\text{Sm}_2 \text{O}_3 / \text{IL} \right) - \left(E_{\nu} \right) \left(\text{Ge} \right)$$
(Equation 4.1)

The valence band offsets, ΔE_{ν} (Sm₂O₃/IL) were estimated 3.17, 2.98, 2.92 eV for 300 °C, 400 °C and 500°C, respectively. The conduction band offset, CBO (ΔE_c) of Sm₂O₃/IL/Ge can be estimated using the following Eq. 4.2, where E_g (Sm₂O₃/IL) and E_g (Ge) are the energy bandgaps of Sm₂O₃/IL and Ge respectively.

$$\Delta E_c (\mathrm{Sm}_2\mathrm{O}_3/\mathrm{IL}) = E_g (\mathrm{Sm}_2\mathrm{O}_3/\mathrm{IL}) - E_g (\mathrm{Ge}) - \Delta E_v (\mathrm{Sm}_2\mathrm{O}_3/\mathrm{IL}) \quad \text{(Equation 4.2)}$$

The energy bandgap of Sm₂O₃/IL for different oxidation/nitridation temperature were extracted from their respective O1s plasmon loss XPS spectra as shown in Fig. 4.9b. The energy bandgaps E_g (Sm₂O₃/IL) were estimated to be ranging between 5.56 – 6.25 eV for the investigated oxidation/nitridation temperature. The energy bandgap, E_g (Ge) value of 0.67 eV was obtained from the report of (Kamata, 2008). Fig. 4.10 summarizes the schematic energy band alignment of Sm₂O₃/IL/Ge corresponding to different oxidation/nitridation temperature. The CBO, ΔE_c (Sm₂O₃/IL) were estimated 1.72, 2.60 and

2.05 eV respectively, for 300 °C, 400 °C and 500 °C sample as shown in Fig. 4.10. According to the band offset schematic it can be suggested that both VBO and CBO for oxidation/nitridation temperature (300 - 500) °C of Sm₂O₃ on Ge substrate is above 1 eV which makes it eligible contender for high-k dielectric on Ge based MOS devices. In addition, 400 °C exhibits the highest value of ΔE_c (Sm₂O₃/IL) compared to other investigated samples. The increment of ΔE_c (Sm₂O₃/IL) indicates reduction of defect gap states in the conduction band edge of Sm₂O₃/IL with Ge substrate. It is anticipated that, formation of stochiometric Sm_2O_3 and stable interfacial Sm-O-Ge at this temperature reduces defect gap states. During the formation of Sm-O-Ge the four folded configuration of Sm atom would permit to channel with four unoccupied bonds of Ge atoms eventually removing the Ge dangling bond causing defect gap state free Sm₂O₃/IL edge with Ge. The increase in ΔE_c (Sm₂O₃/IL) reduces the leakage current of and breakdown voltage of the device which is in well agreement with the *J-E* characteristics discussed in section 4.2.2. Furthermore, Fig. 4.9b reveals that energy bandgap, E_g of (Sm₂O₃/IL) for 300 °C and 500 °C drops by 0.69 eV and 0.61 eV, respectively than the 400 °C sample bandgap $E_g \sim 6.25$ eV. Decreasing of band gap for oxide/Ge interface can be attributed to the generation of defect states due to GeO (g) desorption (Kita, Lee, Nishimura, Nagashio, & Toriumi, 2009). Accordingly, for 300 °C and 500 °C samples it is considered that, desorption of volatile GeO (g) through the Sm₂O₃ dielectric film induced oxygen vacancy and Ge²⁺ defects in the Sm₂O₃/IL interface. These defects triggered the generation of interface defect states in the bandgap region of Sm₂O₃/IL interface causing bandgap reduction for these samples. However, all the investigated samples induced higher bandgap E_g (Sm₂O₃/IL) in counter to the typical Sm₂O₃ energy bandgap, $E_g \sim 4.33$ eV. Since, the bandgap was calculated including interfacial layer accordingly the energy bandgap E_g (Sm₂O₃/IL) was higher which might be due to the contribution of sandwiched interfacial layer between the Sm₂O₃ and Ge substrate.



Figure 4.9: Estimation of valance band offset (ΔE_v) and energy bandgap (E_g) for Sm₂O₃/IL/Ge structure at various oxidation/nitridation temperatures 300 °C, 400 °C and 500 °C (a) Valance band edge, E_v and VBO extraction from XPS valence band spectra. (b) Energy band gap deduction from oxygen plasmon loss XPS spectra.



Figure 4.10: Schematic of band alignment for Sm₂O₃/IL/Ge gate stack for different oxidation/nitridation temperatures.

4.2.5 HRTEM Analysis

It is important to realize the relation between film thickness with the functioning oxidation and nitridation temperature of the fabricated Sm₂O₃ gate stack to verify the compliance with MOS scaling limit. The sample underwent for HRTEM analysis was selected based on their electrical properties where 400 °C sample is having the lowest leakage current density and 500 °C sample is having the second lowest leakage current. Hence, the cross-sectional view obtained from HRTEM for sample oxidized/nitride at 400 °C and 500 °C is shown in Fig. 4.11. Two distinguishable layers is clearly visible on the Ge substrate, which confirms the presence of interfacial layer along with oxide layer. The layer close to the Ge substrate is assigned as the IL layer while layer in the top with relatively larger distance from Ge substrate is assigned as Sm₂O₃ dielectric layer. According, XPS and XRD analysis the interfacial layer consists of Ge–O, Ge–N and Sm–O–Ge. The measured thickness of Sm₂O₃ dielectric for 400 °C and 500 °C sample was approximately 3.56 nm and 3.05 nm, respectively while the interfacial layer (IL) was around 2.44 nm and 3.26 nm, respectively. This result indicates that, the IL layer increases, and the oxide layer decreases with the increment of temperature. Based on the
result demonstrated in XPS analysis it can be assumed that degree of oxidation level increases with the increment of oxidation/nitridation temperature which caused severe outdiffusion of Ge into the Sm₂O₃ dielectric for 500 °C sample and thus increased the interfacial thickness. However, the obtained film thickness is not similar with the crystallite size trend of Sm₂O₃ (18.39 – 59.19 nm) and GeO₂ (39.40 – 68.72 nm) found from XRD. The possible reason could be attributed to the crystal alignment (Wong & Cheong, 2012c) as illustrated in Fig. 4.12 in which the approximated crystallite size of Sm₂O₃ (110) plane was not in the direction of $t_{ox(max)}$ or $t_{ox(min)}$ while crystallite size of GeO₂ (110) plane was not in the direction of $t_{IL(max)}$ or $t_{IL(min)}$. The crystal plane orientation of Sm₂O₃ (110) and GeO₂ (110) were used as reference obtained from the XRD pattern.

Furthermore, to verify the crystallinity, diffraction pattern of the HRTEM images was obtained from ImageJ software using fast Fourier transform (FFT) method shown as inset image. Considering the inset FFT image it can be inferred that, 500 °C forms partially crystalline Sm₂O₃ layer which is confirmed by the presence of multiple spotted patterns while interfacial layer is found to be amorphous with cloudy spot pattern. Conversely, both the Sm₂O₃ and interfacial layer were in amorphous form (cloudy spot pattern) for 400 °C. This result is in well agreement with the crystallinity observed in XRD pattern section 4.2.1.





Figure 4.11: Cross-sectional HRTEM images of sputtered Sm/Ge samples oxidized/Nitrided at 400 °C and 500 °C. Inset showing diffraction pattern obtained using FFT method.



Figure 4.12: Schematic illustration of crystal structure arrangement for Sm₂O₃ (110) and GeO₂ (110) with respect to the thickness from the Ge substrate baseline.

4.2.6 Oxidation and Nitridation Mechanism Model

Based on the observation reported in XRD, XPS and HRTEM a possible oxidation and nitridation mechanism for $Sm_2O_3/IL/Ge$ gate stack has been shown in Fig. 4.13 for various oxidized/nitrided sample at 300 °C, 400 °C, 500 °C and 600 °C. Initially, for all the investigated sample N₂O diffused and reacted with sputtered Sm of the Ge substrate surface according to Eq. 4.3:

$$2Sm + 3N_2O(g) \rightarrow Sm_2O_3 + 3N_2(g)$$
 (Equation 4.3)

Meanwhile, the extent of oxidation/nitridation increased for all the temperatures thus, N₂O gained enough energy to diffuse through the Sm₂O₃ dielectric and reacted with Ge substrate into two step reaction. In the first step only GeO₂ was formed with Ge wafer according to the reaction of Eq. 4.4 while in the second step both GeO₂ and Ge₃N₄ was formed by Eq. 4.5. It was assumed that some extent of Ge dangling bonds was passivated during the reaction of Eq. 4.5 creating an amorphous Ge₃N₄ in the interfacial layer. Presence of Ge₃N₄ for all the temperatures were confirmed in XRD analysis.

$$Ge + 2N_2O(g) \rightarrow GeO_2 + 2N_2(g)$$
 (Equation 4.4)

$$4\text{Ge} + 2N_2O(g) \rightarrow \text{GeO}_2 + \text{Ge}_3N_4$$
 (Equation.4.5)

Concurrently, for 400 °C and 500 °C sample Ge atom out-diffused towards the Sm₂O₃ dielectric and reacted with Sm atom, since Ge atoms were active at these temperatures (Kamata, 2008; Lei et al., 2017). Subsequently, SmGeO_x was formed in the IL layer according to Eq. 4.6 which was confirmed in XPS observation.

$$Sm_2O_3 + Ge \rightarrow SmGeO_x$$
 (Equation 4.6)

Afterwards, for all the investigated sample once GeO₂ was formed volatile GeO (g) gas desorption occurred due to the redox reaction between GeO₂ and Ge wafer for all the investigated sample according to Eq. 4.7 (Kamata, 2008; Kita et al., 2008). For, 300 °C there was continuous desorption of GeO (g) into the dielectric through the interfacial layer which resulted in defective dielectric interface as shown in Fig. 4.13a

$$3\text{GeO}_2 + 3\text{Ge} \rightarrow 6\text{GeO}(g)$$
 (Equation 4.7)

Since, at 300 °C temperature the extent of oxidation/nitridation was not adequate to fully oxidize the Sm_2O_3 dielectric hence additional O_2 in-diffused toward IL layer which strengthened the detrimental redox reaction of Eq. 4.7. Subsequently, for 400 °C and 500 °C sample GeO (g) desorbed into the IL layer and reacted with interfacial $SmGeO_x$. As a product denser $Sm_xGe_yO_z$ was formed with an additional growth of Ge sub-oxide GeO_x, by trapping the volatile GeO through the reactive reaction of Eq. 4.8:

$$SmGeO_x + GeO \rightarrow Sm_xGe_yO_z + GeO_x$$
 (Equation 4.8)

Regarding the trapping process of GeO (g) into SmGeO_x, it is assumed that at the initial phase of the oxidation/nitridation an oxygen deficient SmGeO_x was produced. Afterwards desorption of GeO (g) favoured the oxygen deficient SmGeO_x to restore some of oxygen loss balancing the propensity of Sm and O (Mitrovic et al., 2014; Nishimura et al., 2011; H. Wang et al., 2012). This preceded to oxygen densification of oxygen deficient SmGeO_x and thereby formed an amorphous sub-stoichiometric $Sm_xGe_yO_z$ with residual GeO_x in the IL layer as shown in Fig. 4.13b. Therefore, restricted the GeO (g) volatilization toward the Sm_2O_3 dielectric film. Since Sm_4Ge_7 was found in XRD for 600 °C, it is assumed to be formed by the reaction of Eq. 4.9.

$$2Sm_2O_3 + 20GeO \rightarrow Sm_4Ge_7 + 13GeO_2$$
 (Equation 4.9)

It is thought that excessive desorption of GeO(g) at this temperature damaged the Sm_2O_3 dielectric leaving several metallic defects as shown in Fig. 4.13c which resulted in a short circuit as illustrated in section 4.2.2 *J-E* characteristics.



Figure 4.13: Schematic of a possible oxidation and nitridation mechanism and layer distribution model for different temperature oxidized/nitrided sputtered Sm/Ge sample a. 300 °C b. 400 °C and 500 °Cand c. 600 °C.

4.2.7 Capacitance-Voltage (C-V) Characteristics

Fig. 4.14 illustrates the *C*-*V* characteristics of the investigated samples of different oxidation/nitridation temperature 300 °C, 400 °C and 500 °C. The *C*-*V* measurement was taken at frequency of 100 kHz and gate bias was bi-directionally swept from -1 V to +1.5 V. The inset of Fig. 4.14 shows the relationship of effective dielectric constant (*k*) and different oxidation/nitridation temperature. The effective dielectric constant (*k*) were extracted by using the Eq. 4.10 (Goley & Hudait, 2014).

$$C_{ox} = \frac{k\varepsilon_o A}{t_{ox}}$$
(Equation 4.10)

Where, C_{ox} is the oxide capacitance, (k) is the dielectric constant, ε_o is the permittivity of free space, $8.854 \times 10^{-12} Fm^{-1}$, A is the capacitor area and t_{ox} is the total physical oxide thickness including Sm₂O₃ layer and the interfacial layer of the deposited film. Since, A and ε_o are constant for this work, it can be anticipated that the oxide capacitance is proportional to $\frac{k}{t_{ox}}$. Hence, the oxide capacitance is influenced by the physical oxide thickness and dielectric constant.

From Fig. 4.14 it is clearly visible that, 400 °C sample yielded highest oxide capacitance among the investigated samples. It was evident from the HRTEM analysis that 400 °C induces lower thickness than 500 °C hence, increasing the oxide capacitance. Moreover, the highest dielectric constant, k_{eff} value was obtained for 400 °C which leads to the higher oxide capacitance. The increase of k_{eff} can be attributed to the higher concentration of Sm₂O₃ and better thermal stability of Sm₂O₃/IL/Ge structure at this oxidation/nitridation condition (Pan & Huang, 2010). Although the lowest Sm₂O₃/IL thickness were estimated at 300 °C yet exhibits the lowest oxide capacitance which is due to lowering of effective dielectric constant, k_{eff} . The reduction of dielectric constant can be

attributed to the degraded interface quality of oxide/IL/Ge structure. In addition, the equivalent oxide thickness (*EOT*) estimated for the Sm₂O₃/IL/Ge MOS interface were 0.75 nm, 0.61 nm, and 0.68 nm for 300 °C, 400 °C and 500 °C, respectively obtained from their respective dielectric constant, k_{eff} value.



Figure 4.14: 100 kHz, C-V curve measured in forward and reverse bias condition for oxidized/nitrided Sm₂O₃/IL/gate stack at various temperatures (300 °C – 500°C). The inset is the effective dielectric constant of the investigated samples as a function of oxidation/nitridation temperatures.

In *C-V* plot, depletion region was found in positive bias side for all the investigated capacitors. This finding indicates that flatband voltage (V_{FB}) shifted positively, which confirms the presence of negative oxide charges or native defects in the gate stack (C. C. Cheng et al., 2007; J. Gao et al., 2017; Wong & Cheong, 2012c). Several reported literatures have demonstrated that oxygen vacancies generated by GeO desorption causes flatband voltage (V_{FB}) shift by generating additional negative charges (Cao et al., 2017; C. C. Cheng et al., 2007). It is clearly visible from Fig. 4.14 that, 400 °C sample shows an obvious negative shift in compared to other investigated samples. This indicates generation of fewer negative oxide charges for 400 °C sample which might be due to alleviation of oxide defects or existence of singly or doubly negatively charged oxygen interstitial (J. Gao et al., 2017).

Hysteresis phenomenon has been n oticed in the *C-V* curve between the forward bias and reverse condition for all the investigated samples. The occurrence of hysteresis phenomena has been attributed to the presence of slow trapped charges located in the bulk oxide of gate dielectric or in the adjacent region between interfacial layer and high-*k* interface (Ke et al., 2018; Wong & Cheong, 2012c). Hence, it can be projected that induced hysteresis effect for $Sm_2O_3/IL/Ge$ MOS capacitor is associated with the amount of defects/traps generated in the Sm_2O_3 gate insulator and at the Sm_2O_3/IL interface. Since slow trap density (*STD*) are related with hysteresis hence, *STD* has been approximated according to equation Eq. 4.11 (Wong & Cheong, 2012c).

$$STD = \frac{\Delta VC_{ox}}{qA}$$
(Equation 4.11)

Where, ΔV is the flatband voltage difference between hysteresis curve, A is the capacitor area and q is the electronic charge. Fig. 4.15 shows the relationship of slow trap density and hysteresis voltage calculated from the *C-V* curve with respect to different oxidation/nitridation temperature. The recorded hysteresis voltage shown in Fig. 4.15 indicates that 400 °C sample exhibits the lowest hysteresis voltage ~ 50 mV while the hysteresis voltage subsequently increases for 500 °C and 300 °C samples. Fig. 4.15 also reveals that 400 °C sample yielded the lowest *STD* ~ 1.62 × 10¹² cm⁻² while the *STD* increases subsequently for 500 °C and 300 °C sample. Hence, it is obvious that the reduction of slow trap density contributed to lowering the hysteresis effect for 400 °C sample. On the other hand, the increment of slow trap density enhanced the hysteresis effect for 300 °C and 500°C. It is worth noting that hysteresis voltage increases proportionally to the slow trap density. The increment of slow trap density can be attributed to the tunnelling of large amount carrier charge (electron or hole) from the Ge substrate into the interfacial layer or in bulk oxide (Ke et al., 2018). The *STD* value ~ 1.62×10^{12} cm⁻² obtained for 400 °C sample in our work is comparatively lower than the *STD* value ~ 5.56×10^{12} cm⁻² of Sm₂O₃/Si structure (Goh et al., 2017b). Furthermore, noticeable hump was observed for both 400 °C and 500 °C while a small kink appears for 300 °C in the inversion region. This phenomenon might be induced by the generation of slow interface states located in Sm₂O₃/IL or in bulk Sm₂O₃ interface due to N⁺ atom out-diffusion during oxidation and nitridation process (Misra et al., 2006).



Figure 4.15: Slow trap density and hysteresis voltage as function for Sm₂O₃/IL/Ge gate stack at different oxidation/nitridation temperature (300 – 500) °C.

Fig. 4.16 shows the average interface trap density D_{it} as function of trap energy level (E_C – E) of for different oxidation-nitridation temperature from (300 – 500°C) °C. The average interface trap density D_{it} was approximated using Terman method ascribed in Eq. 4.12 (Schroder, 2015; Wong & Cheong, 2011b).

$$\boldsymbol{D}_{it} = \frac{\Delta V_g \boldsymbol{C}_{ox}}{\varphi_s q A} \tag{Equation 4.12}$$

Where, φ_s is the surface potential of substrate Ge at a specified gate voltage V_g and ΔV_g indicates the difference between ideal gate voltage and experimental gate voltage. The surface potential of a particular capacitance was obtained from the ideal MOS capacitor. The gate voltage was obtained from experimental C-V curve with same capacitance as that of φ_s . The steps were repeated until a relevant D_{it} versus ($E_c - E$) curve was obtained as shown in Fig. 4.16. The observation reveals that 400 °C sample yielded the lowest average interface trap density, D_{it} was approximated to be 4.76×10^{13} eV⁻¹ cm⁻¹ ². Concerning the root of interface traps, it is believed that inclusion of Ge dangling bonds (DB) and the oxygen-defect (OD) are generally liable for the charge trapping procedure at the dielectric/Ge interface (Baldovino, Molle, & Fanciulli, 2008; Bethge et al., 2014). According to Fig. 4.16 it is evident that 400 °C sample exhibits fewer DB and OD density than 300 °C and 500°C. Since OD is primarily generated from GeO desorption and thus reduction of OD indicates the suppression of GeO desorption at 400 °C complementary with the XPS observation (Chroneos et al., 2012; Mitrovic et al., 2014). Meanwhile, the reduction of DB might be due to formation amorphous Ge₃N₄ and stoichiometric GeO₂ as observed in XRD and XPS analysis. The exclusion of DB can also be clarified from the valency passivation approach owing to beneficial trivalent characteristic of Sm³⁺. In this approach Sm atom owing distinct valency than Ge would breach into oxygen deficient

GeO₂ and consequently displaces unbonded Ge atoms by forming interfacial Sm–O–Ge (C. C. Lin et al., 2014).

Comparing with the previous literature, the lowest $D_{it} \sim 4.76 \times 10^{13} \text{ eV}^{-1} \text{ cm}^{-2}$ achieved for 400 °C sample in this investigation is significantly higher than the post deposition annealed (PDA) Sm₂O₃/Ge interface $D_{it} \sim 5.1 \times 10^{11} \text{ eV}^{-1} \text{ cm}^{-2}$ and Sm₂O₃/Si interface $D_{it} \sim 3.8 \times 10^{12} \text{ eV}^{-1} \text{ cm}^{-2}$ as reported in (Chin & Cheong, 2011; C. C. Lin et al., 2014). However, $D_{it} \sim 10^{13} \text{ eV}^{-1} \text{ cm}^{-2}$ obtained in this investigation is relatively lower than the Sm₂O₃/Si interface $D_{it} \sim 10^{14} \text{ eV}^{-1} \text{ cm}^{-2}$ fabricated using sputtering process followed by thermal oxidation in O₂ ambient (Goh et al., 2017b).



Figure 4.16: The average interface trap density D_{it} of Sm₂O₃/IL/Ge gate stack for different oxidation and nitridation temperature (300 – 500) °C.

The C-V measurement was also performed at various frequencies of 50 kHz, 100 kHz and 500 kHz to explore the accumulation and inversion trend for different oxidized/nitride sample at (300 - 500) °C. According to Fig. 4.17 frequency dispersion phenomenon is observed in both accumulation and inversion region for all the investigated sample. Strong frequency dispersion at accumulation region has been observed for 300 °C sample has been where the accumulation capacitance dropped abruptly at higher frequency of 500 kHz. Meanwhile, 400 °C sample revealed the least frequency dispersion. Kouda et al., (2012) reported that frequency dispersion phenomenon at the accumulation region is related to the frequency dependent changes of dielectric constant caused by the oxygen vacancies (Kouda et al., 2012). According to this notion, it can be ascribed that 300 °C oxidation/nitridation condition favours the generation of more oxygen vacancies due to GeO (g) desorption causing dispersive accumulation capacitance during the frequency dispersion. Hence, it can be suggested that more defective Sm₂O₃ film was generated at this oxidation/nitridation temperature. On the other hand, small frequency dispersion in the accumulation region for 400 °C sample indicates that Sm₂O₃ has gained structural uniformity due to reduction of oxygen vacancies, which is caused by the barrier role of interfacial Sm-O-Ge during the oxidation/nitridation process. Moreover, presence of interfacial traps triggers the variation in the weak inversion region during the frequency dispersion (Zhao et al., 2018). Accordingly, the lower interface trap density obtained for 400 °C sample contributed to generate small frequency dispersion in the inversion region than those of 300 °C and 500 °C sample. An anomalous hump/kink has been observed in the weak inversion region for all the investigated samples which is attributed to the presence of large interfacial trap densities.

Based on the findings of C-V measurement analysis it is obvious that, 400 °C is the optimum oxidation/nitridation condition for Sm₂O₃/Ge gate stack which enforces the alleviation of defect/traps in the bulk Sm₂O₃ and as well as in the Sm₂O₃/IL. Therefore, it can be suggested that 400 °C oxidation and nitridation temperature shows the best C-Vcharacteristics.



Figure 4.17: *C-V* characteristics measured at different frequencies of 50 kHz, 100 kHz and 500 kHz for Sm₂O₃/IL/Ge gate stack at various oxidation/nitridation temperature (300 – 500) °C.

4.3 Influence of Various Thermal Oxidation/nitridation Durations on Sputtered Sm/Ge in N₂O ambient.

4.3.1 XRD Analysis

Fig. 4.18 displays XRD pattern at a close-up 2 θ range (10° - 60°) obtained for different thermal oxidation/nitridation durations (5-20) min in order to identify the crystal structure and crystallinity of the chemical compounds formed during oxidation/nitridation process. For all the investigated oxidation/nitridation durations three peaks were spotted at different 20 angle of 27.16°, 47.28°, 48.26° and 54.04° corresponding to trigonal Sm₂O₃ (tri-Sm₂O₃) phase (100), (003), and (103). Besides, another weaker Tri-Sm₂O₃ peak at 48.26° with a plane orientation (110) was spotted for both 15 and 20 min excluding 5 and 10 min. Two additional blunt diffraction peaks were also found at 31.15° and 56.36° belonging to (101) and (200) planes, respectively for only 15 min. The trigonal phase of Sm₂O₃ fitted well with the ICSD reference code no. 647455. Among the tri- Sm_2O_3 peaks (003) plane is suggested to be the preferential plane orientation for all the samples. Furthermore, the tetragonal GeO₂ (tetra - GeO₂) diffraction peaks appeared at 29.15°, 42.69° and 57.17° revealing phase orientation (110), (111) and (211), respectively. Another GeO₂ peak of trigonal phase (tri-GeO₂) was noticed for all the samples at 39.22° which corresponds to (102) plane. Meanwhile one addition tri-GeO₂ was also identified at 19.36° which belongs to (100) for 5,10 and 20 min except 15 min. The preferential plane orientation for tetra-GeO₂ and tri-GeO₂ is suggested to be (110) and (102)planes, respectively. The ICSD reference code of 158,593 and 53870 ensures the formation of tetra-GeO₂ and tri-GeO₂, respectively. Two diffraction peaks of cubic Ge₃N₄ (c-Ge₃N₄) were found at diffraction angle 31.72° and 31.76° for 15 and 20 min with plane orientation (211) and (311), respectively. Because of amorphous phase no visible Ge_3N_4 peaks was exposed for 5 and 10 min. The (311) plane is recommended as the preferential plane orientation for c-Ge₃N₄. The peak of c-Ge₃N₄ was verified from ICSD reference code no. 156338. Based on XRD diffraction patterns growth of Sm_2O_3 dielectric has been verified. Besides, the XRD patterns also suggested the growth of an interfacial layer (IL) due to detection of GeO_2 and Ge_3N_4 diffraction peaks. The presence of IL has been further verified in later section 4.3.2 and 4.3.3.

Figure 4.19 displays the intensity variation of preferential plane orientations of tri-Sm₂O₃ (003), tetra-GeO₂ (110) and tri-GeO₂ (102) to verify the impact of different oxidation/nitridation durations on the crystallinity of Sm₂O₃/IL. Regarding, Sm₂O₃ it is observed that the intensity increased gradually with the increase of oxidation durations. It is well known that the increase of intensity suggests the increment of crystallinity. Accordingly, the Sm₂O₃ compound is less crystalline for 5, 10 and 15 min since these durations induced the lower intensities. Meanwhile the most crystalline Sm₂O₃ structure is suggested for 20 min because the intensity of this duration increases substantially compared to other oxidation/nitridation durations. Both tetra-GeO₂ (110) and tri- GeO₂ (102) revealed similar intensity trend for all the samples. It is observed that, for GeO₂ phases the intensity increased to highest for 10 min while decreased subsequently for 5, 20, and 15 min. Consequently, GeO₂ becomes less crystalline at 15 min while the 10 min caused the most crystalline GeO₂ structure. On the basis of above consequences, it is predicted that 15 min oxidation/nitridation duration formed more amorphous interfaces since both Sm₂O₃ and GeO₂ is less crystalline at these durations.



Figure 4.18: XRD diffraction patterns for sputtered Sm/Ge film thermally oxidized/nitrided at several durations (5 - 20) min.



Figure 4.19: Intensity trend of preferential plane orientations of tri-Sm₂O₃ (003), tetra-GeO₂ (110) and tri- GeO₂ (102) as function of thermal oxidation/nitridation durations (5 - 20) min.

4.3.2 Raman Analysis

Fig. 4.20 illustrates the findings of Raman spectrums for different thermal oxidation/nitridation durations (5-20) min. The peak appeared at 300.95 cm⁻¹ (Batra, Kabiraj, & Kanjilal, 2007; Lei et al., 2018) is denoted as Substrate Ge for all the investigated oxidized/nitrided samples. Three peaks corresponded to Sm₂O₃ was spotted at 162.41 cm⁻¹ (Jiang, Liu, Lin, Li, & Li, 2013; Mandal, Sarkar, Deb, & Chakrabarti, 2014), 177.62 cm⁻¹ (Jiang et al., 2013), 411.20 cm⁻¹ (Hongo, Kondo, Nakamura, & Atou, 2007; Mandal et al., 2014). Meanwhile, the peaks of GeO₂ were identified at 221.72 cm⁻¹ (Lignie, Hermet, Fraysse, & Armand, 2015), 517.32 cm⁻¹ (Kartopu et al., 2003) for all the samples while the GeO₂ peaks at 445.44 cm⁻¹ (Kartopu et al., 2003) was detected for 5,10 and 20 min except 15 min. A single peak corresponding to Ge₃N₄ was identified at 459.44 cm⁻¹ (Deb, Dong, Hubert, McMillan, & Sankey, 2000; Lei et al., 2018) for all the investigated oxidation/nitridation durations. However, an additional unknown peak was detected at 353.12 cm⁻¹ for extended durations of 15 and 20 min. Since this peak neither corresponds to Sm_2O_3 , GeO_2 , or Ge_3N_4 and so it is inferred as interfacial $Sm_xGe_yO_z$ following the XPS spectrums of later section 4.3.3 where existence of Sm_xGe_yO_z was found at 15 and 20 min. Although, Sm-O-Ge peaks was not found in XRD patterns which is predicted due to its amorphous crystal structure. The inset image of Fig. 4.20 shows the intensity of Sm_2O_3 , GeO₂ and Ge₃N₄ at 177.62 cm⁻¹ 517.32 cm⁻¹ and 445.44 cm⁻¹, respectively. It is noticed that Sm₂O₃ and Ge₃N₄ increases at extended durations 15 and 20 min where the highest intensity of these compounds was produced at 15 min' oxidation/nitridation duration. Meanwhile, the generation of GeO₂ soaring higher at shorter durations of 5 and 10 min and reduced at elevated durations of 15 and 20 min where 10 min induced the highest intensity of GeO₂. The observation of RAMAN spectrums is complementary with the findings of XPS and XRD.



Figure 4.20: Raman Spectrums for sputtered Sm/Ge film thermally oxidized and nitrided under different oxidation/nitridation durations (5 - 20) min.

4.3.3 XPS Analysis

The evolution of core level narrow scan spectra of Sm 3d, Ge 3d and O1s at different thermal oxidation/nitridation durations 5, 10, 15 and 20 min were conducted to investigate the chemical bonding states configurations of Sm_2O_3/Ge stack. The narrow scan spectrums are shown in Fig. 4.21, 4.22 and 4.23.

Fig. 4.21 illustrates the variation of Sm 3d spectra at different oxidation/nitridation durations (5 - 20) min. From Fig. 4.21 it is apparent that, the line position of Sm 3d centroid peaks for 5 and 10 min oxidation/nitridation shifted negatively by 0.28 eV and 0.32 eV in the direction of low binding energy (B.E) with respect to reported Sm₂O₃ reference B.E of Sm 3d = 1084.0 eV (C. C. Lin et al., 2014). Primarily, the negative shift of Sm 3d spectra along low B.E implies that, the interfacial layer is deficient of Sm-O-Ge content during shorter oxidation/nitridation durations of 5 and 10 min. Conversely, when the oxidation/nitridation durations is extended to 15 and 20 min the reference line position of Sm 3d centroid peaks shifted positively by 0.23 eV and 0.18 eV in the direction of higher (B.E). This outcome clearly implies the creation of Sm-O-Ge content close to the dielectric interface within the interfacial region during the extended oxidation/nitridation of 15 and 20 min. The generation of interfacial Sm-O-Ge compound for these durations can be attributed due to partial exclusion of (Sm-O-Sm) sub-oxide configuration. Since Sm atoms are less electronegative than Ge hence, the catalytic oxidation of Ge atoms substituted Sm atoms by penetrating into Sm₂O₃ causing spontaneous growth of Sm-O-Ge content near the dielectric interface at elevated oxidation/nitridation durations (C. C. Lin et al., 2014). Regarding the Sm-O-Ge content deficiency for 5 and 10 min it is presumed that, due to shorter oxidation/nitridation time the activation energy of Ge atoms from GeO₂ network or substrate Ge was inadequate to initiate the catalytic oxidation of Ge atoms with Sm_2O_3 interface. Consequently, creation of Sm-O-Ge compound was prevented.

Fig. 4.21 also illustrates the deconvoluted compounds of Sm 3d peaks in order to inspect the influence of different oxidation/nitridation durations on thermodynamic stability and evolution of Sm₂O₃ dielectric interface. Typically, the peak corresponding to Sm₂O₃ appears at lower B.E of 1083.90 eV while Sm–O–Ge peaks were found at relatively higher B.E of 1086.02 eV (C. C. Lin et al., 2014). Accordingly, a single peak

corresponding at lower B.E of 1083.76 eV and 1083.63 eV appeared for 5 and 10 min, respectively is assigned as Sm₂O₃. On the other hand, the 15 and 20 min composed two peaks where the former peaks represented as Sm₂O₃ which emerged at lower B.E region of 1083.93 eV and 1083.81 eV, respectively. Meanwhile the latter peak was designated for Sm–O–Ge which appeared at higher B.E of 1085.23 eV and 1085.19 eV for 15 and 20 min samples, respectively. Comparing all the investigated durations, the peak corresponding to Sm–O–Sm exhibited the highest integrated area at 15 min followed by 20 and 5 min while the lowest integrated area of Sm-O-Sm was observed for 10 min oxidation/nitridation. This observation suggested that 10 min oxidation/nitridation condition accelerates more pronounced out-diffusion of O atoms from dielectric interface itself forming thermally unstable sub-stochiometric Sm₂O₃ film. Contrariwise, 15 min oxidation/nitridation is more convenient to inhibit the out-diffusion of O atoms which stimulated more stochiometric Sm₂O₃ film. In addition, creation of Sm–O–Ge interlayer with relatively higher B.E also contributed to the stabilization of Sm₂O₃ film for this duration. Although, both 15 and 20 min generated Sm–O–Ge interlayer but 15 min revealed better stoichiometry of Sm₂O₃ which can be attributed to the uniform in-diffusion of Ge atoms into Sm₂O₃ dielectric layer causing balanced elimination of (Sm-O-Sm) sub-oxide during Sm–O–Ge formation.

Thus, the findings of Sm 3d suggests that, merely Sm_2O_3 was generated during 5 and 10 min while co-existence of Sm_2O_3 dielectric film and interfacial Sm-O-Ge content were evident when oxidation/nitridation durations elevated to 15 and 20 min. Furthermore, 15 min oxidation/nitridation induced more stochiometric Sm_2O_3 dielectric interface and relatively stable Sm-O-Ge interfacial compound. The formation of Sm_2O_3 can be governed by the Eq. 4.13 and Eq.14 where sputtered Sm^+ reacted with the atmospheric N₂O for all the investigated samples to produce Sm_2O_3 dielectric interface on the uppermost surface. Meanwhile, the generation of spontaneous Sm-O-Ge content across the interfacial layer for 15 and 20 min took place according to the vigorous chemical reaction of Eq. 4.14 by the penetration of Ge atoms into Sm₂O₃ interface.

$$Sm^+ + N_2O \rightarrow SmO^+ + N_2$$
 (Equation 4.13)

$$\text{SmO}^+ + \text{N}_2\text{O} \rightarrow \text{Sm}_2\text{O}_3 + \text{N}_2$$
 (Equation 4.14)

$$Sm_2O_3 + Ge \rightarrow SmGeO_x$$
 (Equation 4.15)



Figure 4.21: Deconvoluted narrow scan spectrums of Sm 3d at different oxidation and nitridation durations (5 – 20) min for Sm₂O₃/Ge stack.

Fig. 4.22 signifies the evolution of several interfacial compounds obtained from Ge 3d spectrums at various oxidation/nitridation durations (5 – 20) min. According to Fig. 4.22, a single peak composing mixture of various interfacial compounds have been deconvoluted with reference to elemental substrate Ge^0 at 29.6 eV (Mitrovic et al., 2014). The interfacial chemical bonding states were deconvoluted into three Ge suboxides (GeO_x) of Ge¹⁺ (Ge₂O), Ge²⁺ (GeO), Ge³⁺ (Ge₂O₃), and a native oxide Ge⁴⁺ (GeO₂) at B.E shift of +0.8 eV, +1.8 eV, +2.8 eV and > +3 eV, respectively (J. Song et al., 2009). Another additional interfacial chemical bonding state of Ge* fitted well for 15 and 20 min at B.E shift of +2.4 eV. It has been reported that, rare earth germanate typically appears between +1.7 to +2.8 eV owing to nearest neighbour effect which shows distinct difference with respect to GeO₂ of > +3eV shift (Mitrovic et al., 2014). Hence, the Ge* content found between Ge²⁺ and Ge³⁺ is designated as Sm–O–Ge. However, no such Sm–O–Ge peak was found for 5 and 10 min samples which is consistent with the Sm 3d spectrums.

According to Fig. 4.22, the content of Ge^{1+} (Ge₂O) and the native oxide Ge^{4+} (GeO₂) was formed for all the investigated samples. Meanwhile, Ge^{3+} (Ge₂O₃) was exposed for 5,10 and 20 min samples excluding 15 min. However, it is noteworthy that desorption of volatile GeO compound appeared for 5 and 10 min samples while GeO was deficient during oxidation/nitridation of 15 and 20 min samples. The above findings imply an amorphous IL structure beneath the Sm₂O₃ dielectric layer involving Ge₂O, GeO, Sm–O–Ge, Ge₂O₃, and GeO₂ contents. However, the growth of these IL compounds shows variance in densities for different oxidation/nitridation durations which effected the thermal stability of Sm₂O₃/Ge stack. From Fig 4.22, it is also observed that the growth of sub-stoichiometric unstable Ge suboxides (GeO_x) compounds is greater for 10 min followed by 5 min sample. On the contrary, the growth of suboxides (GeO_x) shrinks at 20 min and the lowest integrated area was recorded at 15 min sample. Furthermore, in several literatures it has been asserted that among all the Ge oxides excessive growth of native

 GeO_2 (Ge^{4+}) and GeO (Ge^{1+}) compounds would cause an abrupt thermodynamic instability and degradation of Ge based MOS devices (Juan. Gao et al., 2017; D. Wang, He, Fang, et al., 2019). Since, lowering of GeO_2 (Ge⁴⁺) and the deficiency of GeO (Ge²⁺) was observed for 15 and 20 min accordingly, these durations expose more thermally stable IL interface than those of 5 and 10 min sample. The reason of better thermal stability for 15 and 20 min durations is due to the evolution of Sm-O-Ge content which suppressed the desorption of volatile GeO and compressed the GeO_x and GeO_2 content for these samples. Regarding the GeO suppression mechanism, it is predicted that formerly an oxygen deficient SmGeO_x were generated which trapped the out-diffusing GeO species and as final product densified Sm_xGe_vO_z was produced by preserving oxygen loss (C. H. Lee et al., 2011; H. Wang et al., 2012). Amongst all the samples 10 min duration prevailed the highest integrated area of GeO_2 (Ge⁴⁺) and GeO (Ge²⁺) species suggesting the most thermally unstable interfacial interface. The increase of GeO_2 (Ge^{4+}) and GeO (Ge^{1+}) might be associated with the interdiffusion and re-arrangement mechanism of oxygen substances and unsaturated Ge atoms during thermal oxidation/nitridation and cooling process. Hence, 10 min oxidation/nitridation condition would decline the electrical features of the Sm₂O₃/Ge stack. Now, comparing 15 and 20 min samples, Ge₂O (Ge¹⁺) and Ge₂O₃ (Ge³⁺) content decreases while a slight increase in Sm-O-Ge content is observed for 15 min sample. This finding suggests that 15 min oxidation/nitridation increases the activation energy of Ge promoting out-diffusion of Ge atoms to generate denser Sm-O-Ge which controlled the distribution of oxygen substances in such way so that GeO_x growth was minimized. Furthermore, intense GeO_2 (Ge⁴⁺) peak is noticeable at this duration alike other samples, but the peak shifted toward higher B.E while the density slightly dropped. This outcome suggests that GeO_2 (Ge⁴⁺) became more stochiometric when the oxidation/nitridation duration reached 15 min. The formation of stochiometric GeO₂ can also be attributed to the initial oxygen deficient SmGeO_x which later regulated the propensity

of O and Sm distribution to retain the integrity of interfacial native GeO₂. Thus, among all the oxidized/nitrided samples 15 min duration is more suitable condition to achieve thermodynamically stable interfacial layer with less sub-stochiometric Ge sub oxides defects and more stochiometric GeO₂.

In summary, based on Ge 3d spectra the interfacial oxidation/nitridation mechanism of forming interfacial oxides such as GeO_x , GeO_2 and desorption process of GeO can be expressed by the following hypothetical chemical reaction. It should be noted that, the interfacial reactions of Eq. 4.16 and Eq.17 occurs for all the samples while Eq 4.18 takes place for 5, 10, and 20 min oxidation/nitridation. The desorption process of volatile GeO has been hindered for the 15 and 20 min following Eq. 4.19.

$$Ge + 2N_2O \rightarrow GeO_2 + 2N_2$$
 (Equation 4.16)

$$3\text{GeO}_2 + 3\text{Ge} \rightarrow 6\text{GeO}$$
 (Equation 4.17)

$$4Ge + 2O_2 \rightarrow Ge_2O + Ge_2O_3 \qquad (Equation 4.18)$$

 $SmGeO_x + GeO \rightarrow Sm_xGe_yO_z + GeO_x$ (Equation 4.19)



Figure 4.22: Deconvoluted Ge 3d narrow scan spectrums of oxidized/nitrided sputtered Sm/Ge for different oxidation/nitridation durations (5 – 20) min.

Fig. 4.23 depicts the evolution of O1s spectra as function of several oxidation/nitridation durations (5 – 20) min. The O1s spectra shows that all the investigated samples can be deconvoluted into two common peaks at 530.40 eV and 532.0 eV while an extra peak at 531.80 eV was found for 15 and 20 min samples. The peaks at 530.40 eV, 531.80 eV and 532.00 eV have been allocated for Sm–O–Sm, Sm–O–Ge and Ge–O, respectively (Pan & Huang, 2010; D. Wang, He, Fang, et al., 2019). From O1s spectrum it is apparent that, the oxygen content within the Sm–O–Sm configuration enhanced largely at 15 min oxidation/nitridation followed by 20 and 5 min while substantial decrease is noticed for 10 min. This observation is supportive with the Sm 3d spectra, which again confirms that 15 min oxidation/nitridation condition significantly promotes the growth of stochiometric Sm₂O₃ while 10 min condition is less suitable to form stochiometric Sm₂O₃ interface. Considering, the Ge-O content an opposite growth phenomenon was spotted where the Ge-O species showed the largest expansion for 10 min followed by 5 min sample. Since the O1s peak corresponding to Sm-O-Sm content lessened and Ge-O content increased for 5 and 10 min, hence it is obvious that the disproportionate outdiffusion of O atoms from Sm₂O₃ network contributed to enhance Ge-O species for these samples. As a result, the IL region grown beneath the Sm₂O₃ film is densified by unstable GeO_x and GeO₂ for 5 and 10 min oxidation/nitridation which is complementary with Ge 3d spectra. Meanwhile, O1s corresponding to Ge-O species decreased subsequently for 20 and 15 min which signifies the reduction of GeO_x and GeO₂. The lowest Ge–O content for 15 min in O1s spectra suggests that among all samples the smallest amount of oxygen atoms diffused from Sm_2O_3 itself at this duration. Thus, based on both O1s and Ge 3d spectra it is obvious that 10 min is more convenient for growing Ge-O compounds while 15 min oxidation/nitridation impedes the growth the Ge-O contents. In addition, the Sm-O-Ge content found in O1s spectra also follows identical growth phenomena observed in Sm 3d and Ge 3d spectra where this content maximized at 15 min oxidation/nitridation condition.

The extent of oxidation/nitridation level was estimated by plotting the B.E of Sm_2O_3 extracted from Sm 3d spectrums as function of various oxidation/nitridation durations 5, 10, 15 and 20 min displayed in Fig. 4.24. According to Fig. 4.24 the B.E for shorter oxidation/nitridation of 5 and 10 min and the longest oxidation/nitridation of 20 min emerged below the baseline of full oxidation level indicating under-oxidized condition. Since, 10 min sample is distributed in the lowest distance below full oxidation level hence, structural quality of Sm_2O_3 at this duration will be most defective due to deep

under-condition which adversely effects the electrical properties. Among all the investigated samples the longer oxidation/nitridation duration of 15 min emerged near the fulloxidation baseline.



Figure 4.23: Deconvoluted core level narrow scan spectrums of O1s for different oxidation/nitridation durations (5 – 20) min.



Figure 4.24: Degree of oxidation/nitridation level as function of different durations.

4.3.4 Band Alignment

The band offset value of high-*k* oxide between substrate material is one of the critical issues in utilization of high-*k* material for MOS devices. The band offset value of high-*k* oxide should be minimum 1eV with respect to semiconductor substrate material (Goley & Hudait, 2014; Robertson & Wallace, 2015). Hence, the band alignment of Sm₂O₃/IL/Ge for various oxidation/nitridation has been determined from O1s plasmon loss XPS and valance band, (VB) spectra as described earlier in section 4.2.4. Accordingly, the valance band offsets (VBO), ΔE_v of (Sm₂O₃ + IL) and conduction band offset (CBO), ΔE_c (Sm₂O₃+IL) on substrate Ge are evaluated following Eq. 4.1 and Eq. 4.2, respectively. The valance band edge, E_v (Sm₂O₃ + IL) has been measured by an extrapolation of leading negative slope edge of VB spectrums towards the horizontal baseline of substrate Ge baseline as represented in Fig. 4.25a. The intercept tip is considered as E_v (Sm₂O₃ + IL) which was measured 2.94 eV, 2.72 eV, 3.08 eV and 3.04 eV for oxidation/nitridation durations of 5, 10, 15 and 20 min, respectively. According to Eq. 4.1. VBO, ΔE_v (Sm₂O₃ + IL) with respect to substrate Ge for 5, 10, 15 and 20 min oxidation/nitridation durations are 1.90 eV, 1.82 eV, 2.60 eV and 2.43 eV, respectively.

The energy bandgaps, E_g measured using O1s plasmon loss spectrums obtained from respective XPS survey scan of 5, 10, 15 and 20 min are 5.41 eV, 5.11 eV, 6.35 eV and 6.02 eV, respectively as depicted in Fig. 4.25b. Meanwhile, Fig 4.26 presents the combined band alignment mapping at different oxidation/nitridation durations for Sm₂O₃/IL/Ge stack durations. The CBO, ΔE_c (Sm₂O₃ + IL) measured on the substrate Ge for thermally oxidized/nitrided samples of 5,10,15 and 20 min are 1.90 eV, 1.82 eV, 2.60 eV and 2.43 eV, respectively as in Fig. 4.26. In response to the mapping of band alignment it is apparent that 15 min induced the biggest ΔE_c and ΔE_v followed by 20 and 5 min while decreased to the smallest at 10 min. For high-*k* based MOS devices increment of CBO, ΔE_c and VBO, ΔE_v is considered as an essential aspect which regulates reduction of gate leakage current density (J. Gao et al., 2017). The band offset mapping also indicates 5 min induced more asymmetrical mapping while 15 min induced near-symmetrical mapping among all the samples. The asymmetrical mapping between ΔE_c and ΔE_v would also contribute to increase the gate leakage current (He, Gao, et al., 2014). Therefore, for 15 min the enhanced ΔE_c and ΔE_v with near-symmetrical mapping would contribute to improve the *J-E* features. Meanwhile, the decreased ΔE_c and ΔE_v of 10 min and asymmetrical band offset mapping of 5 min would be a concerning factor in deteriorating device performance. The increase of ΔE_c and ΔE_v with near-symmetrical band alignment for 15 min is related to better interfacial interface caused by the reduction of GeO_x structural defects and other oxygen associated defects observed in XPS analysis. On the contrary, for 5, 10 and 20 min the decrease of ΔE_c and ΔE_v is associated with higher densities defect gap states generated inside the bandgap of Sm₂O₃/IL through volatilization of GeO (g) and diffusion of interfacial GeO_x defects. Due to reduction of ΔE_c and ΔE_v the bandgap of 5, 10 and 20 min dropped by 0.84 eV, 1.14 eV and 0.23 eV, respectively. Since 10 min generated the smallest ΔE_c and ΔE_v causing the lowest bandgap. Furthermore, for all the investigated samples it is noticeable that both ΔE_c and ΔE_v is larger than 1 eV which signifies the eligibility of Sm₂O₃/IL/Ge structure to be a potential contender for high-k based MOS devices.



Figure 4.25: Estimation of valance band offset (ΔE_v) and energy bandgap (E_g) for Sm₂O₃/IL/Ge structure at various oxidation/nitridation durations (5 – 20) min (a) Valance band edge, E_v and VBO extraction from XPS valence band spectra. (b) Energy band gap deduction from oxygen plasmon loss XPS spectra.



Figure 4.26: Band Alignment mapping of (Sm₂O₃ + IL) with respect to Ge substrate for various oxidation/nitridation durations (5 - 20) min.

4.3.5 HRTEM Analysis

The compatibility of the fabricated Sm₂O₃ /Ge stack with respect to ongoing MOS scaling has been verified by conducting HRTEM analysis. In Fig. 4.27 HRTEM cross sectional images disclosed double stacked layer interface for Sm₂O₃/Ge based gate stack. The uppermost surface is regarded as Sm₂O₃ dielectric film interface since it is positioned at larger distant with respect to Ge wafer while the middle position stack layer is denoted as interfacial layer (IL) as it is confined at the nearer interface from the Ge wafer. On the basis of XRD, Raman and XPS analysis the IL layer is composed of various chemical bonding such as Ge–O (i.e., GeO₂ and GeO_x), Ge–N and Sm–O–Ge. In Fig. 4.27 relatively thinner physical oxide thickness $(Sm_2O_3 + IL)$ of 6.00 nm was formed for 15 min duration while the total physical oxide thickness increased by 0.91 nm when the duration was extended to 20 min at value of 6.91 nm. It is notable that, bulk Sm₂O₃ thickness shrank from 3.56 nm to 3.20 nm while the IL thickness incremented from 2.44 nm to 3.71 nm for 15 and 20 min, respectively. Accordingly, the reduction of total thickness at 15 min is caused by inhibition of IL growth which instigated to increase the bulk Sm₂O₃ thickness. Conversely, for 20 min the total oxide becomes thicker due to prompted growth of IL which impeded the growth of high k bulk Sm₂O₃.

Thus, the enhancement of bulk Sm_2O_3 high-*k* interface for 15 min is expected to improve the dielectric constant *k* value of the total gate stack. Meanwhile, the increased growth of IL for 20 min would impact negatively on dielectric constant, *k* and other electrical features. Additionally, total ($Sm_2O_3 + IL$) physical thickness of 5 and 10 min was extracted at value of 4.25 nm and 5.09 nm, respectively from the linear regression of total oxide thickness with respect to 15 and 20 min samples as depicted in Fig. 4.28. It has been assumed that ($Sm_2O_3 + IL$) thickness incremented linearly with extension of oxidation/nitridation durations.

Furthermore, diffraction patterns of Sm₂O₃, IL and Ge substrate for both 15 and 20 min was generated employing Fast Fourier Transformation (FFT) process on HRTEM images regulated by ImageJ Software. The diffraction patterns acquired from FFT is shown as inset image in Fig. 4.27. According to FFT it is apparent that both 15 and 20 min prompted the formation of amorphous bulk Sm₂O₃, and amorphous IL which is confirmed from the exposure of cloudy spotted FFT diffraction patterns. Additionally, detection of multiple spotted diffraction patterns from FFT suggested the polycrystalline orientation of substrate Ge for both the durations.



Figure 4.27: HRTEM cross-sectional morphology for thermally oxidized/nitrided 15 and 20 min samples.



Figure 4.28: Linear regression plot of total oxide thickness of Sm₂O₃/IL/Ge as function of various oxidation/nitridation durations (5 – 20) min.

4.3.6 C-V Characteristics

Fig. 4.29 displays a high frequency (1 MHz) capacitance-voltage (C-V) plot obtained at room temperature for several thermal oxidation/nitridation durations (5 - 20)min. The measurements were taken using both forward and reverse gate bias condition by sweeping bidirectionally between -1 to +2V.



Figure 4.29: *C-V* features recorded at high frequency 1Mhz for $Sm_2O_3/IL/Ge$ based MOS capacitors oxidized/nitrided at different durations (5 – 20) min.

In Fig. 4.29 the accumulation capacitance decreases substantially when the oxidation/nitridation duration is extended from 5 to 10 min but increases with a longer duration of 15 and 20 min. However, a significant drop of accumulation capacitance is observed when the duration is extended from 15 to 20 min. Overall, 15 min oxidation/nitridation condition exhibited the highest oxide capacitance while 10 min induced the lowest. On the basis of Eq. 4.10, the difference in accumulation capacitance level is associated with the total physical oxide thickness and dielectric constant value (*k*) since, $C_{ox} \propto \frac{k}{t_{ox}}$ when *A* and ε_o are constant. Accordingly, the oxide capacitance increases with shrinking of oxide thickness while the increment dielectric constant also attributes to increase the capacitance level. Moreover, the existence of IL has a crucial effect on the overall capacitance of a stack layered capacitor (Wilk et al., 2001).

According to earlier HRTEM Fig. 4.27, it was obvious that15 min owned thinner total oxide thickness as well as a thinner IL in counter to 20 min which might influenced to enhance the accumulation capacitance for 15 min oxidation/nitridation condition. Although, the total physical oxide thickness of 5 and 10 min is estimated to be lower than 15 and 20 min (shown in Fig. 4.28) but yet exhibited lower oxide capacitance. This observation suggested that the accumulation capacitance does not only varies due to oxide thickness but also depends on the interface quality and dielectric constant k value. For further clarification, the effective dielectric constant, C value of Sm₂O₃/IL/Ge gate stack were acquired from C-V curve using Eq. 4.10. According to Fig. 4.31 the k_{eff} values are 17.27 15.82, 31.19, 29.96, under 5,10, 15 and 20 min oxidation/nitridation, respectively. The findings of effective dielectric constant k_{eff} value and oxide capacitance reveal a proportional relationship as interpreted in Eq. 4.10. As observed in XPS, 10 min revealed the most thermally unstable interface and promoted the densification of low-k interfacial compounds (i.e., GeO₂ and GeO_x) which induced smallest k_{eff} value leading to lowest accumulation capacitance. Conversely, 15 min oxidation/nitridation induced largest k_{eff} value causing highest accumulation capacitance which is due to escalation of high-k Sm_2O_3 and suppression of interfacial low-k compounds. Regarding the k value of Sm_2O_3 MOS gate stack Pan & Huang, (2010) reported $k \sim 8.1 - 14.3$ for Sm₂O₃/(p-type) Si prepared using combination of RF sputtering rapid thermal annealing in (Ar/O₂) ambient. In other study, Chin & Cheong, (2011) reported the $k \sim 5.4 - 7.2$ for Sm₂O₃/(n-type) Si fabricated using furnace annealing in (N_2/H_2) ambient. However, the k_{eff} value obtained in this study does not reveals the same range. It has been reported that dielectric constant is largely contingent on type of electrode and substrate, physical characteristics of substrate-oxide interface and deposition procedure (Wong & Cheong, 2012b). Since, total physical thickness including interfacial layer was considered k_{eff} value estimation and amorphous Sm₂O₃ growth was observed in this investigation which might be one of the
possible reasons for k_{eff} value deviation from Sm₂O₃/Si interface (Chin & Cheong, 2011; Pan & Huang, 2010) where crystalline Sm₂O₃ was formed. The *EOT* measured at the corresponding k_{eff} values were found to be in range of 0.75 – 1.25 nm for oxidized/nitrided Sm₂O₃/IL/Ge stack layer in several durations as presented in Fig. 4.30. The 15 min oxidation/nitridation exhibited the lowest *EOT* ~ 0.75 nm due to its highest k and relatively thinner IL.



Figure 4.30: Effective dielectric constant, k_{eff} and equivalent oxide thickness obtained for Sm₂O₃/IL/Ge stack with respect to different oxidation/nitridation durations (5 – 20) min.

Furthermore, in Fig. 4.29 a stretch-out phenomenon towards the positive voltage region has been observed for all the investigated samples which confirms the presence of interfacial defect states at conduction band edge of Ge (He, Gao, et al., 2014). The 10 min oxidation/nitridation condition executed the strong stretch-out phenomenon as well as non-quasi saturation at accumulation regime. Since XPS confirmed increment of Ge–O contents at 10 min hence both of these phenomenon's is mainly due to generation of higher density defect states caused by the transportation of trap charges induced from

Ge–O and Ge–Ge bonding into the CB edge of substrate Ge. Meanwhile, for 15 min sample the stretch-out effect has been reduced substantially which is due to better interfacial interface consisting of less Ge–O bonding and Ge point defects as observed in Ge 3d and O1s spectra of XPS.

The *C-V* measurements of Fig. 4.29 also reveals that the flatband voltage, V_{FB} moved towards positive gate voltage axis for all the oxidation/nitridation durations. The positively shifted V_{FB} signifies the existence of negative oxide trap charges at/near the edge of Sm₂O₃ film interface (J. Gao et al., 2017; Wong & Cheong, 2012c). Amongst all the oxidation/nitridation durations 10 min persuaded the maximum V_{FB} shift followed by 5 and 20 min while 15 min oxidation/nitridation generated minimal shift. Hence, 10 min will generate more negative oxide charges while 15 min will reduce the amount of negative oxide charges which is depicted in Fig. 4.31. Typically, in MOS devices flatband voltage shift is caused by the oxygen vacancies within the high-*k* oxide interface. Accordingly, the extended V_{FB} shift for 5 and 10 min sample is caused by volatilization of GeO (g) as observed in Ge 3d spectra which enhanced oxygen vacancies into dielectric Sm₂O₃ film by adding extra negative oxide trap charges (C. C. Cheng et al., 2007). The smallest voltage shift for 15 min may be implicated by the presence of negative charged (singly or doubly) O interstitial. The effective oxide charge at different oxidation/nitridation/nitridation durations has been estimated by Eq. 20.

$$\boldsymbol{Q}_{eff} = \frac{\Delta \boldsymbol{V}_{FB} \boldsymbol{C}_{ox}}{qA}$$
(Equation 4.20)

Where, Q_{eff} represents effective oxide charge, ΔV_{FB} defines the difference of flatband voltage between ideal and experimental curve and q specifies for electronic charge. From Fig. 4.31 it is apparent that 15 min induced the lowest Q_{eff} of -2.88×10^{12} cm⁻² and when the oxidation/nitridation reached 10 min the negatively oxide charges increased drastically at Q_{eff} of -1.20×10^{13} cm⁻². The negatively charged Q_{eff} is attributed to the acceptor like traps at/near the Sm₂O₃ dielectric interface which could act as fixed charges when it is not sited in the edge of majority band (Ji, Xu, Lai, Li, & Liu, 2011; Wong & Cheong, 2012b). The Sm₂O₃/Ge gate stack fabricated at 15 min oxidation/nitridation condition could reduce the broken bonds and oxygen vacancies within the Sm₂O₃ film which preserves the homogeneousness of Sm₂O₃ interface causing diminution in Q_{eff} . Conversely, the 10 min increased the Q_{eff} due to degraded Sm₂O₃ interface quality caused by the severe O atoms out-diffusion from Sm₂O₃ film. The lowest $Q_{eff} \sim -$ 2.88 × 10¹² cm⁻² achieved for Sm₂O₃/Ge interface is significantly lower than Sm₂O₃/Si (Goh et al., 2017b) and Tm₂O₃/Ge (Žurauskaitėa et al., 2018).

Besides, fixed oxide charges the interface dipole formation can also contribute to the shift of flat-band voltage in the case of high-*k* dielectric stacked gate insulator (Kita & Kamata, 2017; Kita & Toriumi, 2009). It has been reported that flatband voltage shift due to dipoles are physically originated from the chemical bonding differences caused by the oxygen movement initiated from oxygen area density difference at interface within high-*k* oxide/interfacial oxide (Kita & Toriumi, 2009). It was pointed that oxygen movement between the high-*k* /IL interface induces Frenkel type defect of vacancy—interstitial pair which corresponds to form interface dipoles. However, the amount of dipole formation may change according to the process condition which effects the V_{FB} shifting. The contribution of interface dipoles in regulating V_{FB} shift can be explained from the difference of chemical bonding characteristics of dielectric Sm–O and interfacial Ge–O denoted in XPS and Raman analysis. Since strong deformation of Ge–O and Sm–O bonding's were observable at 10 min and so it is anticipated that the free energy of Sm₂O₃ and IL interface is increased activating more oxygen movement. As a result, anomalous coordination of negatively charged oxygen between Sm₂O₃ and IL (GeO₂ or GeO_x) has been enhanced inducing large amount of interface dipoles by maximizing the vacancy-interestitial pair leading to the largest V_{FB} shift. Meanwhile, it is predicted that reduction of flatband voltage (relatively negative shift) is caused by minimization of charge imbalance that corresponds to interface dipoles by lessening oxygen density abruptness in Ge–O and Sm–O interface (Kita & Kamata, 2017). The relatively negative V_{FB} shift for 15 min can also be attributed the minimum interface dipole formation due to denser Sm–O–Ge formation and comparatively more nitrogen incorporation which impeded imbalance charge generation caused by anomalous oxygen transmission mechanism from Sm₂O₃ towards GeO₂ or GeO_x interface. The interface dipole within high-*k*/high-*k* was not considered rather high-*k*/IL region was considered since deformation of covalent Ge–O could trigger much higher driving force in oxygen diffusion than those of Sm–O ionic bonds for high-*k* dielectrics (Kita & Toriumi, 2009).

A clearly visible hysteresis effects in the *C-V* curves have been noticed, ranging from 50 – 200 mv which is presented in Fig. 4.32. It was pointed that hysteresis effect typically occurs during the bidirectional swept when an electron filling a trap and leaving the traps does not coincide at same gate bias voltage which is caused by the existence of slow traps states (He, Liu, et al., 2014; Ke et al., 2018). The slow trap density, (Q_{it}) can be estimated using Eq. 4.11. Fig. 4.31 also shows that, 10 min sample generated the highest $Q_{it} \sim 3.44 \times 10^{12}$ cm⁻² while the lowest $Q_{it} \sim 1.41 \times 10^{12}$ cm⁻² occurred when the duration raised to 15 min. Accordingly, suppression of hysteresis for 15 min has been induced from generation of fewer slow trap states during the oxidation/nitridation process. The slow traps in the gate stack are generally caused by the (i) defective high-*k* interface (ii) defect states localized at/nearby high-*k*/IL interface and (iii) tunneling of charged carriers (electron or hole) from Ge wafer towards high-*k*/IL interface (C. C. Cheng et al., 2007; Ke, Yu, Chang, Takenaka, & Takagi, 2016). From XPS measurement it was established that 10 min oxidation/nitridation upsurged the oxygen content diffusion

from Sm₂O₃ itself which escalated the rapid chemical reaction between diffused Ge and O atoms causing higher defect states at/nearby Sm₂O₃/GeO_x interface. Consequently, triggering the slow traps states at this duration. Conversely, for 15 min barrier role of Sm-Ge-O inhibited such inter-diffusion leading to fewest slow traps. According to the STD findings, tunneling of both electron and hole towards Sm₂O₃/IL interface from substrate Ge increased the most for 10 min oxidation/nitridation while transmission of carrier charges lessened at 15 min. Besides, accumulation of interstitial N^+ atoms at high-k or near high-k/IL also induces additional slow trap states (Misra et al., 2006). Accordingly, the increase of STD for shorter durations 5 and 10 min directs that these oxidation/nitridation conditions promote the piling of N⁺ atoms which is asymmetrically distributed into the Sm₂O₃/IL or Sm₂O₃ interface. Meanwhile, it is suggested that longer duration of 15 and 20 min decreases extent of out-diffusing N⁺ atom and also maintains a homogenous distribution since compensation of STD is being observed. These indications are supportive with Raman and XRD analysis, which showed more intensified and crystalline Ge₃N₄ for 15 and 20 min due to less N⁺ atoms out-diffusion, while the non-uniform accumulation of N⁺ atoms induced an amorphous Ge₃N₄ with lowest intensity for 5 and 10 min. The kink and hump appeared at weak inversion regime for several oxidation/nitridation durations is associated with the additional slow trap states caused by N⁺ atoms distribution. The alteration of STD with respect to various durations exposed similar pattern with Q_{eff} . The $Q_{it} \sim 1.41 \times 10^{12}$ cm⁻² for 15 min obtained in this study is inferior to previously reported STD (Q_{it}) of Sm₂O₃/Si ~ 5.56 × 10¹² cm⁻² (Goh et al., 2017b).



Figure 4.31: Effective oxide charge, Q_{eff} and slow trap density, Q_{it} acquired from *C-V* plot of Sm₂O₃/IL/Ge gate stack at different oxidation/nitridation durations (5 – 20) min.



Figure 4.32: Hysteresis voltage variations of $Sm_2O_3/IL/Ge$ gate stack for various thermal oxidation/nitridation durations (5 - 20) min.

Furthermore, in Fig. 4.33 the average interface trap density D_{it} as function of trap energy level $(E_c - E)$ for different oxidation/nitridation durations (5 - 20) min were extracted from C-V curve using Terman method which is governed by Eq. 4.12. From Fig. 4.33 it was found that the average interface trap density D_{it} scattered between ~ $10^{13} - 10^{14} \text{ eV}^{-1} \text{ cm}^{-2}$ at trap energy level $(E_c - E) = 0.09 - 0.27 \text{ eV}$. Amongst all the oxidation/nitridation condition 10 min provides the highest $D_{it} \sim 2.49 \times 10^{14} \,\mathrm{eV^{-1} \, cm^{-1}}$ ² and the lowest $D_{it} \sim 4.84 \times 10^{13} \text{ eV}^{-1} \text{ cm}^{-2}$ at mid-gap of $(E_c - E) = 0.2 \text{ eV}$ was acquired for 15 min. Concerning oxide/Ge interfaces typically, D_{it} is enhanced due to degraded interface quality by the volatilization of GeO (g) and decomposition of GeO₂ (Matsubara et al., 2008; Zhao et al., 2018). The reduction of D_{it} for 15 min can be attributed to generation of higher density of Sm-O-Ge and Ge₃N₄ as observed in XPS and Raman spectra. The diffusion barrier role of Sm-O-Ge and Ge₃N₄ played a vital part to saturate the oxygen vacancies and in alleviation of un-bonded Ge caused by desorption and decomposition of GeO and GeO_2 . It is anticipated that, growth of denser Sm-O-Geand Ge₃N₄ at this duration restricted the intermixing of Ge and O atoms near the substrate Ge interface and within interfacial GeO₂ network. Eventually, prompted less defective retarded the expansion of sub-stochiometric Ge suboxides (GeO_x) and causing reduction of interface trap density. Due to the same reason 20 min oxidation/nitridation caused lower D_{it} than 5 and 10 min samples. On the contrary, the highest D_{it} for 10 min is caused by extreme GeO (g) volatilization and GeOx densified IL layer due to enhancement of oxygen related defects and Ge dangling bonds. Since this duration was not suitable for growing Sm-O-Ge. Besides, creation of sub-stoichiometric Sm₂O₃ also generated more charged traps confining spots located at/near Sm2O3/IL which included additional interfacial traps at this duration.

The inset image of Fig. 4.33 shows the total interface trap density D_{total} acquired by estimating the integrated area under D_{it} vs $(E_c - E)$. The pattern of D_{total} as function of different oxidation/nitridation durations follows analogous variation as of effective fixed oxide charge, (Q_{it}) and STD, (Q_{it}) . The lowest D_{total} attained for this investigation is 1.2×10^{13} cm⁻² during oxidation/nitridation of 15 min. In comparison with previous reports the minimum $D_{it} \sim 4.84 \times 10^{13}$ eV⁻¹ cm⁻² (at energy level, $E_c - E \sim 0.2$ eV) for this investigation is larger than post annealed Sm₂O₃/Ge interface $D_{it} \sim 5.10 \times 10^{11}$ eV⁻¹ cm⁻² (C. C. Lin et al., 2014) and multi stacked HfO₂/Al₂O₃/GeO₈/Ge interface $D_{it} \sim$ 2.0×10^{11} eV⁻¹ cm⁻² (R. Zhang et al., 2013) . Although, in this study relatively smaller $D_{it} \sim 10^{13}$ eV⁻¹ cm⁻² and $D_{total} \sim 1.2 \times 10^{13}$ cm⁻² has been integrated than the Sm₂O₃/Si thermally oxidized gate stack with a $D_{it} \sim 10^{14}$ eV⁻¹ cm⁻² and $D_{total} \sim$ 7.30×10^{13} cm⁻² (Goh et al., 2017b).

From the overall *C*-*V* features, it is noted that, $\text{Sm}_2\text{O}_3/\text{Ge}$ gate stack oxidized/nitrided at15 min could exhibit a high *k* value of 31.19 with a low *EOT* ~ 0.75 nm but expenses higher $D_{it} \sim 10^{13} \text{ eV}^{-1}\text{cm}^{-2}$ which is still inconvenient for MOS application hence, a D_{it} reduction strategy is required. In summary, it is suggested that 15 min thermal oxidation/nitridation durations possesses the best *C*-*V* features for Sm₂O₃/IL/Ge gate stack.



Figure 4.33: Average interface trap density D_{it} approximated with respect to trap energy level ($E_C - E$) for Sm₂O₃/IL/Ge gate stack developed under different oxidation/nitridation (5 – 20) min. Inset image presenting total interface density as function of different durations.

4.3.7 J-E Characteristics

Fig. 4.34 depicts the electrical features of gate leakage current density, versus electrical field *E* for the gate stacks investigated at different thermal oxidation/nitridation durations (5 – 20) min. In Fig. 4.34 two-step electrical breakdown field is noticed due to creation of interfacial layer (IL) interface. It is assumed that any of the interface either Sm₂O₃ or IL was broken at pre-mature electric field causing two step breakdown fields. The electrical field broken at lower stage is classified as soft breakdown denoting E_{SBD} and the broken electrical field at higher stage is quantified as hard breakdown E_{HBD} .

The leakage current density and breakdown field for various instigated durations occurred at E_{HBD} is considered for quantifying electrical features of J-E curve. The Sm₂O₃/ IL/Ge stack which induces lowest J_g and highest E_{HBD} is regarded as the finest J-E featured MOS capacitor. From the J-E curve the E_{HBD} is observed at 7.81 MV cm⁻¹, 5.67 MV cm⁻¹, 13.31 MV cm⁻¹ and 9.98 MV cm⁻¹ for 5, 10, 15 and 20 min, respectively. Meanwhile, the leakage current density value at E_{HBD} was recorded at 10^{-3} , 10^{-4} , 10^{-5} order of magnitude for 5, 10 and 20 min, respectively whereas J_q value dropped to 10^{-6} order magnitude when the oxidation/nitridation duration reached 15 min. Comparing all the samples, 15 min oxidation/nitridation durations induced the lowest leakage current density, $J_g \sim 8.38 \times 10^{-6}$ A cm⁻² and largest hard breakdown field $E_{HBD} \sim 13.31$ MV cm⁻¹. Besides, this duration also exhibited the largest soft breakdown $E_{SBD} \sim 4.67 \text{ MV}$ cm⁻¹ at gate leakage current density, 5.91×10^{-6} A cm⁻². The increase of leakage current for 5 and 10 min is due to volatilization of GeO (g). Since 10 min condition accelerated the GeO (g) desorption as confirmed in XPS analysis and so causing lowest J_g due to deterioration of insulation properties at Sm₂O₃ interface. Suppression of GeO (g) for 15 and 20 min gate stack attributed to improve the leakage current densities. From several previous reports (Wong & Cheong, 2011b, 2012c) it was found that increase of IL layer causes lower electrical breakdown field. The HRTEM images (Fig. 4.27) evidenced that, the IL layer is reduced at 15 min in counter to 20 min hence a larger breakdown field and one order lower leakage current has been generated at 15 min. Besides, the biggest CBO, ΔE_c obtained for 15 min would limit the movement of electrons from substrate Ge to the gate electrode through Sm₂O₃/IL which also instigated to achieve lowest J_g magnitude. Conversely, the lowest CBO, ΔE_c observed for 10 min assisted the electrons to cross over the barrier height between (Sm₂O₃ + IL) and substrate Ge causing largest leakage current density while initiating faster electrical breakdown.

In addition, the reduction of leakage current density at 15 min is also complementary with the XRD findings where more amorphous Sm₂O₃ was formed during the 15 min oxidation/nitridation which diminished the current conduction path due to grain consolidation. Furthermore, the fewest Q_{eff} , Q_{it} , D_{it} also aided in generating largest breakdown field for 15 min oxidation/nitridation condition. The lowest $J_g \sim 8.38 \times 10^{-6}$ A cm⁻² and largest $E_{HBD} \sim 13.31$ MV cm⁻¹ exhibited for Sm₂O₃/Ge gate stack is higher than the ZrO₂/Ge (Lei et al., 2018), La₂O₃/Ge (Zhao et al., 2018), Sm₂O₃/Si, (Goh et al., 2016b) and Nd₂O₃/Si (Hetherin et al., 2017a). Therefore, it is suggested that 15 min oxidation/nitridation condition facilitates best electrical properties in *J-E* plot.



Figure 4.34: *J-E* features for $Sm_2O_3/IL/Ge$ based MOS capacitors at different oxidation/nitridation durations (5 - 20) min.

4.4 Influence of Various Thermal Oxidation/Nitridation Durations on Sputtered Ho/Ge in N₂O Ambient.

4.4.1 XRD Analysis

A systematic study of XRD patterns were conducted to get detail insight on the crystal orientation and structure of the fabricated gate stack. Fig. 4.35 displays the close up XRD patterns at 2 θ range within (10° – 60°) for various oxidization/nitridation durations 5 min, 10 min, 15 min and 20 min. Two strong peaks and a weaker peak belong to cubic Ho₂O₃ denoted as c-Ho₂O₃ were detected at 29.16°, 59.08° and 48.20° corresponding to plane orientations (222), (631) and (440), respectively for all the investigated samples. Besides, two more weaker peaks of c-Ho₂O₃ also appeared at 35.56° with plane orientation (411) was detected for 10 and 15 min samples while the peak at 47.35° corresponding to c-Ho₂O₃ with plane orientation (512) appeared for the 5, 10 and 20 min samples. Meanwhile, four additional relatively blunt peaks of c-Ho₂O₃ were detected at 41.97°, 54.15°, 56.34° and 57.17° corresponding to plane orientations of (431), (602), (541) and (622), respectively for 10 min samples. The c-Ho₂O₃ peaks phases matched with the Inorganic Crystal Structure Database (ICSD) reference code no. 33655 and 152458. The diffraction peaks of GeO₂ appeared in tetragonal phase denoted as t-GeO₂ found at 27.16° and 39.11° with plane orientations (110) and (201), respectively for 5 and 10 min. Meanwhile, GeO₂ was not detected for 15 and 20 min samples which indicates the amorphous structure of GeO₂ for these durations. The t-GeO₂ peaks were confirmed by the ICSD reference code no. 9015579. Moreover, a sharp diffraction peak of Ge₃N₄ revealed at 31.73° corresponding to plane (211) for 10 and 15 min while the peak became weaker for 5 min sample. Due to amorphous structure of Ge₃N₄ no diffraction pattern of Ge₃N₄ was detected for 20 min sample. The diffraction peak corresponded well with the ICSD reference code no. 152551 which signifies the cubic phase of Ge₃N₄. The findings from XRD spectra validates formation of Ho₂O₃ dielectric layer and existence of Ge₃N₄

and GeO₂ suggests the creation of interfacial layer (IL) between the dielectric oxide film layer and Ge substrate.

Furthermore, according to the XRD pattern it is noticed that peak intensity of c-Ho₂O₃ at 29.16° (222), 59.08° (631) and 48.20° (440) increases when the oxidation/nitridation duration extended from 5 min to 10 and 15 min but gradually decreasing from 10 to 20 min as shown in Fig. 4.36. It is apparent that 10 min exhibits the highest Ho₂O₃ peak intensity followed by 15 and 5 min samples and 20 min sample exhibits the lowest intensity. Since increment of intensity indicates the degree of crystallinity and thus, it can be suggested that 10 and 15 min oxidation/nitridation durations enhances crystallinity of Ho₂O₃ dielectric interface while oxidation/nitridation of both shorter time (5 min) and longer time (20 min) decreases crystallinity of Ho₂O₃ film. Comparing all the investigated sample it can be inferred that the dielectric oxide film is more crystalline for 10 min sample and least crystalline for 20 min sample. The decrease of Ho₂O₃ intensity for 5 and 20 min samples may be induced from large amount of defects caused by the excessive O atom diffusion from the dielectric interface. Thus, the peaks became broader with lower intensities. The release of O atom from Ho₂O₃ film for 5 and 20 min samples were also apparent from XPS analysis as discussed in section 4.5.3.



Figure 4.35: XRD spectra of the Ho/Ge samples oxidized/nitrided at different durations (5 min, 10 min, 15 min, and 20 min).



Figure 4.36: Cubic, c-Ho₂O₃ intensities at various oxidation/nitridation durations (5 min, 10 min, 15 min, and 20 min).

The crystallite sizes, *D* and micro-strains, ε of the detected elements has been estimated from the peak broadening of the XRD pattern employing Williamson–Hall plot. According to Eq. 3.5, a graph of $\beta_{\tau} \cos \theta$ as a function of $4\varepsilon \sin \theta$ for Ho₂O₃ has been plotted as shown in Fig. 4.37. The well fitted points with regression co-efficient $r^2 = 1$ were considered for linear fitting from the range of distributed values. As defined in Eq. 3.5, the intercept of the graph is equal to $\frac{\kappa\lambda}{D}$ which provides the estimation of crystallite size *D* whereas the gradient of linear regression gives the micro-strain, ε value. Crystallite sizes of GeO₂ and Ge₃N₄ were not calculated due to insufficient number of peaks which were insignificant for the approximation.



Figure 4.37: Williamson–Hall (W–H) plot of c-Ho₂O₃ for investigated oxidized/nitrided samples at various durations (5 – 20) min.

The crystallite sizes and micro-strain of Ho₂O₃ calculated from W-H plot is displayed in Fig 4.38. The crystallite sizes of Ho₂O₃ increased from 25.33 nm to 41.25 nm when the oxidation/nitridation duration extended from 5 to 10 min but decreased from 41.61 nm to 29.53 nm after extending the duration from 15 to 20 min. However, it is noticeable that crystallite sizes increased slightly from 41.25 nm to 41.61 nm while incrementing the duration from 10 to 15 min. The small difference of crystallite size between 10 and 15 min samples might be due to well distributed and more homogenous crystallite size. Meanwhile, positive micro-strain value was found for 5 min sample while the other investigated samples yielded negative micro-strains. The highest micro-strain value of 0.0015 was recorded for 5 min sample and the second highest value of -0.0003 was recorded for 20 min oxidation/nitridation duration. However, the micro-strain of 10 and 15 min samples are almost same with a value of -0.0004.



Figure 4.38: Crystallite sizes (triangle symbol, left axis) and micro-strains (circle symbol, right axis) of Ho₂O₃ at various oxidation/nitridation durations (5 min, 10 min, 15 min, and 20 min).

4.4.2 J-E Characteristics

Fig. 4.39 demonstrates the leakage current density, J_g with the function of electrical field, E for various oxidation and nitridation durations 5 min, 10 min, 15 min and 20 min. The current-voltage (I-V) measurement was transformed to (J-E) plot by applying Eq. 3.6 According to the J-E plot the sample which corresponds to lowest leakage current density, J_g and retains highest electrical breakdown field, E_{BD} was regarded as the best electrical behaviour over other samples. Two step breakdowns were recorded for all the investigated samples. Formation of top Ho₂O₃ layer and an interfacial layer between dielectric and Ge substrate is the main reason for the occurrence of two-step breakdowns (Wong & Cheong, 2011c). The initial stage breakdown that occurs in the lower electric field has been labelled as soft breakdown, E_{SBD} where the instant increase of leakage current density remains relatively low. On the other hand, the sharp the breakdown observed on the higher electric field is considered as hard breakdown, E_{HBD} where the increase of instantaneous leakage current density is larger. The electrical field breakdown and leakage current density occurred on the hard breakdown, E_{HBD} position was considered for evaluating best electrical properties among the investigated capacitors. The recorded hard electrical field breakdown, E_{HBD} for 5 min, 10 min, 15 min and 20 min samples were 4.47 MV cm⁻¹, 8.59 MV cm⁻¹, 7.25 MV cm⁻¹ and 2.97 MV cm⁻¹, respectively. It is observed that leakage current density, J_g at E_{HBD} for 10 and 15 min were in 10^{-5} , 10^{-4} order of magnitude respectively while the leakage current density, J_g increased to 10^{-3} order of magnitude for 5 and 20 min samples. Among all the investigated capacitors 10 min sample exhibits highest electrical breakdown field, E_{HBD} of 8.59 MV cm⁻¹ with the lowest leakage current density, J_g of 5.34 \times 10⁻⁵ A cm⁻² at E_{HBD} . In addition, 10 min sample also induces highest soft breakdown field, E_{SBD} of 4.58 MV cm⁻¹ at leakage current density, J_g of 1.77 $\times 10^{-7}$ A cm⁻².

Therefore, 10 min oxidized/nitride sample yields the best *J-E* performance for Ho₂O₃ dielectric based MOS devices. It is noticeable that 20 min sample initially kicked off with a leakage behaviour and leakage current density was unusually high before the hard breakdown which might be due to damaged Ho₂O₃ dielectric layer caused by the extreme desorption of GeO from the Ge substrate during the extended oxidation/nitridation process (Lei et al., 2017). The improvement of dielectric breakdown and leakage current density for 10 min and 15 min samples are possibly related to the alleviation of structural defect comprising oxygen vacancies and Ge dangling bonds in Ho₂O₃/IL interface (Hosoi et al., 2009). Moreover, 10 and 15 min samples of Ho₂O₃/Ge revealed comparatively lower leakage current density, $J_g \sim 10^{-5} - 10^{-4}$ A cm⁻² and higher electrical breakdown field, $E_{BD} \sim 7.25 - 8.59$ MV cm⁻¹ than the reported ZrO₂/Ge system thermally oxidized/nitrided in N₂O ambient [where, $J_g \sim 10^{-3}$ A cm⁻² and $E_{BD} \sim 6.9$ MV cm⁻¹] (Lei et al., 2018).



Figure 4.39: *J-E* curve of various oxidized/nitrided sample at durations (5 min, 10 min, 15 min, and 20 min).

4.4.3 XPS Analysis

XPS analysis was conducted to get insight on the structural and compositional properties of Ho₂O₃ dielectric gate stack. According to the survey scan core level spectra of Ho 4d, Ge 3d, O1s, N1s were detected. Narrow scan of Ho 4d, Ge 3d, O1s were performed for samples oxidized/nitrided at different durations 5 min, 10 min, 15 min and 20 min. Fig. 4.40 depicts the deconvoluted peaks of Ho 4d narrow scan for various oxidation/nitridation durations 5 min, 10 min, 15 min and 20 min. The reference peak position of Ho₂O₃ in the Ho 4d spectra was reported at 161.8 eV (Pan, Chang, et al., 2010; Pan, Yen, et al., 2010). It is clearly visible that the centroid peak of Ho 4d shifts negatively towards the lower binding energy by 1.07 eV, 0.66 eV, 0.87 eV, and 1.17 eV, respectively for 5, 10, 15, and 20 min with respect to reference position of Ho₂O₃. The negative shift towards the lower binding energy indicates the formation of sub-stoichiometric Ho₂O₃ for all the investigated samples (Wong & Cheong, 2011c). This finding suggests the release of O atom from the Ho₂O₃ dielectric itself towards the Ge substrate. Since 10 min sample yields the lowest negative shift of binding energy in other way it can be said that shifting toward higher binding energy in compared to other samples which indicates comparatively less O diffusion than the other investigated samples. Hence, 10 min sample promotes formation of more stoichiometric Ho₂O₃ and exhibits higher growth of Ho₂O₃ than those of 5, 15 and 20 min samples. Conversely, 20 min sample yields the highest negative shift towards the lower binding energy site followed by 5 and 15 min samples. In this case it can be assumed that the growth of Ho₂O₃ is hampered due to excessive O diffusion from the Ho₂O₃ film which induces formation of less stoichiometric Ho₂O₃ dielectric layer while leaving large amount of oxygen vacancies. Therefore, according to the binding energy shift of Ho 4d spectra it is predicted that Ho₂O₃ growth will be highest Ho₂O₃ for 10 min sample followed by 15 min and 5 min while the lowest at 20 min sample. The validation of this assumption will be explained in the O1s spectra.

142

Based on Ho 4d spectra observation, it can be suggested that better growth of Ho₂O₃ is primarily related to the degree of sub-oxide (Ho–O–Ho) transformation and the amount of O atom diffusing from Ho₂O₃ dielectric itself. Furthermore, the negative shift in the direction of lower binding energy also indicates Ho₂O₃ gate stack does not create any Germanate (Ho–O–Ge) compound in the interfacial layer.



Figure 4.40: Ho 4d narrow scan spectra measured for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations (5 – 10) min.

Fig. 4.41 elucidates the Ge 3d spectra of different oxidation/nitridation durations (5-20) min. Ge 3d spectra comprises two broad peaks which incorporates Ge⁰ and mixture of various chemical bonding states of distinctive Ge sub-oxides. It has been reported that, Ge 3d peak positions typically appears in the range of 28.6 eV to 29.6 eV (Mitrovic et al., 2014) The chemical composition of Ge suboxides (GeO_x) and native oxide (GeO₂) were deconvoluted into Ge¹⁺, Ge²⁺, Ge³⁺, Ge⁴⁺ bonding states with binding energy shift of +0.8 eV, +1.8 eV, +2.6 eV and > +3 eV, respectively from the relative reference position of elemental Ge⁰ at 29.0 eV. An additional peak denoted as Ge* was well fitted in between Ge^{2+} and Ge^{3+} with core level energy shift of +2.3 eV. The Ge^{4+} chemical bonding state denotes GeO₂ native oxide of Ge, since it has been projected to occur in the range of +3 to +3.4 eV (Dimoulas et al., 2010; Mitrovic et al., 2014; Sasada et al., 2009). The Ge* is represented as Ge₃N₄, which predictably appears in smaller range than that of Ge^{4+} (GeO₂) +3 eV in between +1.9 to +2.3 eV energy shift (Maeda et al., 2004, 2006; Minoura et al., 2013) The chemical energy shift of Ge sub-oxides lies at +0.8 eV, +1.8 eV and +2.6 eV can be denoted as Ge₂O, GeO and Ge₂O₃ according to the reported literature (J. Song et al., 2007). It is observed that, the content of Ge^{4+} native oxide GeO_2 and Ge* (Ge₃N₄) existed for all the investigated sample. It is notable that, there is a strong presence of volatile species of Ge²⁺ (GeO) for 5, 15, and 20 min samples except 10 min sample. Meanwhile, Ge³⁺ (Ge₂O₃) was found for 5 and 20 min while no Ge³⁺ (Ge₂O₃) were formed both 10 and 15 min samples. Moreover, the content of Ge¹⁺ (Ge₂O) was detected for all the investigated samples except 15 min sample. This observation suggests the formation of amorphous interfacial layer which is evident from the presence of intermixing chemical bonding state of GeO₂ Ge₂O, GeO, Ge₂O₃ and Ge₃N₄.

The influence of various oxidation/nitridation durations upon thermodynamic stability of interfacial layer was investigated from the comparison of integrated area of the deconvoluted peaks as displayed in Fig. 4.41. Ge 3d spectra reveals that the sum of lower binding energy unstable phases of sub-stoichiometric GeO_x (Ge¹⁺ and Ge³⁺) and $Ge^{2+}(GeO)$ are found more for 20 min sample followed by 5 min sample while these unstable phases reduced significantly at oxidation/nitridation duration of 10 min followed by 15 min. The content of $Ge^{4+}(GeO_2)$ increases subsequently for 20 and 5 min samples whereas there is significant reduction of Ge⁴⁺(GeO₂) for 10 and 15 min samples. Meanwhile, it is noticeable that, G*(Ge₃N₄) peaks substantially increases for 10 min sample followed by 15 min sample while Ge₃N₄ decreases subsequently for 5 and 20 min sample. It is noteworthy that, Ge⁴⁺(GeO₂) are dominant compound for 5 and 20 min samples while Ge₃N₄ compound shows more pronounce appearance for 10 and 15 min samples. Hence, it is predicted that GeO₂ and Ge₃N₄ becomes more stoichiometric for 10 and 15 min samples due to the reduction of sub-stoichiometric unstable phases of GeO_x and GeO. It is believed that the content of Ge²⁺ (GeO) and Ge⁴⁺(GeO₂) are more thermally unstable than those of Ge¹⁺ (Ge₂O) and Ge³⁺ (Ge₂O₃) (Juan. Gao et al., 2017; D. Wang, He, Fang, et al., 2019). Consequently, presence of large amount Ge²⁺ (GeO) and Ge⁴⁺(GeO₂) will cause sharp degradation on thermodynamic stability of gate stack. In addition, it is well documented that Ge₃N₄ is more thermodynamically stable with high-k permittivity in counter to GeO₂ (Kutsuki et al., 2009; Q. L. Li et al., 2011; Maeda et al., 2004). Accordingly, the interfacial layer rich with Ge-N bonding rather than Ge-O compounds will exhibit more thermally stable interfacial layer. Based on this assumption and above discussion it can be suggested that more thermally stable interfacial layer was formed for 10 and 15 min oxidation/nitridation since these samples consist of more stoichiometric Ge₃N₄ compound and fewer sub-stoichiometric GeO₂, GeO_x and GeO than those of 5 and 20 min samples. In addition, 10 min sample yielded the highest growth of Ge₃N₄ and

lowest growth of $Ge^{4+}(GeO_2)$ and GeO_x (Ge^{1+}) content without forming any $Ge^{2+}(GeO)$ species indicating most thermally stable interfacial layer among all the investigated samples. Conversely, 20 min sample exhibits the most thermally unstable interfacial layer since this sample reveals the highest density of Ge⁴⁺(GeO₂), Ge²⁺(GeO) and lowest growth of Ge₃N₄. The increase of sub-stoichiometric GeO₂ and volatile GeO species will have severe detrimental consequences on the electrical performance of the gate stack. The aspect of GeO₂ and GeO_x densification and Ge₃N₄ reduction has been explained in later section 3.6. Considering, 10 and 15 min samples fewer GeO₂, GeO_x formation and reduction of volatile GeO species are related to the pronounce growth of Ge₃N₄ compound. It is assumed that substantial incorporation of N⁺ atom increases Ge₃N₄ density at this oxidation/nitridation durations which obstructed the O vacancy diffusion paths and also retarded the disproportionate transportation of O atom into the Ge substrate. Consequently, hindered the growth of unstable GeO_x compounds and reduced volatile GeO desorption while maintaining the integrity of GeO₂ (Copetti & Soares, 2016; Etcheverry et al., 2019). Therefore, based on the Ge 3d spectra observation it can be suggested that 10 min sample exhibits the superior configuration of stoichiometric contents with minority composition of GeO_x traps that might help in improving the device performance of the high-k interface.



Figure 4.41: Deconvolution of O1s narrow band spectra for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations (5 – 10) min.

Fig. 4.42 shows the O1s spectra for investigated samples at various oxidation/nitridation durations (5 – 20) min. The O1s spectra were deconvoluted into two peaks of Ho₂O₃ and Ge–O content. The Ho₂O₃ was well fitted in the lower binding energy site at peak position 529.80 eV and Ge–O content fitted well in the higher binding energy site at peak position 531.10 eV (J. Gao et al., 2017; Pan, Chang, et al., 2010). According to the O1s spectra 10 min sample exhibited the highest integrated peak area of Ho₂O₃ followed by 15 and 5 min samples while 20 min oxidation/nitridation duration induced lowest Ho₂O₃ peak area. This finding indicates that 10 min oxidation/nitridation is more favourable for higher Ho₂O₃ growth while longer time ~ 20 min oxidation/nitridation is less convenient for the growth of Ho₂O₃. This observation supports the Ho₂O₃ growth prediction assumed in the Ho 4d spectra. In counter to Ho₂O₃ growth a reverse effect is observed for the growth of Ge–O species. It is observed that, Ge–O species growth is highest for 20 min sample followed by 5 min and 15 min samples while decreases substantially for 10 min sample with the lowest Ge–O growth. Since Ho₂O₃ peak reduced for 5 and 20 min samples hence, it is evident that the increased growth of Ge–O component at these oxidation/nitridation durations is due to large amount of O atoms diffusion from Ho₂O₃ dielectric itself toward the Ge substrate. The additional in-diffused O atoms contributed to form more unstable phases of GeO₂ or GeO_x interfacial layer compound between Ho₂O₃ dielectric film and substrate Ge. The Ge–O growth observation in O1s spectra is consistent with the Ge 3d spectra.



Figure 4.42: Deconvolution of O1s narrow band spectra for oxidized/nitrided sputtered Ho/Ge samples at different oxidation/nitridation durations (5 – 10) min.

The degree of oxidation/nitridation level for Ho_2O_3 growth were analyzed by plotting the Ho 4d binding energy shift with respect to oxidation/nitridation durations as depicted in Fig. 4.43. It is notable that Ho 4d binding energy are below the full oxidation level for all the investigated samples which indicates all samples are in under-oxidized condition. However, the 5 min and 20 min samples are located substantially in far distance below the full oxidation level which reveals more profound under-oxidized condition. On the other hand, the 10 min sample is closer to the full oxidation level followed by the 15 min sample which indicates increase of oxidation level for these samples. In conjunction with the *J-E* properties as described in Fig. 4.39, it can be inferred that the deep underoxidation condition of Ho_2O_3 has an adverse impact on the electrical field breakdown. The 10 min oxidized/nitrided duration condition generates moderate and controlled oxidation/nitridation rate leading to highest electrical field breakdown.



Figure 4.43: Peak position of Ho 4d narrow scan spectra as function of various oxidation/nitridation durations estimating the degree of oxidation level.

4.4.4 Band Alignment

The band alignment of Ho₂O₃/IL/Ge system has been developed by applying the same technique as described in section 4.3.2. Accordingly, XPS valance band (VB) spectra were used to estimate the valance band (VB) edges, E_v and valance band offsets (VBO) while energy bandgaps, E_g were extracted from O1s plasmon loss XPS spectra as shown in Fig. 4.44. The conventional tailed region observed at low binding energy site of VB spectra is being considered as Ge substrate signal. The reference position of VB edge, E_v (Ge) was considered at 0.10 eV and was assumed to be constant for various oxidation/nitridation duration (Cao et al., 2017). The VB edge, E_v of Ho₂O₃/IL was approximated by the linear extrapolation intercept of maximum negative slope leading edge in

VB spectra with the minimum edge of Ge substrate baseline as demonstrated in Fig. 4.44a. The measured VB edges, E_v of Ho₂O₃/IL with respect to substrate Ge were approximated to be 4.39, 4.77, 4.54 and 4.26 eV for 5 min, 10 min, 15 min and 20 min samples, respectively. The differences between VB edge, E_v of Ho₂O₃/IL and VB edge, E_v of Ge ascribed in Eq. 4.21 provides the VBO, ΔE_v of Ho₂O₃/IL.

$$\Delta E_{v} (\text{Ho}_{2}\text{O}_{3}/\text{IL}) = E_{v} (\text{Ho}_{2}\text{O}_{3}/\text{IL}) - E_{v} (\text{Ge}) \qquad (\text{Equation 4.21})$$

The VBO, ΔE_{v} (Ho₂O₃/IL) has been determined to be 4.29 eV, 4.67 eV, 4.44 eV and 4.16 eV for 5 min, 10 min, 15 min and 20 min samples, respectively. Finally, the conduction band offset (CBO), ΔE_{c} of Ho₂O₃/IL/Ge has been determined from Eq. 4.22.

$$\Delta E_c (\text{Ho}_2\text{O}_3/\text{IL}) = E_g (\text{Ho}_2\text{O}_3/\text{IL}) - E_g (\text{Ge}) - \Delta E_v (\text{Ho}_2\text{O}_3/\text{IL})$$
(Equation 4.22)

Where, E_g (Ho₂O₃/IL) and E_g (Ge) denotes energy bandgaps of Ho₂O₃/IL and substrate Ge, respectively. The energy bandgaps, E_g (Ho₂O₃/IL) was deduced by the intercept of linear extrapolation from respective O1s plasmon loss of XPS spectrums, for each of the oxidized/nitrided investigated samples as illustrated in Fig. 4.44b. The E_g (Ho₂O₃/IL) has been estimated to be 6.83 eV, 7.81 eV, 7.37 eV and 6.43 eV for oxidation/nitridation durations 5, 10, 15 and 20 min, respectively. The E_g (Ge) ~ 0.67 eV has been considered (Kamata, 2008). The band alignment schematic of Ho₂O₃/IL/Ge structure for different oxidation/nitridation durations is illustrated in Fig. 4.45. The CBO, ΔE_c (Ho₂O₃/IL) interface for various oxidation/nitridation durations were estimated 1.87 2.47, 2.26 and 1.60 eV for 5, 10, 15 and 20 min, respectively.

According to the band alignment schematic it is notable that all the investigated samples exhibited VBO and CBO above 1 eV which fulfilled the fundamental requirement of being an eligible candidate for high-*k* dielectric in MOS devices.

It has been reported that large conduction band offset facilitates in decreasing leakage current density and increasing electrical breakdown field (J. Gao et al., 2017; Wong & Cheong, 2011a). Accordingly, it is anticipated that the largest CBO, ΔE_c (Ho₂O₃/IL) obtained for 10 min sample restrained inclusion of electrons between Ho₂O₃/IL and Ge substrate which compensated the leakage current density and as well as enhanced the electrical breakdown field at this oxidation/nitridation duration. Contrariwise, for 20 min sample the decrease in ΔE_c (Ho₂O₃/IL) exaggerated the electron tunnelling that could exceed the barrier height between Ho₂O₃/IL and Ge interface and thereby causing highest leakage current density with lowest electrical breakdown field. Ultimately, continual decrease of CBO and VBO increases the gate leakage and lowers the breakdown field eventually deteriorating the device capability which is consistent with the *J-E* performance as shown earlier Fig. 4.39.

Furthermore, it is also observed that the energy bandgap, E_g (Ho₂O₃/IL) drops by 0.44 –1.38 eV for 5, 15 and 20 min with respect to the 10 min sample energy bandgap, $E_g \sim 7.81$ eV. The reduction of E_g is caused by the creation of defect gap states owing to GeO desorption and other GeO_x defects (Kita et al., 2009). Accordingly, it is predicted that higher density defect gap states have been generated into the Ho₂O₃/IL energy bandgap region through oxygen vacancies and Ge²⁺ defects diffusion induced from volatile GeO desorption and GeO_x interlayer causing energy bandgap reduction for 5, 15 and 20 min samples. Since 20 min sample induced highest GeO desorption and GeO_x defects consequently the smallest bandgap has been achieved. However, the energy bandgaps, $E_g \sim 6.43 - 7.81$ eV evaluated for Ho₂O₃/IL/Ge system is comparatively higher than typical Ho₂O₃ bandgap, $E_g \sim 5.3$ eV which can be contributed to the inclusion of interfacial layer, since the bandgap was calculated considering both (Ho₂O₃ + IL) layer.



Figure 4.44: Valance band offset (ΔE_v) and energy bandgap (E_g) Ho₂O₃/IL/Ge structure at various oxidation/nitridation durations (5 – 20) min (a) Extraction of valance band edge, E_v and VBO estimation from valance band spectra (b) Energy bandgap extraction from O1s plasmon loss XPS spectrum.



Figure 4.45: Band alignment schematic Ho₂O₃/IL/Ge stack for various oxidation and nitridation durations (5 – 20) min.

4.4.5 HRTEM Analysis

In order to justify the suitability of Ho₂O₃ as high-*k* gate oxide material with the MOS scaling limit it is imperative to understand the effect of oxidation/nitridation process on the physical thickness of the functioning Ho₂O₃/Ge gate stack. Fig. 4.46 displays the cross-sectional images acquired from HRTEM analysis for the oxidation/nitridation durations 10 min and 20 min. It has been spotted that both the samples comprise two distinct layers. The top layer at the surface is considered to be dielectric Ho₂O₃ layer since it is located at far distance position from substrate Ge baseline. Meanwhile the layer sandwiched between Ho₂O₃, and the Ge substrate is considered as interfacial layer (IL) as it is relatively closer to Ge substrate. Based on the XPS and XRD analysis it is considered that the interfacial layer is combination of various Ge–O (i.e. GeO₂ and GeO_x) compounds and Ge₃N₄. It is apparent from Fig. 4.46 that Ho₂O₃ dielectric thickness layer increases for 10 min sample while interfacial increases for 20 min sample. The thickness of Ho₂O₃ layer measured from HRTEM images are 4.33 nm and 3.60 nm for 10 min and 20 min samples, respectively. Meanwhile the interfacial layer is measured to be 4.39 nm

and 6.18 nm, for 10 min and 20 min samples, respectively. Accordingly, it can be suggested that the oxidation/nitridation rate of forming dielectric oxide is higher for 10 min sample while oxidation/nitridation rate of forming interfacial layer is higher for the extended duration of 20 min sample. This finding also indicates that interfacial layer becomes dominant during longer oxidation/nitridation. The total thickness of 5 min and 15 min samples were approximated using linear regression plot as displayed in Fig. 4.47, assuming the total thickness of the gate stack increases linearly with the increment of oxidation/nitridation duration. The total thickness approximated from linear regression plot is 8.22 nm and 9.18 nm for 5 and 15 min, respectively while the total thickness measured from the HRTEM images Figure 12 was 8.72 nm and 9.78 nm for 10 and 15 min, respectively.

Furthermore, the crystallinity of the Ho₂O₃/IL film was also verified by analyzing diffraction pattern of HRTEM images. Fast Fourier Transform (FFT) measurement was conducted over HRTEM images using ImageJ software in order to obtain the diffraction pattern displayed as inset image in Fig. 4.46. Presence of cloudy spots pattern in Ho₂O₃ layer as well as in interfacial layer confirms that both of these layers are in amorphous phase for both 10 min and 20 min samples. Moreover, both the sample reveals polycrystalline structure of Ge substrate which has been validated by the existence of multiple spotted FFT image diffraction pattern.



Figure 4.46: Structural cross-sectional morphology obtained from HRTEM for oxidized/nitrided sputtered Ho/Ge samples at 10 min and 20 min durations. Inset images shows FFT diffraction.



Figure 4.47: Determination of Ho₂O₃/IL/Ge stack physical thickness for investigated samples at various oxidation/nitridation durations (5 – 10) min.

4.4.6 Oxidation/Nitridation Mechanism Model

Oxidation and nitridation mechanism process for Ho₂O₃/IL/Ge stack has been established based on the correlation of HRTEM and XPS analysis considering oxidized/nitrided samples at duration of 10 and 20 min as illustrated in Fig. 4.48. The schematic of Ho sputtered on Ge substrate before thermal oxidation/nitridation is shown in Fig. 4.48a. Initially, during the beginning of oxidation/nitridation process N₂O diffused from the atmosphere and reacted with the top surface sputtered Ho of the Ge substrate following Eq. 4.23

$$2\text{Ho} + 3\text{N}_2\text{O}(g) \rightarrow \text{Ho}_2\text{O}_3 + 3\text{N}_2(g)$$
 (Equation 4.23)

Afterwards, N₂O received adequate energy to diffuse throughout the Ho₂O₃ film layer while forming interfacial layer comprising GeO₂ and Ge₃N₄ compounds for both the samples, by the reaction of substrate Ge and in-diffused N₂O gas according to Eq. 4.24. Corresponding to XPS analysis substantial reduction of GeO₂ and an additional growth of Ge₃N₄ was observed for 10 min. Hence, at these oxidation/nitridation durations Ge dangling bonds from GeO₂ network reacted with the atmospheric in-diffused N₂O generating Ge₃N₄ rich interfacial layer according to Eq. 4.25. Meanwhile, the residual product of O atom in Eq. 4.25 out-diffuses toward the Ho₂O₃ interface which filled up the oxygen vacancy of Ho₂O₃ contributing to the thicker dielectric layer for 10 min oxidation/nitridation duration. On the other hand, GeO₂ rich interfacial layer was revealed for 20 min sample. Due to longer oxidation time large extent of O atom in-diffused from the Ho₂O₃ film which reacted with the out-diffused Ge atom from unstable phases of Ge₃N₄ and GeO₂ network producing large amount of GeO₂ compound in the interfacial layer according to Eq. 4.26.

$$4\text{Ge} + 2\text{N}_2\text{O}(\text{g}) \rightarrow \text{GeO}_2 + \text{Ge}_3\text{N}_4$$
 (Equation 4.24)

$$6\text{GeO}_2 + 4\text{N}_2\text{O}(g) \rightarrow 2\text{Ge}_3\text{N}_4 + 8\text{O}_2(g)$$
 (Equation 4.25)

$$3Ge + 3O_2(g) \rightarrow 3GeO_2$$
 (Equation 4.26)

Considering, the 20 min sample once GeO₂ is formed by the Eq. 4.24 and Eq. 4.26 afterwards GeO₂ reacting with the substrate Ge causing GeO (g) desorption due to oxygen vacancy (V_o) diffusion through GeO₂ interface by redox reaction of Eq. 4.27 (Kita et al., 2008). Moreover, this detrimental reaction is also reinforced by the enhanced O atom in-diffusion from Ho₂O₃ film interface consequentially generated severely defective Ho₂O₃ film interface as shown in Figure 4.48c.

$$GeO_2 + Ge \rightarrow 2GeO(g)$$
 (Equation 4.27)

Meanwhile, GeO (g) desorption disappears for 10 min sample. It has been suggested that when any material is deposited above GeO_2 that retards GeO (g) desorption tends to annihilate oxygen vacancy diffusion (V_o) following the disproportionation reaction of Eq. 4.28 (S. K. Wang, Liu, & Toriumi, 2012).

$$2\text{GeO} \rightarrow \text{GeO}_2 + \text{Ge}$$
 (Equation 4.28)

Hence, at this oxidation/nitridation duration generation of Ge₃N₄ rich interfacial compound suppressed the desorption of volatile GeO by blocking oxygen vacancy diffusivity through the Ho₂O₃/IL interface initiating reverse GeO reaction ascribed in Eq. 4.28 Accordingly, less defective Ho₂O₃ interface was obtained for 10 min sample as shown in Fig. 4.48b.

Furthermore, existence of Ge₂O, Ge₂O₃ and reduction of Ge₃N₄ was evident for 20 min sample in XPS analysis. It is assumed that Ge₃N₄ bonding deteriorated by producing additional Ge suboxides Ge₂O and Ge₂O₃ due to the reaction between in-diffusing O atom from Ho₂O₃ and out-diffusing Ge atom from Ge₃N₄ according to Equ.4.29 and Eq. 4.30. Unlike 20 min sample, small amount of Ge₂O detected for 10 min sample which was formed by the reaction between out-diffused Ge from GeO₂ network or Ge wafer and in-diffused O atom from Ho₂O₃ dielectric as shown in Eq. 4.31. However, these reactions induced additional defects in the interfacial layer for both the samples.

$$4Ge_3N_4 + 3O_2(g) \rightarrow 6Ge_2O + 8N_2(g)$$
 (Equation 4.29)

$$4Ge_3N_4 + 9O_2(g) \rightarrow 6Ge_2O_3 + 8N_2(g)$$
 (Equation 4.30)

$$4Ge + O_2(g) \rightarrow 2Ge_2O$$
 (Equation 4.31)
Based on this model it can be summarized that sufficient oxidation/nitridation of Ho_2O_3 at 10 min duration minimized the inter-diffusion of O and Ge atom causing thicker Ho_2O_3 layer with less GeO_x defect and more consolidated $Ho_2O_3/IL/Ge$ gate stack is formed. When the oxidation/nitridation is extended to 20 min a reverse phenomenon has been observed where interfacial layer becomes thicker since diffusion rate of O and Ge atom increases due to longer oxidation/nitridation time. Therefore, atoms become less consolidated causing more defective Ho_2O_3/IL interface.



Figure 4.48: Oxidation/nitridation mechanism model for sputtered Ho/Ge after subsequent oxidation/nitridation at different duration 10 and 20 min. Dotted circle represents oxygen vacancies, blue filled circle denotes in-diffusing oxygen and IL is interfacial layer.

4.4.7 C-V Characteristics

Fig. 4.49 elucidates the typical current – voltage (*C*–*V*) characteristics for different oxidation/nitridation durations 5min, 10 min, 15 min, 20 min where gate voltage of 6 V to +1 V has been swept directionally at high frequency of 1 MHz. It is observed that the capacitance at accumulation region increases as the oxidation/nitridation extended from 5 min to 10 min but decreases while the duration incremented from 10 min to 20 min. The highest accumulation capacitance is recorded for 10 min followed by 15 and 5 min samples while 20 min oxidation/nitridation duration exhibited the lowest capacitance. The variation of accumulation capacitance can be explained based on the formula of Eq. 4.10 (Goley & Hudait, 2014; Wong & Cheong, 2011b). The Eq. 4.10 shows that oxide capacitance is proportional to $\frac{k}{t_{ox}}$. Accordingly, the degree of capacitance increases with higher dielectric constant *k* value and reduction of total film thickness, $t_{ox} = \text{Ho}_2\text{O}_3$ + IL. Based on this relationship the highest capacitance for 10 min sample can be attributed to the higher dielectric constant, *k* (inset of Fig. 4.49) and reduction of total oxide thickness t_{ox} (Fig. 4.46).

The effective dielectric constant k_{eff} was evaluated using Eq. 4.10 where C_{ox} has been achieved from C-V curve while total physical oxide thickness (Ho₂O₃ + IL) was extracted using HRTEM images. Inset image inserted in Fig. 4.49 reveals the correlation between effective dielectric constant k_{eff} value and accumulation capacitance with respect to various oxidation/nitridation durations. It is noticeable that the changes of accumulation capacitance followed the same trend according to the of variation effective dielectric constant k_{eff} values for Ho₂O₃/ IL/Ge are 9.50, 13.60, 13.09 and 8.52 for 5, 10, 15 and 20 min samples, respectively. As reported by Pan et al., (2010) the *k* value of sputtered Ho₂O₃ deposited on Si substrate ranges between ~ 6.5-10.1. Paivasaari et al., (2005) reported the *k* value of ~ 9.9 for Ho₂O₃ with respect to SiO₂/Si

substrate. However, the 20 min sample min has a lower k_{eff} value which might come from the higher composition of low-k (Ge–O) interfacial compound and lower composition of high-k Ho₂O₃ compound. On the other hand, improvement of k_{eff} value for 10 min sample is due to better thermal stability of Ho₂O₃/IL interface caused by the densification of high-k Ho₂O₃ compound and reduction of low-k (Ge–O) interfacial compound. In contrast with observation of XPS analysis (Fig. 4.43) the k_{eff} value increases with higher Ho₂O₃ density while k_{eff} value reduces for higher Ge–O density. Therefore, the dielectric constant, k of Ho₂O₃/ IL/Ge structure is much dependent on the growth rate of high-k Ho₂O₃ film and low-k (Ge–O) interfacial content during the oxidation/nitridation process.



Figure 4.49: (a) *C–V* characteristics of Ho₂O₃/IL/Ge MOS capacitors measured at 1 MHz for bidirectionally swept gate bias for various oxidation/ nitridation durations (5 min, 10 min, 15 min, and 20 min). The inset displays effective dielectric constant and accumulation capacitance and effective dielectric constant as function oxidation and nitridation durations.

From the *C*-*V* plot, it is apparent that depletion region appeared in the negative bias region for all the investigated capacitors. The negative shift of flatband voltage, V_{FB} confirms the accumulation of positive oxide charges accumulated in the Ho₂O₃ interface during the thermal oxidation/nitridation process (Taoka et al., 2011). The effective positive charges Q_{eff} are calculated using the Eq. 4.20. The calculated effective positive charges for all the oxidation/nitridation durations are shown in Fig. 4.50. The source of positive oxide charges is possibly related to the large amount of defects existing in the Ho₂O₃ interface generated from dangling bonds and oxygen vacancies (Deng et al., 2011; Pan, Chang, et al., 2010). The 10 min sample exhibits the lowest positive oxide charges of $Q_{eff} \sim 8.80 \times 10^{12}$ cm⁻² and 20 min sample exhibit the highest positive oxide charges of $Q_{eff} \sim 1.41 \times 10^{13}$ cm⁻². The compensation of positive oxide charges for 10 min sample indicates that Ho₂O₃ has gained more homogenous structure at this oxidation/nitridation condition due to reduction of oxygen vacancies and other oxygen related defects.

Furthermore, hysteresis phenomena have been observed for all the investigated samples ranges between 343 mV – 882 mV as shown in Figure 4.51. It has been reported that presence of charged slow traps located in the border of high-*k* and interfacial layer interface or in the bulk of high-*k* dielectric causes hysteresis effect (Ke et al., 2018; Wong & Cheong, 2011b). Hence, evaluating the amount of slow trap density (*STD*), Q_{it} can explain the variation of hysteresis voltage since slow trap density is indirectly related with the hysteresis behaviour based on Eq. 4.11. According to the estimated *STD*, (Q_{it}) as displayed in Fig. 4.50 it is identified that 10 min sample yielded the lowest $Q_{it} \sim 2.93 \times 10^{12}$ cm⁻² while the 20 min sample exhibited the highest $Q_{it} \sim 4.24 \times 10^{12}$ cm⁻². Based on the findings of *STD* and hysteresis voltage displayed in Fig. 4.50 and Fig. 4.51 it can be remarked that hysteresis voltage reveals a proportional relationship with

the slow trap density. Accordingly, lowering of *STD* compensated hysteresis voltage for 10 min oxy-nitrided sample while generation of large amount slow trap density during 20 min oxidation/nitridation enhanced the hysteresis phenomena. The origin of slow traps is predicted due to out-diffusion of N⁺ atom and also tunnelling of carrier charges (electrons or holes) from Ge substrate (Ke et al., 2018; Misra et al., 2006) Hence, the increase of slow traps for 20 min sample is believed due to excessive N⁺ atom piling up towards the dielectric interface and large amount of carrier charges transportation from Ge substrate into the Ho₂O₃ film or Ho₂O₃/IL interface. Furthermore, the variation of *STD* and Q_{it} in response to various oxidation/nitridation duration follows the identical trend as Q_{eff} .



Figure 4.50: Amount of Effective oxide charge, Q_{eff} and slow trap density, Q_{it} existing in the Ho₂O₃/IL/Ge MOS capacitors for oxidized/nitrided samples at durations (5 – 20) min.



Figure 4.51: Changes in hysteresis voltage with the variation of oxidation and nitridation durations (5 – 20) min for Ho₂O₃/IL/Ge MOS capacitors.

Fig. 4.52 illustrates average interface trap density, D_{it} with respect to trap energy level ($E_C - E$) for various oxidation and nitridation durations (5 – 20) min. The Terman method has been employed for approximation of average interface trap density D_{it} for the oxidized/nitrided Ho₂O₃/IL/Ge MOS capacitors according to Eq. 4.12. According to the D_{it} vs ($E_C - E$) curves, oxidized/nitrided film of 10 min duration poses the lowest average interface trap density, $D_{it} \sim 10^{13} - 10^{14}$ eV⁻¹cm⁻² at ($E_C - E$) = 0.09 – 0.27 eV. For the estimation of total interface trap density D_{total} of the energy range ($E_C - E$), area under $D_{it} - (E_C - E)$ was determined which is portrayed as inset image in Fig. 4.52. The lowest D_{total} was found for 10 min sample approximated to be 1.97×10^{13} cm⁻² while extending the oxidation/nitridation to 20 min exposed the highest $D_{total} \sim 6.51 \times 10^{13}$ cm⁻². The variability trend of D_{total} with respect to oxidation/nitridation duration is similar with the variation of Q_{eff} and Q_{it} .

Considering the root of interfacial traps, it is predicted that structural defect comprising oxygen-defect (OD) and Ge dangling bonds that existed between dielectric film and Ge substrate are primarily associated with the charge capturing process for the MOS interface (Bethge et al., 2014). It is believed that decomposition of GeO₂ and desorption of GeO is the major source for oxygen vacancy incorporation and insertion of dangling or broken bonds (Kita et al., 2008; S. K. Wang et al., 2010). In contrast with XPS analysis the minimum interface trap density for 10 min sample should be attributed to the formation of Ge₃N₄ rich interfacial layer which effectively blocked the GeO desorption as well as acted as diffusion barrier against oxygen in-diffusion and Ge out-diffusion. As a result, oxygen vacancies and Ge dangling bond (DB) has been minimized that lead to decrease the interface trap density. Moreover, tendency of forming pure Ho₂O₃ compensated oxygen deficiency and passivated growth of undesirable GeO_x interlayer content, which also contributed to reduce interface traps at this oxidation/nitridation duration. On the contrary, higher interface trap density for 20 min sample indicates enhancement of unsaturated oxygen vacancies and surplus of Ge dangling bonds, which is caused by the bond weakening of interfacial Ge₃N₄ compound and excessive oxygen in-diffusion from Ho₂O₃ film. Consequently, strengthened the GeO (g) desorption and induced large amount of GeO_x defects leading to higher interface traps at the Ho₂O₃ and Ho₂O₃/IL interface. In addition, formation of sub-stoichiometric Ho₂O₃ also reinforced the charge trapping sites in the dielectric/IL interface.

Comparing with previous literature the lowest average interface trap density, $D_{it} \sim 9.39 \times 10^{13} \text{ eV}^{-1} \text{cm}^{-2}$ (at mid $E_C - E = 0.2 \text{ eV}$) of Ho₂O₃/IL/Ge stack for 10 min oxidized/nitrided sample value is higher than the reported post annealed Ho₂O₃/Si interface $D_{it} \sim 3.00 \times 10^{11} \text{ eV}^{-1} \text{cm}^{-2}$ (Pan, Chang, et al., 2010). However, a lower $Q_{eff} \sim 8.80 \times 10^{12} \text{ cm}^{-2}$, $Q_{it} \sim 2.93 \times 10^{12} \text{ cm}^{-2}$, $D_{it} \sim 10^{13} \text{ eV}^{-1} \text{cm}^{-2}$ (at mid $E_C - E = 0.2 \text{ eV}$) and $D_{total} \sim 1.97 \times 10^{13} \text{ cm}^{-2}$ has been acquired in this work in counter to

thermally oxidized Sm₂O₃/Si interface (Goh et al., 2017b) [$Q_{eff} \sim 2.81 \times 10^{13}$ cm⁻², $Q_{it} \sim 5.56 \times 10^{12}$ cm⁻² , $D_{it} \sim 10^{14}$ eV⁻¹cm⁻² (at mid $E_C - E = 0.2$ eV) and $D_{total} \sim 7.30 \times 10^{13}$ cm⁻²]. Based on the above discussion it can be concluded that 10 min oxidation/nitridation duration holds the superior electrical properties of *C-V* characteristics for Ho₂O₃/IL/Ge gate stack.



Figure 4.52: Average interface trap density D_{it} and total interface density contained in Ho₂O₃/IL/Ge MOS capacitors at different oxidation/nitridation durations (5 – 20) min.

4.5 Comparison Between Oxidized/Nitrided Sputtered Sm/Ge and Ho/Ge Stack

The comparison of Sm₂O₃/Ge and Ho₂O₃/Ge stack has been demonstrated by comparing the physical and electrical properties obtained at their optimized thermal oxidation/nitridation deposition condition in N₂O ambient. After conducting thermal oxidation/nitridation of sputtered Sm/Ge and Ho/Ge at different temperatures and durations the best possible thermal deposition condition has been achieved as shown in Table 4.1. The optimized thermal oxidation/nitridation duration at constant temperature 400 °C for oxidized/nitrided Sm/Ge is achieved at 15 min while 10 min thermal oxidation/nitridation duration has been optimized for oxidized/nitrided sputtered Ho/Ge. The comparison has been considered based on the physical and electrical properties achieved at optimized condition.

Thermal Oxidation and Nitridation	Oxidation Ni- tridation Ambient	Constant Temperature (°C)	Duration (min)
Sputtered Sm/Ge	N-O	400	15
Sputtered Ho/Ge	1N2O -	400	10

 Table 4.1: Optimum thermal oxidation/nitridation durations for sputtered Sm/Ge and Ho/Ge at constant 400 °C.

4.5.1 Comparison of Physical Properties

The comparison of several physical properties for thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge includes the findings of XRD, XPS and HRTEM are summarized in Table 4.2. From the comparison of the findings of XRD it is found that after oxidation/nitridation in N₂O ambient trigonal phase of Sm₂O₃ has been formed with a preferential plane orientation of (003) for oxidized/nitrided Sm thin film while cubic Ho₂O₃ phase structure corresponding to preferential plane of (631) is being produced for

the oxidized/nitrided Ho thin film. Moreover, from XRD analysis it is also found that Ho₂O₃ introduces larger crystallite size than Sm₂O₃ while both Sm₂O₃ and Ho₂O₃ oxides comprises negative micro-strain of same value. According to the XPS analysis for oxidized/nitrided Sm/Ge in N₂O ambient, exposed the formation of stoichiometric Sm₂O₃ with an interfacial layer (IL) consisting of sub-stoichiometric mixed compounds Sm_xGe_vO_z, native oxide GeO₂, suboxides GeO_x, with Ge₃N₄ has been formed. Meanwhile, sub-stochiometric Ho₂O₃ along with IL of intermixing sub-stoichiometric content of native oxide GeO₂, sub-oxides GeO_x and Ge₃N₄. Unlike interfacial Sm-O-Ge of Sm₂O₃/IL/Ge stack no such interfacial rare earth germanate (Ho-O -Ge) was formed in the Ho₂O₃/IL/Ge stack. Moreover, the band alignment mapping extracted using XPS valence spectra and O1s plasmon loss XPS spectra it is found that Sm2O3/IL interface exhibited higher CBO, ΔE_c than Ho₂O₃ /IL interface with respect to Ge substrate. From the band offset mapping it is also noticed that Sm₂O₃/IL exposed more symmetrical band alignment than the Ho₂O₃/IL interface. Although Ho₂O₃/IL interface exhibited larger energy band gap, E_q than Sm₂O₃/IL but the reduction of ΔE_c for Ho₂O₃ /IL interface indicates that relatively more defect gap states has been introduced between the conduction band edge of Ho₂O₃/IL interface and substrate Ge. Eventually, it is conspicuous that thermally oxidized/nitrided sputtered Sm/Ge generates more thermally stable Sm₂O₃/IL interface than that of Ho₂O₃/IL interface. Lastly, coming to the HRTEM analysis it is found that both oxidized/nitrided Sm/Ge and Ho/Ge exhibits distinguishable bulk oxide and interfacial layer where Sm₂O₃/IL/Ge stack exhibited the lower total oxide thickness including lower thickness of both bulk and interfacial layer compared to Ho₂O₃ /IL/Ge stack. Both of these stacks induced amorphous interface for both bulk oxide and interfacial layer.

Analysis Technique	Features	Thermally oxi- dized/nitrided sputtered Sm/Ge	Thermally oxi- dized/nitrided sputtered Ho/Ge
XRD	Phase of Ox- ide	Trigonal-Sm ₂ O ₃	Cubic-Ho ₂ O ₃
	Plane Orienta- tion	(100), (003), (103), (110), (101) & (200)	(222), (631), (440), (411), (512), (431), (602), (541) & (622)
	Preferred Plane orienta- tion	(003)	(631)
	Crystallite size (D) (nm)	32.48	41.25
XPS	Chemical Composition of bulk	<u>- 0.0004</u> Sm–O	- 0.0004 Но-О
	Chemical Composition of IL	Sm _x Ge _y O _z , Ge–O & Ge–N	Ge–O & Ge-N
	Dielectric Stoichiometry	Stoichiometric Sm ₂ O ₃	Sub-stoichiometric Ho ₂ O ₃
	$ \frac{E_g \text{ (eV) of}}{\text{Sm}_2\text{O}_3/\text{IL \&}} $ Ho2O3/IL	6.25	7.81
	$\frac{\Delta E_c \text{ (eV) of}}{\Delta E_c \text{ (eV) of}}$	2.60	2.47
	$\frac{\Delta E_{v} \text{ (eV) of}}{\Delta E_{v} \text{ (eV) of}}$ $\frac{\Delta E_{v} \text{ (eV) of}}{Sm_{2}O_{3}/IL \&}$ $Ho_{2}O_{3}/IL$	2.98	4.67
HRTEM	Bulk thickness	3.56	4.33
	Interfacial layer thick- ness, t_{ii} (nm)	2.44	4.39
	Total oxide	6	8.72
	Crystallinity of bulk and IL	Amorphous	Amorphous
	Crystallinity of bulk and IL	Amorphous	Amorphous

Table 4.2: Comparison of physical properties between thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge.

4.5.2 Comparison of Electrical Properties

The electrical properties of thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge is shown in Table 4.3. According to the C-V features, it is noticed that $Sm_2O_3/IL/Ge$ gate stack enhanced the gate capacitance, C_{ox} than the Ho₂O₃/IL/Ge gate stack. This has been attributed to the lowering of total oxide thickness and increment of k values of the gate stack. From the Table 4.2, it is noticed that $Sm_2O_3/IL/Ge$ gate stack induces more than double k value than the Ho₂O₃/IL/Ge gate stack. Additionally, from the C-V curve of Ho₂O₃/IL/Ge gate stack it is observed that the flat band voltage, (ΔV_{fb}) shifted negatively inducing positive fixed oxide charge (Q_{eff}) with the bulk Ho₂O₃ interface. On other hand, C-V curve Sm₂O₃/IL/Ge gate stack indicates the presence of negative Q_{eff} within the bulk Ho₂O₃ interface by a positive shift of ΔV_{fb} . Since Ho₂O₃/IL/Ge exhibited larger Q_{eff} which caused wider ΔV_{FB} shift for Ho₂O₃/IL/Ge gate stack. As of for Sm₂O₃/IL/Ge gate stack relatively lower hysteresis and slow trap density (Q_{it}) has been noticed than the Ho₂O₃/IL/Ge indicating less transmission of carrier charge from the Ge substrate. Moreover, Ho₂O₃/IL/Ge gate stacks generates higher D_{it} and D_{total} in counterpart to Sm₂O₃/IL/Ge. This attributes to the existence of more Ge dangling bonds and oxygen vacancies which induced enormous, charged traps within the Ho₂O₃/IL interface. However, the D_{it} is quite high for both the samples. Now, comparing the J-E properties it is observed that Sm₂O₃/IL/Ge gate stack induces lower leakage current density at a higher electrical breakdown field than the Ho₂O₃/IL/Ge gate stack. Comparatively lower Q_{eff}, Q_{it}, D_{it} and higher ΔE_c contributed to improve the *J*-*E* properties for Sm based gate stack. Additionally, the improvement of electrical properties for Sm₂O₃/IL/Ge gate stack in counter to Ho₂O₃/IL/Ge gate stack is due to relatively more thermally stable and less defective Sm₂O₃/IL interface formation. This can be attributed to the homogeneity of Sm₂O₃ and trapping effect of interfacial Sm–O–Ge effectively reduced oxygen-defects

by restricting GeO desorption. Besides, valency passivation approach of Sm-O-Ge and formation of interfacial Ge_3N_4 due to activation of N incorporation also lessened the Ge dangling bonds.

Electrical Properties	Unit	Sm ₂ O ₃ /IL/Ge	Ho ₂ O ₃ /IL/Ge
Capacitance	μF/cm ²	Gate stack	Gale stack
C_{ox}		4.60	1.38
Dielectric Constant, k	-	31.19	13.60
Leakage Current Density, J_g	A cm ⁻²	8.38×10^{-6}	5.34×10^{-5}
Electrical Breakdown Field, <i>E_{BD}</i>	MV cm ⁻²	13.31	8.59
Fixed Oxide Charge, <i>Q_{eff}</i>	cm ⁻²	-2.88×10^{12}	8.80×10^{12}
Slow trap Density, <i>Q_{it}</i>	cm ⁻²	1.41×10^{12}	2.93×10^{12}
Average interface trap density, D _{it}	$eV^{-1} cm^{-2}$	4.84×10^{13}	9.39 × 10 ¹³
Total interface trap den- sity, <i>D_{total}</i>	cm ⁻²	1.20×10^{13}	1.97×10^{13}

 Table 4.3: Comparison of electrical features between thermally oxidized/nitrided sputtered Sm/Ge and Ho/Ge.

CHAPTER 5: CONCLUSION AND RECOMMENDATIONS

5.1 Novelty and Significance of the Findings

For last few decades the gate oxide thickness of Si based MOS technology has been continuously downscaled to cater the ongoing demand of low power consumption and high speed devices. Si based MOS technology has reached its potential limit due to drastic downscaling of SiO₂ and low hole mobility of Si. Accordingly, higher hole carrier alternative Ge and high-k gate oxide has gained widespread importance to replace SiO_2/Si . Various REOs with combination of Ge substrate has been widely investigated by other researchers for last few years, but REOs/Ge MOS devices are subjected to several limitations in terms of electrical and thermodynamic properties when compared to conventional SiO₂/Si MOS device. Among the REOs, both Sm₂O₃ and Ho₂O₃ owns promising features to be used as high-k oxides for MOS devices, owing higher dielectric constant k value, higher band gap value with less hygroscopic behaviour. Despite of having such attractive features the investigation of Sm₂O₃/Ge is very limited, while Ho₂O₃/Ge has not been reported till to date using the industrial deposition method of thermal oxidation and nitridation. Therefore, this study investigated the deposition condition of thermal oxidation and nitridation for sputtered Sm/Ge and Ho/Ge based on the electrical, physical, and thermodynamic properties. Moreover, the band alignment mapping and underlying thermal oxidation and nitridation mechanism of oxidized/nitrided Sm/Ge and Ho/Ge has also been reported.

In this study, Sm_2O_3/Ge stack based capacitor prepared from thermally oxidized/nitrided sputtered metallic Sm on Ge semiconductor in N₂O ambient for several oxidation/nitridation temperatures (300 – 600) °C and durations (5 – 20) minutes have been

comprehensively investigated. Besides, the influence of different thermal oxidation/nitridation durations (5 - 20) minutes in N₂O gas ambient for transforming metallic Ho sputtered on Ge substrate have been systemically investigated to develop Ho₂O₃/Ge based on metal-oxide-semiconductor (MOS) device. The film crystallinity, chemical composition and interface chemical bonding states stability was characterized from X-ray diffraction (XRD), Raman spectroscopy and X-ray photoelectron spectroscopy (XPS). Structural morphology characterization using high resolution transmission electron microscope (HRTEM). The optimum deposition condition for thermally oxidized/nitrided Sm/Ge was achieved at 400 °C and 15 min. On the contrary, a duration of 10 min at constant temperature 400 °C has been optimized for thermally oxidized/nitrided Ho/Ge. From the findings of electrical properties, it is apparent that the optimized Sm₂O₃/IL/Ge stack in this investigation provides improved dielectric constant, $k \sim 31.19$ eV, leakage current density, $J_g \sim 8.38 \times 10^{-6}$ A cm⁻² at enhanced breakdown field, E_{BD} of 13.31 MV cm⁻¹ compared to the previous reported work of (Goh et al., 2017b; C. C. Lin et al., 2014). Besides, EOT as low as 0.75 nm and higher $\Delta E_c \sim 2.60$ eV has been achieved which is complementary for the ongoing MOS scaling trend. Moreover, the optimized Ho₂O₃/Ge gate stack in this investigation also exposed higher electrical breakdown, $E_{BD} \sim 8.59 \text{ MV cm}^{-1}$ ¹ and higher dielectric constant $k \sim 13.60$ comparing with previous reported works, (Pan, Chang, et al., 2010). Conduction band offset of $\Delta E_c \sim 2.60$ eV and $\Delta E_c \sim 2.47$ eV has been achieved for Sm₂O₃/IL/Ge and Ho₂O₃/IL/Ge, respectively which impeded the electron tunneling through gate oxide eventually contributed on reducing gate leakage current density.

5.1.1 Effect of Various Thermal Oxidation/Nitridation Temperature on Sputtered Sm Thin Film Based on Ge Substrate

In summary, a comprehensive study of Sm₂O₃/IL/Ge gate stack prepared at various thermal oxidation and nitridation temperature (300 - 600) °C in N₂O ambient were presented considering the physical and electrical characteristics of MOS devices. Based on XRD and XPS characterization formation of Trigonal Sm₂O₃ and presence of interfacial layer comprising of Ge-O, Ge-N, and Sm-O-Ge were identified. From XPS. It was found that thermal oxidation and nitridation at 400 °C was more favourable to form a uniform amorphous Sm₂O₃ film and more stoichiometric thermally stable amorphous interfacial Sm_xGe_vO_z with less GeO_x and interface defects. It was evident that, formation of uniform Sm_xGe_yO_z plays vital role in suppressing desorption of volatile GeO. Thus, at this temperature moderate C-V characteristic were obtained with higher capacitance and higher dielectric constant, less flat band voltage shift, low slow trap density with reduced average interface trap density and frequency dispersion in counter to another investigated sample. The band alignment at this temperature shows higher conduction band offset, ΔE_c and high enough VBO, ΔE_v . Hence, the high CBO impeded electron tunneling and enhanced the electric breakdown field, $E_{BD} \sim 13.31$ at low leakage current density $J_q \sim$ 8.38×10^{-6} A cm⁻². The overall improvement at 400 °C has been attributed to the sufficient N₂O incorporation to reach full oxidation level that produced stoichiometric Sm₂O₃ and stable interfacial Sm–O–Ge eventually minimized unbonded Sm³⁺ and Ge defects that maintained the integrity of stacking interface. Based on the results from XRD, XPS and HRTEM a possible oxidation and nitridation model has been proposed. Therefore, it is suggested that deposition temperature of 400 °C is more favourable for thermal oxidation/nitridation of sputtered Sm/Ge in N2O ambient for developing Sm2O3/Ge gate stack.

5.1.2 Effect of Various Thermal Oxidation/Nitridation Durations on Sputtered Sm Film Based on Ge Substrate

The impact of different oxidation/nitridation durations (5-20) minutes on structural characteristics, chemical bonding states, band alignment mapping and electrical features for oxidizing/nitriding sputtered metal Sm on Ge prepared in N₂O ambient at 400 °C were presented. On the basis of XPS, Raman, XRD findings formation of Sm₂O₃ thin film layer with incorporation of an interfacial layer involving corelated content of Ge-O and Ge-N was confirmed for all the investigated durations. An additional interfacial Sm-O-Ge content was only detected for the extended durations of 15 and 20 min validated from Raman and XPS analysis. Besides, XRD analysis confirmed trigonal phase of Sm₂O₃, tetragonal and trigonal phase of GeO₂ and cubic phase of Ge₃N₄. The HRTEM image also supported the formation double stacked layer film involving a Sm₂O₃ dielectric layer mounted on IL interface. The XPS analysis indicated that 15 min oxidation/nitridation induced more thermally stable and less defective interface due to the growth of more stochiometric Sm₂O₃ and reduction of sub-stoichiometric GeO_x. Additionally, based on Raman and XPS analysis it was anticipated that formation of denser interfacial Ge_3N_4 and growth of stable $Sm_xGe_vO_7$ assisted the alleviation of oxygen vacancies and Ge dangling eventually suppressed GeO (g) volatilization at 15 min oxidation/nitridation duration. The electrical features also clearly suggests that 15 min is more convenient oxidation/nitridation condition at constant temperature of 400 °C to achieve superior C-Vproperties and smallest gate leakage current density of 10^{-6} A cm⁻² order magnitude at breakdown field $E_{BD} \sim 13.31 \text{ MV cm}^{-1}$.

Moreover, this duration also revealed a symmetrical band alignment mapping with superior $\Delta E_c \sim 2.60 \text{ eV}$ and $\Delta E_v \sim 2.98 \text{ eV}$ which reduced the gate leakage. Furthermore, a *EOT* of 0.75 nm was achieved for this duration which is correlated to lower interfacial thickness observed in HRTEM analysis and higher dielectric constant, $k \sim 31.19 \text{ eV}$ obtained from *C-V* curve. Overall, the improvement in *C-V* features, *J-E* characteristic, and ΔE_c for 15 minutes has been ascribed due to suppression of GeO (g), lessening of defective GeO_x while lowering the oxygen associated traps within Sm₂O₃ film. Contrariwise shorter oxidation/nitridation duration such as 10 minutes degrades the homogeneity of Sm₂O₃ and triggers detrimental GeO (g) volatilization along with severe GeO_x defects formation thereby, worsening the interface quality and downgraded the electrical features. Therefore, as concluding remark it is suggested that 15 minutes thermal oxidation/nitridation duration is the optimum requirement in N₂O ambient at 400 °C for Sm₂O₃/Ge based MOS gate stack.

5.1.3 Effect of Various Thermal Oxidation/Nitridation Durations on Sputtered Ho Film Based on Ge Substrate

In this work, investigation of chemical compositions, structural morphology and electrical phenomena of Ho₂O₃/IL/Ge structure were fabricated through oxidation/nitridation in N₂O gas ambient at various durations (5 – 20) min has been reported. According to the XRD and XPS analysis growth of sub-stoichiometric cubic -Ho₂O₃ and intermixing interfacial layer consisting of Ge sub-oxides GeO_x, native oxide GeO₂ and Ge₃N₄ has been confirmed. XPS analysis demonstrates that for 10 min sample formation of relatively stoichiometric Ho₂O₃ rather than sub-stoichiometric Ge–O contents helped to improve the thermal stability and interface defect alleviation. It was also apparent that formation of Ge₃N₄ rich interfacial layer during the oxidation/nitridation process aided in blocking vacancy diffusivity which ultimately suppressed the volatile GeO gas desorption. The band alignment clearly evidenced that this oxidation/nitridation condition is also advantageous to attain high band offsets ($\Delta E_c \sim 2.47 \text{ eV}$ and $\Delta E_v \sim 4.67 \text{ eV}$). Thus, lowest leakage current density $J_g \sim 5.34 \times 10^{-5} \text{ A cm}^{-2}$ and higher electric breakdown field $E_{BD} \sim 8.59$ MV cm⁻¹ were achieved. Moreover, this sample induces higher gate oxide capacitance, larger effective dielectric constant $k \sim 13.60$ and relatively lower interface trap density which confirmed that 10 min oxidation/nitridation condition is more favourable to increase the growth of high-k Ho₂O₃ while maintaining structural homogeneity. On the contrary longer oxidation/nitridation such as 20 min sample deteriorates the uniformity of Ho₂O₃ and favours the growth of detrimental Ge–O species thereby, degrades the thermal stability and electrical properties. Thickness estimated from HRTEM also reveals that 10 min oxidation/nitridation induced thinner Ho₂O₃/IL ~ 8.72 nm but needed to be further downscaled for being acceptable with the current scaling trend of MOS devices. Therefore, it can be concluded that oxidized/nitrided sample at 400 °C for 10 min duration in N₂O ambient is the optimized condition for Ho₂O₃/IL/Ge gate stack.

5.1.4 Comparison Between Thermally Oxidized/Nitrided Sputtered Sm/Ge and Ho/Ge

According to the findings achieved in terms of physical and electrical properties, the optimum thermal oxidation/nitridation durations for oxidized/nitrided Sm/Ge and Ho/Ge are 15 and 10 minutes, respectively at constant temperature 400 °C. The obtained physical and electrical properties of both oxidized/nitrided Sm/Ge and Ho/Ge have been summarized in Table 4.2 & 4.3. As of for the for Sm based sample at optimum deposition condition it is found that stoichiometric, trigonal-Sm₂O₃ has been formed while sub-stochiometric cubic Ho₂O₃ has been detected for Ho based sample. Additionally, both the gate stack includes an interfacial layer beneath the dielectric interface where oxidized/nitrided Sm/Ge induced both the lower bulk oxide and IL thickness than oxidized/nitrided Ho/Ge stack. The Sm₂O₃ /IL/Ge stack exposing higher CBO, with symmetrical band alignment mapping and higher dielectric constant, inducing lower leakage current density at high electrical breakdown field and higher gate capacitance than Ho₂O₃/IL/Ge. Although both Sm₂O₃ /IL/Ge and Ho₂O₃/IL/Ge gate stack experienced high interface trap density which requires further investigation for improvement. Accordingly, Sm based gate stack exhibits more thermally stable Sm₂O₃ /IL interface with higher electrical properties in counter to Ho₂O₃/IL interface of Ho based gate stack. According to the physical and electrical properties found in this study constitutes that both Sm₂O₃ and Ho₂O₃ could serve as dielectric material for future high-*k*/Ge stack based metal oxide semiconductor technology. Overall, it can be concluded that Sm₂O₃ is a superior choice in counter to Ho₂O₃ for developing Ge based gate stack using thermal oxidation/nitridation.

5.2 Recommendation for Future Research

The findings of this investigation are anticipated to provide new insight on the realization of REOs for MOS application which requires comprehensive evaluation for future revolution of MOS devices. The following concepts are suggested for future prospective research.

- (i) It will be interesting to investigate the impact of post deposition annealing (PDA) temperature in various gas ambient (such as Ar, N₂, and O₂) for further improvement of thermally oxidized/nitrided sputtered Sm/Ge or Ho/Ge.
- (ii) The influence of N₂O gas concentration on the electrical and physical properties for sputtered Sm/Ge or Ho/Ge following thermal oxidation/nitridation could further advance the process of utilizing Sm₂O₃ and Ho₂O₃ as alternative High-k.
- (iii) Thermal oxidation/nitridation for developing bilayer gate stack with combination of (Sm₂O₃ + TMOs)/Ge or (Ho₂O₃ + TMOs)/Ge where TMOs such as Al₂O₃ or ZrO₂ could be utilized for realization of Ge based MOS devices.

(iv) To gain further insight for Sm₂O₃/Ge or Ho₂O₃/Ge gate stack the reliability evaluation in terms of bias temperature instability (BTI) performance and current conduction mechanism will be another potential aspect for further research.

REFERENCES

- Adachi, G. Y., & Imanaka, N. (1998). The binary rare earth oxides. *Chemical Reviews*, 98(4), 1479–1514. doi:10.1021/cr940055h
- Alers, G. B., Fleming, R. M., Wong, Y. H., Dennis, B., Pinczuk, A., Redinbo, G., ... Hasan, Z. (1998). Nitrogen plasma annealing for low temperature Ta₂O₅ films. *Applied Physics Letters*, 72(11), 1308–1310. doi:10.1063/1.120569
- Arora, R., Schmidt, B. W., Fleetwood, D. M., Schrimpf, R., Galloway, K., Rogers, B., ... Lucovsky, G. (2009). Temperature Stress Response of Germanium MOS with HfSiON Dielectric. ECS Transactions, 19(2), 803–814. doi:10.1149/ma2009-01/19/807
- Balakrishnan, G., Sudhakara, P., Wasy, A., Ho, H. S., Shin, K. S., & Song, J. I. (2013). Epitaxial growth of cerium oxide thin films by pulsed laser deposition. *Thin Solid Films*, 546, 467–471. doi:10.1016/j.tsf.2013.06.048
- Baldovino, S., Molle, A., & Fanciulli, M. (2008). Evidence of dangling bond electrical activity at the Ge/oxide interface. *Applied Physics Letters*, 93(24), 7–10. doi:10.1063/1.3050451
- Batra, Y., Kabiraj, D., & Kanjilal, D. (2007). Charge retention and optical properties of Ge nanocrystals embedded in GeO₂ matrix. *Solid State Communications*, 143(4–5), 213–216. doi:10.1016/j.ssc.2007.05.026
- Bellenger, F., Houssa, M., Delabie, A., Afanasiev, V., Conard, T., Caymax, M., ... Heyns, M. M. (2008). Passivation of Ge(100)/GeO₂/high-κ Gate Stacks Using Thermal Oxide Treatments. *Journal of The Electrochemical Society*, 155(2), G33. doi:10.1149/1.2819626
- Bethge, O., Zimmermann, C., Lutzer, B., Simsek, S., Smoliner, J., Stöger-Pollach, M., ... Bertagnolli, E. (2014). Effective reduction of trap density at the Y₂O₃/Ge interface by rigorous high-temperature oxygen annealing. *Journal of Applied Physics*, *116*(21), 214111. doi:10.1063/1.4903533
- Bhatt, P., Chaudhuri, K., Kothari, S., Nainani, A., & Lodha, S. (2013). Germanium oxynitride gate interlayer dielectric formed on Ge(100) using decoupled plasma nitridation. *Applied Physics Letters*, 103(17). doi:10.1063/1.4826142
- Bom, N. M., Soares, G. V., Krug, C., Pezzi, R. P., Baumvol, I. J. R., & Radtke, C. (2012). Evolution of the Al₂O₃ /Ge(1 0 0) interface for reactively sputter-deposited films submitted to postdeposition anneals. *Applied Surface Science*, 258(15), 5707–5711. doi:10.1016/j.apsusc.2012.02.066
- Botzakaki, M. A., Skoulatakis, G., Xanthopoulos, N., Gianneta, V., Travlos, A., Kennou, S., ... Krontiras, C. A. (2018). Influence of the atomic layer deposition temperature on the structural and electrical properties of Al/Al₂O₃/p-Ge MOS structures. *Journal* of Vacuum Science & Technology A: Vacuum, Surfaces, and Films, 36(1), 01A120. doi:10.1116/1.5003375

- Brunco, D. P., Dimoulas, A., Boukos, N., Houssa, M., Conard, T., Martens, K., ... Heyns, M. M. (2007). Materials and electrical characterization of molecular beam deposited CeO₂ and CeO₂/ HfO₂ bilayers on germanium. *Journal of Applied Physics*, 102(2), 4104. doi:10.1063/1.2756519
- Cao, Y. Q., Wu, B., Wu, D., & Li, A. D. (2017). Interfacial, Electrical, and Band Alignment Characteristics of HfO₂/Ge Stacks with In Situ-Formed SiO₂ Interlayer by Plasma-Enhanced Atomic Layer Deposition. *Nanoscale Research Letters*, 12(370), 1–7. doi:10.1186/s11671-017-2083-z
- Castan, H., Garcia, H., Duenas, S., Bailon, L., Miranda, E., Kukli, K., ... Leskela, M. (2015). Conduction and stability of holmium titanium oxide thin films grown by atomic layer deposition. *Thin Solid Films*, 591, 55–59. doi:10.1016/j.tsf.2015.08.027
- Caymax, M., Houssa, M., Pourtois, G., Bellenger, F., Martens, K., Delabie, A., & Van Elshocht, S. (2008). Interface control of high-k gate dielectrics on Ge. *Applied Surface Science*, 254(19), 6094–6099. doi:10.1016/J.APSUSC.2008.02.134
- Chandra, S. V. J., Jeong, M. I., Park, Y. C., Yoon, J. W., & Choi, C. J. (2011). Effect of annealing ambient on structural and electrical properties of Ge metal-Oxide-Semiconductor capacitors with Pt gate Electrode and HfO2 gate dielectric. *Materials Transactions*, 52(1), 118–123. doi:10.2320/matertrans.M2010324
- Chen, F. H., Hung, M. N., Yang, J. F., Kuo, S. Y., Her, J. L., Matsuda, Y. H., & Pan, T. M. (2013). Effect of surface roughness on electrical characteristics in amorphous InGaZnO thin-film transistors with high-κ Sm₂O₃ dielectrics. *Journal of Physics and Chemistry of Solids*, 74(4), 570–574. doi:10.1016/j.jpcs.2012.12.006
- Cheng, C. C., Chien, C. H., Luo, G. L., Yang, C. H., Kuo, M. L., Lin, J. H., ... Chang, C. Y. (2007). Study of Thermal Stability of HfO_xNy/Ge Capacitors Using Postdeposition Annealing and NH₃ Plasma Pretreatment. *Journal of The Electrochemical Society*, 154(7), G155. doi:10.1149/1.2734875
- Cheng, Z. X., Liu, L., Xu, J. P., Huang, Y., & Lai, P. T. (2016). Impact of Nitrogen Incorporation on the Interface Between Ge and La₂O₃ or Y₂O₃ Gate Dielectric: A Study on the Formation of Germanate. *IEEE Transactions on Electron Devices*, 63(12), 4888–4892. doi:10.1109/TED.2016.2618221
- Chew, C. C., Goh, K. H., Gorji, M. S., Tan, C. G., Ramesh, S., & Wong, Y. H. (2016). Breakdown field enhancement of Si-based MOS capacitor by post-deposition annealing of the reactive sputtered ZrOxNy gate oxide. *Applied Physics A: Materials Science and Processing*, 122(2), 1–6. doi:10.1007/S00339-016-9624-7
- Chin, W. C., & Cheong, K. Y. (2011). Effects of post-deposition annealing temperature and ambient on RF magnetron sputtered Sm₂O₃ gate on n-type silicon substrate. *Journal of Materials Science: Materials in Electronics*, 22(12), 1816–1826. doi:10.1007/s10854-011-0368-z
- Chin, W. C., Cheong, K. Y., & Hassan, Z. (2010). Sm₂O₃ gate dielectric on Si substrate. *Materials Science in Semiconductor Processing*, 13(5–6), 303–314. doi:10.1016/j.mssp.2011.02.001

- Choi, K. J., Shin, W. C., Park, J. B., & Yoon, S. G. (2001). Electrical properties of hafnium oxide gate dielectric deposited by plasma enhanced chemical vapor deposition. *Integrated Ferroelectrics*, 38(1–4), 191–199. doi:10.1080/10584580108016932
- Chroneos, A., Schwingenschlögl, U., & Dimoulas, A. (2012). Impurity diffusion, point defect engineering, and surface/interface passivation in germanium. *Annalen Der Physik*, 524(3–4), 123–132. doi:10.1002/andp.201100246
- Chui, C., Kim, H., & Chi, D. (2002). A sub-400/spl deg/C germanium MOSFET technology with high-/spl kappa/dielectric and metal gate. *Digest. International Electron Devices Meeting*, 437–440. doi:10.1109/IEDM.2002.1175872
- Chui, C. O., Ramanathan, S., Triplett, B. B., McIntyre, P. C., & Saraswat, K. C. (2002). Germanium MOS capacitors incorporating ultrathin high-κ gate dielectric. *IEEE Electron Device Letters*, 23(8), 473–475. doi:10.1109/LED.2002.801319
- Constantinescu, C., Ion, V., Galca, A. C., & Dinescu, M. (2012). Morphological, optical and electrical properties of samarium oxide thin films. *Thin Solid Films*, *520*(20), 6393–6397. doi:10.1016/j.tsf.2012.06.049
- Copetti, G., & Soares, G. V. (2016). Stabilization of the GeO₂ /Ge Interface by Nitrogen Incorporation in a One-Step NO Thermal Oxynitridation. *ACS Applied Materials & Interfaces*, 8(40), 27339–345. doi:10.1021/acsami.6b09244
- Cui, X., Tuokedaerhan, K., Cai, H., & Lu, Z. (2022). Effect of Annealing Temperature on the Microstructure and Optical Properties of Lanthanum-Doped Hafnium Oxide. *Coatings*, 12(4). doi:10.3390/coatings12040439
- Da Silva, S. R. M., Rolim, G. K., Soares, G. V, Baumvol, I. J. R., & Krug, C. (2012). Oxygen transport and GeO₂ stability during thermal oxidation of Ge. *Applied Physics Letters*, *100*(19), 263111. doi:10.1063/1.4712619
- Dakhel, A. A. (2004). Dielectric and optical properties of samarium oxide thin films. *Journal of Alloys and Compounds*, 365(1–2), 233–239. doi:10.1016/S0925-8388(03)00615-7
- Deb, S. K., Dong, J., Hubert, H., McMillan, P. F., & Sankey, O. F. (2000). Raman spectra of the hexagonal and cubic (spinel) forms of Ge₃N₄: An experimental and theoretical study. *Solid State Communications*, *114*(3), 137–142. doi:10.1016/S0038-1098(00)00017-X
- Del Alamo, J. A. (2011). Nanometre-scale electronics with III-V compound semiconductors. *Nature*, 479(7373), 317–323. doi:10.1038/nature10677
- Deng, S., Xie, Q., Deduytsche, D., Schaekers, M., Lin, D., Caymax, M., ... Detavernier, C. (2011). Effective reduction of fixed charge densities in germanium based metaloxide-semiconductor devices. *Applied Physics Letters*, 99(5), 052906. doi:10.1063/1.3622649
- Dimoulas, A., Brunco, D. P., Ferrari, S., Seo, J. W., Panayiotatos, Y., Sotiropoulos, A., ... Heyns, M. M. (2007). Interface engineering for Ge metal-oxide-semiconductor

devices. Thin Solid Films, 515(16), 6337-6343. doi:10.1016/j.tsf.2006.11.129

- Dimoulas, A., Tsoutsou, D., Panayiotatos, Y., Sotiropoulos, A., Mavrou, G., Galata, S. F., & Golias, E. (2010). The role of la surface chemistry in the passivation of Ge. *Applied Physics Letters*, 96(1), 11–14. doi:10.1063/1.3284655
- Dushaq, G., Nayfeh, A., & Rasras, M. (2018). Passivation of Ge/high-κ interface using RF Plasma nitridation. *Semiconductor Science and Technology*, 33(1), 015003. doi:10.1088/1361-6641/aa98cd
- Dushaq, G., Rasras, M., & Nayfeh, A. (2016). Impact of N₂O/NH₃/N₂ Gas Mixture on the Interface Quality of Germanium MOS Capacitors. *ECS Transactions*, 75(8), 661–666. doi:10.1149/07508.0661ecst
- Engstrom, O., Raeissi, B., Hall, S., Buiu, O., Lemme, M. C., Gottlob, H. D. B., ...
 Cherkaoui, K. (2007). Navigation aids in the search for future high-k dielectrics:
 Physical and electrical trends. *Solid-State Electronics*, 51(4), 622–626.
 doi:10.1016/j.sse.2007.02.021
- Etcheverry, L. P., Boudinov, H. I., & Soares, V. (2019). Combining GeO₂ passivation strategies aiming at dielectric layers with superior properties on germanium substrates. *Journal of Materials Chemistry C*, 7(27), 8465–8470. doi:10.1039/c9tc01831j
- Evangelou, E. K., Mavrou, G., Dimoulas, A., & Konofaos, N. (2007). Rare earth oxides as high-k dielectrics for Ge based MOS devices: An electrical study of Pt/Gd₂O₃/Ge capacitors. *Solid-State Electronics*, *51*(1), 164–169. doi:10.1016/j.sse.2006.10.011
- Evangelou, E. K., Rahman, M. S., & Dimoulas, A. (2009). Correlation of charge buildup and stress-induced leakage current in cerium oxide films grown on Ge (100) substrates. *IEEE Transactions on Electron Devices*, 56(3), 399–407. doi:10.1109/TED.2008.2011935
- Evangelou, E. K., Wiemer, C., Fanciulli, M., Sethu, M., & Cranton, W. (2003). Electrical and structural characteristics of yttrium oxide films deposited by rf-magnetron sputtering on. *Journal of Applied Physics*, 94(1), 318–325. doi:10.1063/1.1580644
- Fei, C., Liu, H., Wang, X., Zhao, D., & Wang, S. (2016). Influences of rapid thermal annealing on the characteristics of Al₂O₃\La₂O₃\Si and La₂O₃\Al₂O₃\Si films deposited by atomic layer deposition. *Journal of Materials Science: Materials in Electronics*, 27(8), 8550–8558. doi:10.1007/S10854-016-4872-Z
- Fissel, A., Osten, H. J., & Bugiel, E. (2003). Towards understanding epitaxial growth of alternative high-K dielectrics on Si(001): Application to praseodymium oxide. *Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures*, 21(4), 1765. doi:10.1116/1.1589516
- Frank, D., & Taur, Y. (2002). Design considerations for CMOS near the limits of scaling. *Solid-State Electronics*, *46*(3), 315–320.

- Fu, C. H., Chang-Liao, K. S., Liu, L. J., Li, C. C., Chen, T. C., Cheng, J. W., & Lu, C. C. (2014). An ultralow EOT Ge MOS device with tetragonal HfO₂ and high quality HfxGeyO interfacial layer. *IEEE Transactions on Electron Devices*, 61(8), 2662– 2667. doi:10.1109/TED.2014.2329839
- Fukuda, Y., Okamoto, H., Iwasaki, T., Otani, Y., & Ono, T. (2011). Surface passivation of p-type Ge substrate with high-quality GeN_x layer formed by electron-cyclotronresonance plasma nitridation at low temperature. *Applied Physics Letters*, 99(13), 2– 5. doi:10.1063/1.3647621
- Gao, J., He, G., Fang, Z. B., Lv, J. G., Liu, M., & Sun, Z. Q. (2017). Interface quality modulation, band alignment modification and optimization of electrical properties of HfGdO/Ge gate stacks by nitrogen incorporation. *Journal of Alloys and Compounds*, 695, 2199–2206. doi:10.1016/j.jallcom.2016.11.068
- Gao, Juan., He, G., Sun, Z., Chen, H., Zheng, C., Jin, P., ... Liu, M. (2016). Modification of electrical properties and carrier transportation mechanism of ALD-derived HfO₂/Si gate stacks by Al₂O₃ incorporation. *Journal of Alloys and Compounds*, 667, 352–358. doi:10.1016/J.JALLCOM.2016.01.171
- Gao, Juan., He, G., Xiao, D., Jin, P., Jiang, S., Li, W., ... Zhu, L. (2017). Passivation of Ge surface treated with trimethylaluminum and investigation of electrical properties of HfTiO/Ge gate stacks. *Journal of Materials Science and Technology*, 33(8), 901– 906. doi:10.1016/j.jmst.2017.04.021
- Gillen, R., & Robertson, J. (2013). Electronic structure of lanthanide oxide high K gate oxides. *Microelectronic Engineering*, 109, 72–74. doi:10.1016/J.MEE.2013.03.011
- Goh, K. H., Haseeb, A. S. M. A., & Wong, Y. H. (2016a). Effect of Oxidation Temperature on Physical and Electrical Properties of Sm₂O₃ Thin-Film Gate Oxide on Si Substrate. *Journal of Electronic Materials*, 45(10), 5302–5312. doi:10.1007/s11664-016-4694-z
- Goh, K. H., Haseeb, A. S. M. A., & Wong, Y. H. (2016b). Physical and electrical properties of thermal oxidized Sm₂O₃ gate oxide thin film on Si substrate: Influence of oxidation durations. *Thin Solid Films*, 606, 80–86. doi:10.1016/j.tsf.2016.03.051
- Goh, K. H., Haseeb, A. S. M. A., & Wong, Y. H. (2017a). Lanthanide rare earth oxide thin film as an alternative gate oxide. *Materials Science in Semiconductor Processing*, 68, 302–315. doi:10.1016/j.mssp.2017.06.037
- Goh, K. H., Haseeb, A. S. M. A., & Wong, Y. H. (2017b). Trap-assisted tunneling, capacitance–voltage characteristics, and surface properties of Sm₂O₃ thin film on Si substrate. *Journal of Materials Science: Materials in Electronics*, 28(6), 4725–4731. doi:10.1007/s10854-016-6115-8
- Goley, P. S., & Hudait, M. K. (2014). Germanium based field-effect transistors: Challenges and opportunities. *Materials*, 7(3), 2301–2339. doi:10.3390/ma7032301
- Gordon, R. G., Becker, J., Hausmann, D., & Suh, S. (2001). Vapor deposition of metal oxides and silicate: Posible gate insulators for future microelectronics. *Chemistry of Materials*, 13(8), 2463–2464. doi:10.1021/cm010145k

- Gullu, H. H., & Yildiz, D. E. (2021). Capacitance, conductance, and dielectric characteristics of Al/TiO₂/Si diode. *Journal of Materials Science: Materials in Electronics*, 32(10), 13549–13567. doi:10.1007/S10854-021-05931-5
- He, G., Chen, X., & Sun, Z. (2013). Interface engineering and chemistry of Hf-based high-k dielectrics on III-V substrates. *Surface Science Reports*, 68(1), 68–107. doi:10.1016/j.surfrep.2013.01.002
- He, G., Gao, J., Chen, H., Cui, J., Sun, Z., & Chen, X. (2014). Modulating the interface quality and electrical properties of HfTiO/InGaAs gate stack by atomic-layerdeposition-derived Al₂O₃ passivation layer. ACS Applied Materials and Interfaces, 6(24), 22013–22025. doi:10.1021/am506351u
- He, G., Liu, J., Chen, H., Liu, Y., Sun, Z., Chen, X., ... Zhang, L. (2014). Interface control and modification of band alignment and electrical properties of HfTiO/GaAs gate stacks by nitrogen incorporation. *Journal of Materials Chemistry C*, 2(27), 5299– 5308. doi:10.1039/c4tc00572d
- He, G., Zhu, L., Sun, Z., Wan, Q., & Zhang, L. (2011). Integrations and challenges of novel high-k gate stacks in advanced CMOS technology. *Progress in Materials Science*, 56(5), 475–572. doi:10.1016/j.pmatsci.2011.01.012
- Hetherin, K., Ramesh, S., & Wong, Y. H. (2017a). Effects of thermal oxidation duration on the structural and electrical properties of Nd₂O₃/Si system. *Applied Physics A: Materials Science and Processing*, 123(8), 510. doi:10.1007/s00339-017-1122-z
- Hetherin, K., Ramesh, S., & Wong, Y. H. (2017b). Formation of neodymium oxide by thermal oxidation of sputtered Nd thin film on Si substrate. *Journal of Materials Science: Materials in Electronics*, 28(16), 11994–12003. doi:10.1007/s10854-017-7009-0
- Hirose, M., Mizubayashi, W., Khairurrijal, Ikeda, M., Murakami, H., Kohno, A., ... Miyazaki, S. (2000). Ultrathin gate dielectrics for silicon nanodevices. *Superlattices and Microstructures*, 27(5), 383–393. doi:10.1006/spmi.2000.0861
- Hongo, T., Kondo, K. I., Nakamura, K. G., & Atou, T. (2007). High pressure Raman spectroscopic study of structural phase transition in samarium oxide. *Journal of Materials Science*, 42(8), 2582–2585. doi:10.1007/s10853-006-1417-5
- Hosoi, T., Kutsuki, K., Okamoto, G., Saito, M., Shimura, T., & Watanabe, H. (2009). Origin of flatband voltage shift and unusual minority carrier generation in thermally grown GeO₂/Ge metal-oxide-semiconductor devices. *Applied Physics Letters*, 94(20), 2–5. doi:10.1063/1.3143627
- Houssa, M., Pantisano, L., Ragnarsson, L. Å., Degraeve, R., Schram, T., Pourtois, G., ... Heyns, M. M. (2006). Electrical properties of high-κ gate dielectrics: Challenges, current issues, and possible solutions. *Materials Science and Engineering R: Reports*, 51(4–6), 37–85. doi:10.1016/j.mser.2006.04.001

- Houssa, M., Pourtois, G., Caymax, M., Meuris, M., & Heyns, M. M. (2008). Firstprinciples study of the structural and electronic properties of (100) Ge:Ge (M)O₂ interfaces (M=Al, La, or Hf). *Applied Physics Letters*, 92(24), 5–8. doi:10.1063/1.2944892
- Ieong, M., Doris, B., Kedzierski, J., Rim, K., & Yang, M. (2004). Silicon device scaling to the sub-10-nm regime. *Science*, 306(5704), 2057–2060. doi:10.1126/science.1100731
- Ioannou-Sougleridis, V., Constantoudis, V., Alexe, M., Scholz, R., Vellianitis, G., & Dimoulas, A. (2004). Effects on surface morphology of epitaxial Y2O3 layers on Si (001) after postgrowth annealing. *Thin Solid Films*, 468(1–2), 303–309. doi:10.1016/j.tsf.2004.05.076
- Iwai, H., Ohmi, S., Akama, S., Ohshima, C., Kikuchi, A., Kashiwagi, I., & Yoshihara, Y. (2002). Advanced gate dielectric materials for sub-100 nm CMOS. *Digest. International Electron Devices Meeting*, 626–628. doi:10.1109/IEDM.2002.1175917
- Jagadeesh Chandra, S. V., Choi, C. J., Uthanna, S., & Mohan Rao, G. (2010). Structural and electrical properties of radio frequency magnetron sputtered tantalum oxide films: Influence of post-deposition annealing. *Materials Science in Semiconductor Processing*, 13(4), 245–251. doi:10.1016/j.mssp.2010.08.002
- Jeon, I. S., Park, J., Eom, D., Hwang, C. S., Kim, H. J., Park, C. J., ... Kang, H. K. (2003). Post-annealing effects on fixed charge and slow/fast interface states of TiN/Al₂O₃/p-Si metal-oxide-semiconductor capacitor. *Japanese Journal of Applied Physics, Part 1: Regular Papers and Short Notes and Review Papers*, 42(3), 1222–1226. doi:10.1143/JJAP.42.1222/XML
- Jeon, S., & Hwang, H. (2003). Effect of hygroscopic nature on the electrical characteristics of lanthanide oxides (Pr₂O₃, Sm₂O₃, Gd₂O₃, and Dy₂O₃). *Journal of Applied Physics*, *93*(101), 6393–6395. doi:10.1063/1.1569028
- Jeon, S., Im, K., Yang, H., Lee, H., Sim, H., Choi, S., ... Hwang, H. (2001). Excellent electrical characteristics of lanthanide (Pr, Nd, Sm, Gd, and Dy) oxide and lanthanide-doped oxide for MOS gate dielectric applications. *International Electron Devices Meeting. Technical Digest (Cat. No.01CH37224)*, 20.6.1-20.6.4. doi:10.1109/iedm.2001.979545
- Ji, F., Xu, J. P., Lai, P. T., Li, C. X., & Liu, J. G. (2011). Improved interfacial properties of Ge MOS capacitor with high-k dielectric by using TaON/GeON dual interlayer. *IEEE Electron Device Letters*, *32*(2), 122–124. doi:10.1109/LED.2010.2092749
- Jiang, S., Liu, J., Lin, C., Li, X., & Li, Y. (2013). High-pressure x-ray diffraction and Raman spectroscopy of phase transitions in Sm₂O₃. *Journal of Applied Physics*, 113(11), 113502. doi:10.1063/1.4795504
- Jo, S. J., Ha, J. S., Park, N. K., Kang, D. K., & Kim, B. H. (2006). 5 nm thick lanthanum oxide thin films grown on Si(100) by atomic layer deposition: The effect of postannealing on the electrical properties. *Thin Solid Films*, 513(1–2), 253–257. doi:10.1016/j.tsf.2006.01.008

- Jun, J. H., Jun, J.-H., Wang, C.-H., Won, D.-J., & Choi, D.-J. (2002). Structural and Electrical Properties of a La₂O₃ Thin Film as a Gate Dielectric. *Journal of the Korean Physical Society*, 41(6), 998–1002.
- Jung, H. S., Kim, H. K., Yu, I. H., Lee, S. Y., Lee, J., Park, Jinho., ..., & Hwang, C. S. (2012). Properties of Atomic Layer Deposited HfO₂ Films on Ge Substrates Depending on Process Temperatures. *Journal of The Electrochemical Society*, 159(4), G33–G39. doi:10.1149/2.014204jes
- Kakushima, K., Tsutsui, K., Ohmi, S. I., Ahmet, P., Rao, V. R., & Iwai, H. (2006). Rare earth oxides in microelectronics. In M. Fanciulli & G. Scarel (Eds.), *Rare Earth Oxide Thin Films* (pp. 345–365). Berlin Heidelberg: Springer Verlag. doi:10.1007/11499893 20
- Kamata, Y. (2008). High-k/Ge MOSFETs for future nanoelectronics. *Materials Today*, *11*(1–2), 30–38. doi:10.1016/S1369-7021(07)70350-4
- Kamata, Y., Kamimuta, Y., Ino, T., & Nishiyama, A. (2005). Direct comparison of ZrO2 and HfO₂ on Ge substrate in terms of the realization of ultrathin high-κ gate stacks. *Japanese Journal of Applied Physics*, 44(4S), 2323–2329. doi:10.1143/JJAP.44.2323
- Kanashima, T., Nohira, H., Zenitaka, M., Kajihara, Y., Yamada, S., & Hamaya, K. (2015). Effect of atomic-arrangement matching on La₂O₃/Ge heterostructures for epitaxial high- k -gate-stacks. *Journal of Applied Physics*, 118(22), 1–6. doi:10.1063/1.4937147
- Kartopu, G., Bayliss, S. C., Karavanskii, V. A., Curry, R. J., Turan, R., & Sapelkin, A. V. (2003). On the origin of the 2.2-2.3 eV photoluminescence from chemically etched germanium. *Journal of Luminescence*, 101(4), 275–283. doi:10.1016/S0022-2313(02)00570-7
- Kaya, S., & Yilmaz, E. (2015). A comprehensive study on the frequency-dependent electrical characteristics of Sm₂O₃ MOS capacitors. *IEEE Transactions on Electron Devices*, 62(3), 980–987. doi:10.1109/TED.2015.2389953
- Kaya, S., & Yilmaz, E. (2019). A Detailed Study on Frequency Dependent Electrical Characteristics Capacitors with Dysprosium Oxide Gate Dielectrics. *Semiconductor Science and Technology*, 35, 025002.
- Kaya, S., Yilmaz, E., Karacali, H., Cetinkaya, A. O., & Aktag, A. (2015). Samarium oxide thin films deposited by reactive sputtering: Effects of sputtering power and substrate temperature on microstructure, morphology and electrical properties. *Materials Science in Semiconductor Processing*, 33, 42–48. doi:10.1016/j.mssp.2015.01.035
- Ke, M., Takenaka, M., & Takagi, S. (2018). Impact of Atomic Layer Deposition High k Films on Slow Trap Density in Ge MOS Interfaces with GeO_x Interfacial Layers Formed by Plasma Pre-Oxidation. *IEEE Journal of the Electron Devices Society*, 6, 950–955. doi:10.1109/JEDS.2018.2822758

- Ke, M., Yu, X., Chang, C., Takenaka, M., & Takagi, S. (2016). Properties of slow traps of ALD Al₂O₃/GeOx/Ge nMOSFETs with plasma post oxidation. *Applied Physics Letters*, 109(3), 3–8. doi:10.1063/1.4958890
- Khosla, R., Schwarz, D., Funk, H. S., Guguieva, K., & Schulze, J. (2021). High-quality remote plasma enhanced atomic layer deposition of aluminum oxide thin films for nanoelectronics applications. *Solid-State Electronics*, 185, 108027. doi:10.1016/j.sse.2021.108027
- Kim, D. G., Kim, H. R., Kwon, D. S., Lim, J., Seo, H., Kim, T. K., ... Hwang, C. S. (2021). Comparison of high-k Y₂O₃/TiO2 bilayer and Y-doped TiO₂ thin films on Ge substrate. *Journal of Physics D: Applied Physics*, 54(18), 185110. doi:10.1088/1361-6463/ABDEFE
- Kim, H., Chui, C. O., Saraswat, K. C., & McIntyre, P. C. (2003). Local epitaxial growth of ZrO₂ on Ge (100) substrates by atomic layer epitaxy. *Applied Physics Letters*, 83(13), 2647–2649. doi:10.1063/1.1613031
- Kim, H. D., & Roh, Y. (2006). A study on interface layer with annealing conditions of ZrO₂/ZrSixOy high-k gate oxide. *Journal of the Korean Physical Society*, 49(9), 755–759.
- Kim, W. H., Maeng, W. J., Moon, K. J., Myoung, J. M., & Kim, H. (2010). Growth characteristics and electrical properties of La₂O₃ gate oxides grown by thermal and plasma-enhanced atomic layer deposition. *Thin Solid Films*, 519(1), 362–366. doi:10.1016/j.tsf.2010.07.108
- Kita, K., & Kamata, H. (2017). Demonstration of Large Flatband Voltage Shift by Designing Al₂O₃ /SiO₂ Laminated Structures with Multiple Interface Dipole Layers . ECS Transactions, 80(1), 379–385. doi:10.1149/08001.0379ecst
- Kita, K., Kyuno, K., & Toriumi, A. (2004). Growth mechanism difference of sputtered HfO₂ on Ge and on Si. *Applied Physics Letters*, 85(1), 52–54. doi:10.1063/1.1767607
- Kita, K., Lee, C. H., Nishimura, T., Nagashio, K., & Toriumi, A. (2009). Control of Properties of GeO2 Films and Ge/GeO2 Interfaces by the Suppression of GeO Volatilization. ECS Transactions, 19(2), 101–116. doi:10.1149/1.3122088
- Kita, K., Suzuki, S., Nomura, H., Takahashi, T., Nishimura, T., & Toriumi, A. (2008). Direct evidence of GeO volatilization from GeO₂/Ge and impact of its suppression on GeO₂/Ge metal-insulator-semiconductor characteristics. *Japanese Journal of Applied Physics*, 47(4S), 2349–2353. doi:10.1143/JJAP.47.2349
- Kita, K., & Toriumi, A. (2009). Origin of electric dipoles formed at high- k/ SiO₂ interface. *Applied Physics Letters*, 94(13), 1–4. doi:10.1063/1.3110968
- Kosola, A., Päiväsaari, J., Putkonen, M., & Niinistö, L. (2005). Neodymium oxide and neodymium aluminate thin films by atomic layer deposition. *Thin Solid Films*, 479(1–2), 152–159. doi:10.1016/j.tsf.2004.12.004

- Kouda, M., Suzuki, T., Kakushima, K., Ahmet, P., Iwai, H., & Yasuda, T. (2012). Electrical Properties of CeO₂/La₂O₃ Stacked Gate Dielectrics Fabricated by Chemical Vapor Deposition and Atomic Layer Deposition. *Japanese Journal of Applied Physics*, 51(12R), 121101. doi:10.7567/jjap.51.121101
- Kukli, K., Ritala, M., Pilvi, T., Sajavaara, T., Leskelä, M., Jones, A. C., ... Tobin, P. J. (2004). Evaluation of a praseodymium precursor for atomic layer deposition of oxide dielectric films. *Chemistry of Materials*, 16(24), 5162–5168. doi:10.1021/cm0401793
- Kumar, P. ., Pavithra, K. G., & Naushad, M. (2019). Characterization techniques for nanomaterials. In *Nanomaterials for Solar Cell Applications* (pp. 97–124). Elsevier.
- Kurniawan, T., Cheong, K. Y., Razak, K. A., Lockman, Z., & Ahmad, N. (2011a). Oxidation of sputtered Zr thin film on Si substrate. *Journal of Materials Science: Materials in Electronics*, 22(2), 143–150. doi:10.1007/s10854-010-0103-1
- Kurniawan, T., Cheong, K. Y., Razak, K. A., Lockman, Z., & Ahmad, N. (2011b). Oxidation of sputtered Zr thin film on Si substrate. *Journal of Materials Science: Materials in Electronics*, 22(2), 143–150. doi:10.1007/S10854-010-0103-1
- Kutsuki, K., Okamoto, G., Hosoi, T., Shimura, T., & Watanabe, H. (2009). Germanium oxynitride gate dielectrics formed by plasma nitridation of ultrathin thermal oxides on Ge(100). *Applied Physics Letters*, *95*(2), 93–96. doi:10.1063/1.3171938
- Lale, A., Joly, M., Mekkaoui, S., Joly, X., Scheid, E., Launay, J., & Temple-Boyer, P. (2021). Effect of thermal annealing on the dielectric, passivation and pH detection properties of aluminium oxide thin films deposited by plasma-enhanced atomic layer deposition. *Thin Solid Films*, 732, 138761. doi:10.1016/j.tsf.2021.138761
- Larkin, P. (2017). Infrared and Raman Spectroscopy: Principles and Spectral Interpretation. Elsevier.
- Lee, C. H., Nishimura, T., Nagashio, K., Kita, K., & Toriumi, A. (2011). High-electronmobility Ge/GeO₂ n-MOSFETs with two-step oxidation. *IEEE Transactions on Electron Devices*, 58(5), 1295–1301. doi:10.1109/TED.2011.2111373
- Lee, J. S., Kim, W. H., Oh, I. K., Kim, M. K., Lee, G., Lee, C. W., ... Kim, H. (2014). Atomic layer deposition of Y₂O₃ and yttrium-doped HfO₂ using a newly synthesized Y(iPrCp)2(N-iPr-amd) precursor for a high permittivity gate dielectric. *Applied Surface Science*, 297, 16–21. doi:10.1016/J.APSUSC.2014.01.032
- Lei, Z. C., Goh, K. H., Zainal Abidin, N. I., & Wong, Y. H. (2017). Effect of oxidation temperature on physical and electrical properties of ZrO₂ thin-film gate oxide on Ge substrate. *Thin Solid Films*, 642, 352–358. doi:10.1016/j.tsf.2017.10.008
- Lei, Z. C., Zainal Abidin, N. I., & Wong, Y. H. (2018). Structural, chemical, and electrical properties of ZrO₂/Ge system formed via oxidation/nitridation in N2O gas ambient. *Journal of Materials Science: Materials in Electronics*, 29(15), 12888–12898. doi:10.1007/s10854-018-9408-2

- Leskela, M., Kukli, K., & Ritala, M. (2006). Rare-earth oxide thin films for gate dielectrics in microelectronics. *Journal of Alloys and Compounds*, 418(1–2), 27–34. doi:10.1016/J.JALLCOM.2005.10.061
- Leskela, M., & Ritala, M. (2003). Rare-earth oxide thin films as gate oxides in MOSFET transistors. *Journal of Solid State Chemistry*, 171(1–2), 170–174. doi:10.1016/S0022-4596(02)00204-9
- Li, Q. L., Xie, Q., Jiang, Y. L., Ru, G. P., Qu, X. P., Li, B. Z., ... Detavernier, C. (2011). Annealing induced hysteresis suppression for TiN/HfO₂/GeON/p-Ge capacitor. *Semiconductor Science and Technology*, 26(12), 125003. doi:10.1088/0268-1242/26/12/125003
- Li, S., Wu, Y., Li, G., Yu, H., Fu, K., Wu, Y., ... Li, X. (2019). Ta-doped modified Gd₂O₃ film for a novel high k gate dielectric. *Journal of Materials Science and Technology*, *35*(10), 2305–2311. doi:10.1016/j.jmst.2019.05.028
- Li, Shuan, Lin, Y., Li, G., Yu, H., Tang, S., ..., & Tian, W. (2020). Improved dielectric properties of La₂O₃–ZrO₂ bilayer films for novel gate dielectrics. *Vacuum*, 178, 109448. doi:10.1016/j.vacuum.2020.109448
- Li, Shuan, Wang, W., Lin, Y., Wang, L., & Li, X. (2022). Significantly improved high k dielectric performance: Rare earth oxide as a passivation layer laminated with TiO₂ film. *Journal of Rare Earths*, *178*, 109448. doi:10.1016/J.JRE.2022.05.005
- Liang, S., He, G., Zhu, L., Zheng, C. Y., Gao, J., Wang, D., ... Liu, M. (2018). Modulation of the microstructure, optical, and electrical properties of HfYO gate dielectrics by annealing temperature. *Journal of Alloys and Compounds*, 735, 1427–1434. doi:10.1016/j.jallcom.2017.11.261
- Lignie, A., Hermet, P., Fraysse, G., & Armand, P. (2015). Raman study of α -quartz-type Ge1-xSixO2 (0 < x \leq 0.067) single crystals for piezoelectric applications. *RSC Advances*, 5(69), 55795–55800. doi:10.1039/c5ra08051g
- Lin, C. C., Wu, Y. H., Wu, C. Y., & Lee, C. W. (2014). Surface passivation of Ge MOS devices by SmGeO_x with sub-nm EOT. *IEEE Electron Device Letters*, 35(3), 384– 386. doi:10.1109/LED.2014.2298871
- Lin, L., Xiong, K., & Robertson, J. (2010). Atomic structure, electronic structure, and band offsets at Ge:GeO: GeO₂ interfaces. *Applied Physics Letters*, 97(24), 2008–2011. doi:10.1063/1.3525371
- Liu, Q., Fang, Z., Liu, S., Tan, Y., & Chen, J. (2014). Band offsets of La₂O₃ films on Ge substrates grown by radio frequency magnetron sputtering. *Materials Letters*, 116, 43–45. doi:10.1016/j.matlet.2013.10.077
- Liu, W., He, G., Wang, D., Yu, H., Gao, Q., Liu, Y., & Fang, Z. (2022). Interface Chemistry Evolution and Leakage Current Conduction Mechanism Determination in Rare-Earth-Doped Hf-Based Gate Dielectrics. *IEEE Transactions on Electron Devices*, 69(1), 242–247. doi:10.1109/TED.2021.3130501

- Lo, S. H., Buchanan, D. A., Taur, Y., & Wang, W. (1997). Quantum-mechanical modeling of electron tunneling current from the inversion layer of ultra-thin-oxide nMOSFET's. *IEEE Electron Device Letters*, 18(5), 209–211. doi:10.1109/55.568766
- Loo, Y. F., O'Kane, R., Jones, A. C., Aspinall, H. C., Potter, R. J., Chalker, P. R., & Smith, L. M. (2005). Deposition of HfO₂ and ZrO₂ films by liquid injection MOCVD using new monomeric alkoxide precursors. *Journal of Materials Chemistry*, 15(19), 1896–1902. doi:doi.org/10.1039/B417389A
- Lu, C., Lee, C. H., Zhang, W., Nishimura, T., Nagashio, K., & Toriumi, A. (2014). Structural and thermodynamic consideration of metal oxide doped GeO₂ for gate stack formation on germanium. *Journal of Applied Physics*, *116*(17), 174103. doi:10.1063/1.4901205
- Ma, C. Y., & Zhang, Q. Y. (2008). Interfacial layer growth of ZrO₂ films on silicon. *Vaccum*, 82(8), 847–851.
- Mack, C. (2011). Fifty years of Moore's law. *IEEE Transactions on Semiconductor* Manufacturing, 24(2), 202–207. doi:10.1109/TSM.2010.2096437
- Maeda, T., Nishizawa, M., Morita, Y., & Takagi, S. (2007). Role of germanium nitride interfacial layers in HfO₂/germanium nitride/germanium metal-insulatorsemiconductor structures. *Applied Physics Letters*, 90(7), 072911. doi:10.1063/1.2679941
- Maeda, T., Yasuda, T., Nishizawa, M., Miyata, N., Morita, Y., & Takagi, S. (2004). Ge metal-insulator-semiconductor structures with Ge₃N₄ dielectrics by direct nitridation of Ge substrates. *Applied Physics Letters*, 85(15), 3181–3183. doi:10.1063/1.1805194
- Maeda, T., Yasuda, T., Nishizawa, M., Miyata, N., Morita, Y., & Takagi, S. (2006). Pure germanium nitride formation by atomic nitrogen radicals for application to Ge metal-insulator-semiconductor structures. *Journal of Applied Physics*, 100(1), 014101. doi:10.1063/1.2206395
- Maeng, W. J., Oh, I. K., Kim, W. H., Kim, M. K., Lee, C. W., Lansalot-Matras, C., ... Kim, H. (2014). Atomic layer deposition of CeO₂/HfO₂ gate dielectrics on Ge substrate. *Applied Surface Science*, 321, 214–218. doi:10.1016/j.apsusc.2014.10.025
- Mallem, K., Jagadeesh Chandra, S. V., Ju, M., Dutta, S., Phanchanan, S., Sanyal, S., ... Yi, J. (2019). Effects of post deposition annealing atmosphere on interfacial and electrical properties of HfO₂ /Ge₃N₄ gate stacks. *Thin Solid Films*, 675, 16–22. doi:10.1016/j.tsf.2019.02.034
- Mandal, J., Sarkar, B. J., Deb, A. K., & Chakrabarti, P. K. (2014). Magnetic phase transition of nanocrystalline Fe-doped samarium oxide (Sm1.90Fe0.10O3). Journal of Magnetism and Magnetic Materials, 371, 35–42. doi:10.1016/j.jmmm.2014.06.068

- Matsubara, H., Sasada, T., Takenaka, M., & Takagi, S. (2008). Evidence of low interface trap density in GeO₂ Ge metal-oxide- semiconductor structures fabricated by thermal oxidation. *Applied Physics Letters*, *93*(3), 1–4. doi:10.1063/1.2959731
- Mavrou, G., Galata, S. F., Sotiropoulos, A., Tsipas, P., Panayiotatos, Y., Dimoulas, A., ... Dieker, C. (2007). Germanium metal-insulator-semiconductor capacitors with rare earth La₂O₃ gate dielectric. *Microelectronic Engineering*, 84(9–10), 2324–2327. doi:10.1016/j.mee.2007.04.036
- Mavrou, G., Galata, S., Tsipas, P., Sotiropoulos, A., Panayiotatos, Y., Dimoulas, A., ... Dieker, C. (2008). Electrical properties of La₂O₃ and HfO₂ /La₂O₃ gate dielectrics for germanium metal-oxide-semiconductor devices. *Journal of Applied Physics*, 103(1), 014506. doi:10.1063/1.2827499
- Mavrou, G., Tsipas, P., Sotiropoulos, A., Galata, S., Panayiotatos, Y., Dimoulas, A., ... Fompeyrine, J. (2008). Very high- κ ZrO₂ with La₂O₃ (LaGeO_x) passivating interfacial layers on germanium substrates. *Applied Physics Letters*, 93(21), 23–26. doi:10.1063/1.3033546
- Minoura, Y., Kasuya, A., Hosoi, T., Shimura, T., Watanabe, H., Minoura, Y., ... Watanabe, H. (2013). Design and control of Ge-based metal-oxide-semiconductor interfaces for high-mobility field-effect transistors with ultrathin oxynitride gate dielectrics. *Applied Physics Express*, 103(3), 033502. doi:10.1063/1.4813829
- Misra, D., Garg, R., Srinivasan, P., Rahim, N., & Chowdhury, N. A. (2006). Interface characterization of high-k dielectrics on Ge substrates. *Materials Science in Semiconductor Processing*, 9(4–5), 741–748. doi:10.1016/j.mssp.2006.08.028
- Mitrovic, I. Z., Althobaiti, M., Weerakkody, A. D., Dhanak, V. R., Linhart, W. M., Veal, T. D., ... Dimoulas, A. (2014). Ge interface engineering using ultra-thin La₂O₃ and Y2O3 films: A study into the effect of deposition temperature. *Journal of Applied Physics*, 115(11), 114102. doi:10.1063/1.4868091
- Muller, D. A., Sorsch, T., Moccio, S., Baumann, F. H., Evans-Lutterodt, K., & Timp, G. (1999). The electronic structure at the atomic scale of ultrathin gate oxides. *Nature*, *399*(6738), 758–761. doi:10.1038/21602
- Nagabharana, R., Kumaraswamy, G., Gundanna, Susheel, K., & Umananda, M. (2020). Effect of thermal annealing on structural and electrical properties of tio₂ thin films. *Thin Solid Films*, *710*, 138262. doi:10.1016/J.TSF.2020.138262
- Naumann, V., Otto, M., Wehrspohn, R. B., Werner, M., & Hagendorf, C. (2012). Interface and Material Characterization of Thin ALD-Al₂O₃ Layers on Crystalline Silicon. *Energy Procedia*, 27, 312–318. doi:doi.org/10.1016/j.egypro.2012.07.069
- Nickel, N. H., Friedrich, F., Rommelure, J. F., & Galtier, P. (2005). Vibrational spectroscopy of undoped and nitrogen-doped ZnO grown by metalorganic chemical vapor deposition. *Applied Physics Letters*, 87(21), 1–3. doi:10.1063/1.2133917
- Nishimura, T., Lee, C. H., Tabata, T., Wang, S. K., Nagashio, K., Kita, K., & Toriumi, A. (2011). High-electron-mobility Ge n-channel metal-oxide-semiconductor fieldeffect transistors with high-pressure oxidized Y₂O₃. *Applied Physics Express*, 4(6),

4-7. doi:10.1143/APEX.4.064201

- Oh, I. K., Kim, K., Lee, Z., Song, J. G., Lee, C. W., Thompson, D., ... Kim, H. (2015). In situ surface cleaning on a Ge substrate using TMA and MgCp₂ for HfO₂-based gate oxides. *Journal of Materials Chemistry C*, 3(19), 4852–4858. doi:10.1039/c4tc02686a
- Ohmi, S., Aha, S., Kikuchi, A., Kashiwagi, I., Ohshima, C., Taguchi, J., ... Iwai, H. (2001). Rare earth metal oxide gate thin films prepared by e-beam deposition. *Extended Abstracts of International Workshop on Gate Insulator. IWGI 2001 (IEEE Cat. No.01EX537)*, 200–204. IEEE. doi:10.1109/IWGI.2001.967583
- Osten, H. J., Laha, A., Czernohorsky, M., Bugiel, E., Dargis, R., & Fissel, A. (2008). Introducing crystalline rare-earth oxides into si technologies. *Physica Status Solidi* (A) Applications and Materials Science, 205(4), 695–707. doi:10.1002/pssa.200723509
- Otani, Y., Itayama, Y., Tanaka, T., Fukuda, Y., Toyota, H., Ono, T., ... Nakagawa, K. (2007). Fabrication of Ta₂O₅ GeN_x gate insulator stack for Ge metal-insulator-semiconductor structures by electron-cyclotron-resonance plasma nitridation and sputtering deposition techniques. *Applied Physics Letters*, 90(14), 1–4. doi:10.1063/1.2720345
- Paivasaari, J., Niinisto, J., Myllymaki, P., Dezelah IV, C., Winter, C. H., Putkonen, M., ... Niinisto, L. (2006). Atomic layer deposition of rare earth oxides. In M. Fanciulli & G. Scarel (Eds.), *Rare Earth Oxide Thin Films* (pp. 15–32). Berlin Heidelberg: Springer Verlag. doi:10.1007/11499893_2
- Paivasaari, J., Putkonen, M., & Niinisto, L. (2005). A comparative study on lanthanide oxide thin films grown by atomic layer deposition. *Thin Solid Films*, 472(1–2), 275– 281. doi:10.1016/j.tsf.2004.06.160
- Pampillon, M. A., Feijoo, P. C., San Andres, E., Lucía, M. L., Del Prado, A., & Toledano-Luque, M. (2011). Anomalous thermal oxidation of gadolinium thin films deposited on silicon by high pressure sputtering. *Microelectronic Engineering*, 88(9), 2991– 2996. doi:10.1016/j.mee.2011.04.058
- Pan, T. M., Chang, W. T., & Chiu, F. C. (2010). Structural and electrical properties of thin Ho₂O₃ gate dielectrics. *Thin Solid Films*, 519(2), 923–927. doi:10.1016/j.tsf.2010.09.002
- Pan, T. M., & Huang, C. C. (2010). Effects of oxygen content and postdeposition annealing on the physical and electrical properties of thin Sm₂O₃ gate dielectrics. *Applied Surface Science*, 256(23), 7186–7193. doi:10.1016/j.apsusc.2010.05.048
- Pan, T. M., Huang, C. C., You, S. X., & Yeh, C. C. (2008). Effect of annealing on the structural and electrical properties of high- k Sm₂O₃ dielectrics. *Electrochemical and Solid-State Letters*, 11(12), 62–65. doi:10.1149/1.2990226
- Pan, T. M., & Huang, M. De. (2011). Structural properties and sensing characteristics of high-k Ho₂O₃ sensing film-based electrolyte-insulator-semiconductor. *Materials Chemistry and Physics*, 129(3), 919–924. doi:10.1016/j.matchemphys.2011.05.032

- Pan, T. M., Huang, M. De, Lin, C. W., & Wu, M. H. (2010). Development of high-κ HoTiO₃ sensing membrane for pH detection and glucose biosensing. *Sensors and Actuators, B: Chemical*, 144(1), 139–145. doi:10.1016/j.snb.2009.10.049
- Pan, T. M., & Lee, J. D. (2007). Physical and Electrical Properties of Yttrium Oxide Gate Dielectrics on Si Substrate with NH₃ Plasma Treatment. *Journal of The Electrochemical Society*, 154(8), H698. doi:10.1149/1.2742808/META
- Pan, T. M., Yen, L. C., Hu, C. W., & Sheng, C. T. (2010). Structural and Electrical Properties of High-k HoTiO₃ Gate Dielectrics. *ECS Transactions*, 28(1), 241–245. doi:10.1149/1.3375607
- Park, D. G., & Kim, T. K. (2005). Effects of fluorine and chlorine on the gate oxide integrity of W/TiN/SiO₂/Si metal-oxide-semiconductor structure. *Thin Solid Films*, 483(1–2), 232–238. doi:10.1016/j.tsf.2004.12.003
- Prabhakaran, K., Maeda, F., Watanabe, Y., & Ogino, T. (2000). Thermal decomposition pathway of Ge and Si oxides: Observation of a distinct difference. *Thin Solid Films*, *369*(1), 289–292. doi:10.1016/S0040-6090(00)00881-6
- Quah, H. J., & Cheong, K. Y. (2015). Effects of annealing time on the electrical properties of the Y₂O₃ gate on silicon. *Journal of Experimental Nanoscience*, *10*(1), 19–28. doi:10.1080/17458080.2013.781689
- Radamson, H. & Thylen, L. (2014). Monolithic Nanoscale Photonics-Electronics Integration in Silicon and Other Group IV Elements. Academic Press.
- Rahman, M. M., Kim, J. G., Kim, D. H., & Kim, T. W. (2019). Characterization of Al incorporation into HfO2 dielectric by atomic layer deposition. *Micromachines*, 10(6), 1–11. doi:10.3390/mi10060361
- Ramana, C. V., Vemuri, R. S., Kaichev, V. V., Kochubey, V. A., Saraev, A. A., & Atuchin, V. V. (2011). X-ray photoelectron spectroscopy depth profiling of La ₂O ₃/Si thin films deposited by reactive magnetron sputtering. ACS Applied Materials and Interfaces, 3(11), 4370–4373. doi:10.1021/AM201021M
- Rao, A., Verma, A., & Singh, B. R. (2015). Ionizing radiation effects on electrical and reliability characteristics of sputtered Ta₂O₅/Si interface. *Radiation Effects and Defects in Solids*, 170(6), 510–518. doi:10.1080/10420150.2015.1052433
- Robertson, J. (2002). Band offsets of high dielectric constant gate oxides on silicon. *Journal of Non-Crystalline Solids*, 303(1), 94–100. doi:10.1016/S0022-3093(02)00972-9
- Robertson, J. (2004). High dielectric constant oxides. *The European Physical Journal of Applied Physics*, 28, 265–291. doi:10.1051/epjap:2004206
- Robertson, J. (2006). High dielectric constant gate oxides for metal oxide Si transistors. *Reports on Progress in Physics*, 69(2), 327.
- Robertson, J., & Wallace, R. M. (2015). High-K materials and metal gates for CMOS applications. *Materials Science and Engineering R: Reports*, 88, 1–41. doi:10.1016/j.mser.2014.11.001
- Rolim, G. K., Gobbi, A., Soares, G. V., & Radtke, C. (2015). Oxygen transport and incorporation in Pt/HfO₂ stacks deposited on germanium and silicon. *Journal of Physical Chemistry C*, 119(8), 4079–4084. doi:10.1021/JP511127C
- Sahari, S. K., Ohta, A., Matsui, M., Mishima, K., Murakami, H., Higashi, S., & Miyazaki, S. (2013). Kinetics of thermally oxidation of Ge(100) surface. *Journal of Physics: Conference Series*, 417(1). doi:10.1088/1742-6596/417/1/012014
- Saraswat, K. C., Chui, C. O., Krishnamohan, T., Nayfeh, A., & McIntyre, P. (2005). Ge based high performance nanoscale MOSFETs. *Microelectronic Engineering*, 80, 15–21. doi:10.1016/j.mee.2005.04.038
- Sasada, T., Nakakita, Y., Takenaka, M., & Takagi, S. (2009). Surface orientation dependence of interface properties of GeO₂/Ge metal-oxide-semiconductor structures fabricated by thermal oxidation. *Journal of Applied Physics*, 106(7), 073716. doi:10.1063/1.3234395
- Scarel, G., Svane, A., & Fanciulli, M. (2006). Scientific and technological issues related to rare earth oxides: An introduction. In M. Fanciulli & G. Scarel (Eds.), *Rare Earth Oxide Thin Films* (pp. 1–14). Berlin, Heidelberg: Springer Verlag. doi:10.1007/11499893_1
- Schamm, S., Coulon, P. E., Miao, S., Volkos, S. N., Lu, L. H., Lamagna, L., ... Fanciulli, M. (2009). Chemical/Structural Nanocharacterization and Electrical Properties of ALD-Grown La₂O₃/Si Interfaces for Advanced Gate Stacks. *Journal of The Electrochemical Society*, 156(1), H1. doi:10.1149/1.3000594/META
- Schroder, D. K. (2015). Semiconductor Material and Device Characterization. John Wiley & Sons.
- Sekhar, M., Nanda Kumar Reddy, N., Venkata Rao, B., Mohan Rao, G., & Uthanna, S. (2014). Influence of sputter power on structural and electrical properties of TiO₂ films for Al/TiO₂/Si gate capacitors. *Surface and Interface Analysis*, 46(7), 465–471. doi:10.1002/SIA.5538
- Sen, B., Wong, H., Molina, J., Iwai, H., Ng, J. A., Kakushima, K., & Sarkar, C. K. (2007). Trapping characteristics of lanthanum oxide gate dielectric film explored from temperature dependent current-voltage and capacitance-voltage measurements. *Solid-State Electronics*, 51(3), 475–480. doi:10.1016/j.sse.2007.01.032
- Seo, Y., Lee, T. I., Yoon, C. M., Park, B. E., Hwang, W. S., Kim, H., ... Cho, B. J. (2017). The Impact of an Ultrathin Y₂O₃ Layer on GeO₂ Passivation in Ge MOS Gate Stacks. *IEEE Transactions on Electron Devices*, 64(8), 3303–3307. doi:10.1109/TED.2017.2710182

- Shao, Q. Y., Li, A. D., Ling, H. Q., Wu, D., Wang, Y., Feng, Y., ... Ming, N. Ben. (2003). Growth and characterization of Al₂O₃ gate dielectric films by low-pressure metalorganic chemical vapor deposition. *Microelectronic Engineering*, 66(1–4), 842–848. doi:10.1016/S0167-9317(02)01009-2
- Sharma, R. K., Kumar, A., & Anthony, J. M. (2001). Advances in high-k dielectric gate materials for future ULSI devices. JOM, 53(6), 53–55. doi:10.1007/S11837-001-0105-9
- Sicard, E., & Aziz, S. (2011). Introducing 65 nm technology in Microwind3. Hal-033243.
- Simoen, E., Mitard, J., Hellings, G., Eneman, G., De Jaeger, B., Witters, L., ... Claeys, C. (2012). Challenges and opportunities in advanced Ge pMOSFETs. *Materials Science in Semiconductor Processing*, 15(6), 588–600. doi:10.1016/j.mssp.2012.04.017
- Song, J., Kakushima, K., Ahmet, P., Tsutsui, K., Sugii, N., Hattori, T., & Iwai, H. (2007). Characteristics of ultrathin lanthanum oxide films on germanium substrate: Comparison with those on silicon substrate. *Japanese Journal of Applied Physics*, *Part 2: Letters*, 46(12–16), 4–7. doi:10.1143/JJAP.46.L376
- Song, J., Kakushima, K., Ahmet, P., Tsutsui, K., Sugii, N., Hattori, T., & Iwai, H. (2009). Post metallization annealing study in La₂O₃/Ge MOS structure. *Microelectronic Engineering*, 86(7–9), 1638–1641. doi:10.1016/j.mee.2009.03.051
- Song, J. Q., Qian, L. X., & Lai, P. T. (2018). Improved Performance of Amorphous InGaZnO Thin-Film Transistor by Hf Incorporation in La₂O₃ Gate Dielectric. *IEEE Transactions on Device and Materials Reliability*, 18(3), 333–336. doi:10.1109/TDMR.2018.2840881
- Spassov, D., Atanassova, E., & Virovska, D. (2006). Electrical characteristics of Ta₂O₅ based capacitors with different gate electrodes. *Applied Physics A: Materials Science and Processing*, *82*, 55–62. doi:10.1007/s00339-005-3300-7
- Srivastava, A., Nahar, R. K., & Sarkar, C. K. (2011). Study of the effect of thermal annealing on high k hafnium oxide thin film structure and electrical properties of MOS and MIM devices. *Journal of Materials Science: Materials in Electronics*, 22(7), 882–889. doi:10.1007/S10854-010-0230-8
- Tan, S. Y. (2010). Control of interface traps in HfO₂ gate dielectric on silicon. *Journal of Electronic Materials*, *39*(11), 2435–2440. doi:10.1007/S11664-010-1323-0
- Taoka, N., Mizubayashi, W., Morita, Y., Migita, S., Ota, H., & Takagi, S. (2011). Nature of interface traps in Ge metal-insulator-semiconductor structures with GeO₂ interfacial layers. *Journal of Applied Physics*, 109(8), 084508. doi:10.1063/1.3575332
- Taur, Y. (2002). CMOS design near the limit of scaling. *Ibm Journal of Research and Development*, 46(2/3), 213–222. doi:10.1147/rd.462.0213

- Thompson, S. E., & Parthasarathy, S. (2006). Moore's law: the future of Si microelectronics. *Materials Today*, 9(6), 20–25. doi:10.1016/S1369-7021(06)71539-5
- Tirmali, P., Khairnar, A., Joshi, B., Electronics, A. M.-S., & 2011, U. (2011). Structural and electrical characteristics of RF-sputtered HfO₂ high-k based MOS capacitors. *Solid-State Electronics*, *62*(1), 44–47.
- Venkata Rao, G., Kumar, M., Rajesh, T. V., Rama Koti Reddy, D. V., Anjaneyulu, D., Sainath, B., & Jagadeesh Chandra, S. V. (2018). Investigations on the Nitride Interface Engineering at HfO₂/Ge stacks for MOS devices. *Materials Today: Proceedings*, 5(1), 650–656. doi:10.1016/j.matpr.2017.11.129
- Wang, D., He, G., Fang, Z., Hao, L., Sun, Z., & Liu, Y. (2019). Interface chemistry modulation and dielectric optimization of TMA-passivated HfDyO: X/Ge gate stacks using doping concentration and thermal treatment. *RSC Advances*, 10(2), 938–951. doi:10.1039/c9ra08335a
- Wang, D., He, G., Hao, L., Gao, J., & Zhang, M. (2019). Comparative passivation effect of ALD-driven HfO₂ and Al₂O₃ buffer layers on the interface chemistry and electrical characteristics of Dy-based gate dielectrics. *Journal of Materials Chemistry C*, 7(7), 1955–1965. doi:10.1039/c8tc05736b
- Wang, H., Chroneos, A., Dimoulas, A., & Schwingenschlögl, U. (2012). Interaction of oxygen vacancies in yttrium germanates. *Physical Chemistry Chemical Physics*, 14(42), 14630–14634. doi:10.1039/c2cp42380d
- Wang, S. J., Huan, A. C. H., Foo, Y. L., Chai, J. W., Pan, J. S., Li, Q., ... Ong, C. K. (2004). Energy-band alignments at ZrO₂/Si, SiGe, and Ge interfaces. *Applied Physics Letters*, 85(19), 4418–4420. doi:10.1063/1.1819988
- Wang, S. K., Kita, K., Lee, C. H., Tabata, T., Nishimura, T., Nagashio, K., & Toriumi, A. (2010). Desorption kinetics of GeO from GeO₂ /Ge structure. *Journal of Applied Physics*, 108(5), 054104. doi:10.1063/1.3475990
- Wang, S. K., Liu, H. G., & Toriumi, A. (2012). Kinetic study of GeO disproportionation into a GeO₂/Ge system using x-ray photoelectron spectroscopy. *Applied Physics Letters*, 101(6), 2–7. doi:10.1063/1.4738892
- Wang, X., Nishimura, T., Yajima, T., & Toriumi, A. (2017). Thermal oxidation kinetics of germanium. *Applied Physics Letters*, 111(5), 052101. doi:10.1063/1.4997298
- Watanabe, H., Kutsuki, K., Kasuya, A., Hideshima, I., Okamoto, G., Saito, S., ... Shimura, T. (2012). Gate stack technology for advanced high-mobility Ge-channel metal-oxide-semiconductor devices - Fundamental aspects of germanium oxides and application of plasma nitridation technique for fabrication of scalable oxynitride dielectrics. *Current Applied Physics*, 12, S10–S19. doi:10.1016/j.cap.2012.04.025
- Weinreich, W., Ignatova, V. A., Wilde, L., Teichert, S., Lemberger, M., Bauer, A. J., ... Schröder, U. (2009). Influence of N₂ and NH₃ annealing on the nitrogen incorporation and k -value of thin ZrO₂ layers. *Journal of Applied Physics*, 106(3). doi:10.1063/1.3187829

- Wilk, G. D., Wallace, R. M., & Anthony, J. M. (2001). High-κ gate dielectrics: Current status and materials properties considerations. *Journal of Applied Physics*, 89(10), 5243–5275. doi:10.1063/1.1361065
- Wong, Y. H., & Cheong, K. Y. (2010). ZrO₂ thin films on Si substrate. Journal of Materials Science: Materials in Electronics, 21(10), 980–993. doi:10.1007/s10854-010-0144-5
- Wong, Y. H., & Cheong, K. Y. (2011a). Band alignment and enhanced breakdown field of simultaneously oxidized and nitrided Zr film on Si. *Nanoscale Research Letters*, 6(1), 1–5. doi:10.1186/1556-276x-6-489
- Wong, Y. H., & Cheong, K. Y. (2011b). Electrical Characteristics of Oxidized/Nitrided Zr Thin Film on Si. *Journal of The Electrochemical Society*, 158(12), H1270-78. doi:10.1149/2.106112jes
- Wong, Y. H., & Cheong, K. Y. (2011c). Thermal oxidation and nitridation of sputtered Zr thin film on Si via N2O gas. *Journal of Alloys and Compounds*, 509(35), 8728– 8737. doi:10.1016/j.jallcom.2011.06.041
- Wong, Y. H., & Cheong, K. Y. (2012a). Formation of Zr-oxynitride thin films on 4H-SiC substrate. *Thin Solid Films*, 520(22), 6822–6829. doi:10.1016/j.tsf.2012.07.036
- Wong, Y. H., & Cheong, K. Y. (2012b). Metal-Oxide-Semiconductor Characteristics of Zr-Oxynitride Thin Film on 4H-SiC Substrate. *Journal of The Electrochemical Society*, 159(3), H293–H299. doi:10.1149/2.081203jes
- Wong, Y. H., & Cheong, K. Y. (2012c). Properties of thermally oxidized and nitrided Zroxynitride thin film on 4H-SiC in diluted N₂O ambient. *Materials Chemistry and Physics*, 136(2–3), 624–637. doi:10.1016/j.matchemphys.2012.07.035
- Wong, Y. H., Lei, Z. C., & Abidin, N. I. Z. (2021). Surface and interface characteristics of annealed ZrO₂/Ge oxide-semiconductor structure in argon ambient. *Surfaces and Interfaces*, 23, 101007. doi:10.1016/J.SURFIN.2021.101007
- Xiang, J., Li, T., Yu, J., Wang, X., Ma, X., Gao, J., ... Wang, W. (2020). Influence of interlayer GeOx thickness on band alignment of Al₂O₃/GeO_x/Ge structure. *Materials Science in Semiconductor Processing*, 113, 105051. doi:10.1016/J.MSSP.2020.105051
- Xie, Q., Deng, S., Schaekers, M., Lin, D., Caymax, M., Delabie, A., ... Detavernier, C. (2012). Germanium surface passivation and atomic layer deposition of high-k dielectrics - A tutorial review on Ge-based MOS capacitors. *Semiconductor Science* and Technology, 27(7). doi:10.1088/0268-1242/27/7/074012
- Yang, C., Fan, H., Qiu, S., Xi, Y., & Fu, Y. (2009). Microstructure and dielectric properties of La₂O₃ films prepared by ion beam assistant electron-beam evaporation. *Journal of Non-Crystalline Solids*, 355(1), 33–37. doi:10.1016/j.jnoncrysol.2008.09.029

- Yang, D., Xue, L. J., & Devine, R. A. B. (2003). Charge trapping in and electrical properties of pulsed laser deposited Sm₂O₃ films. *Journal of Applied Physics*, 93(11), 9389–9391. doi:10.1063/1.1569660
- Yang, X., Wang, S. K., Zhang, X., Sun, B., Zhao, W., Chang, H. D., ... Liu, H. (2014). Al₂O₃/GeO_x gate stack on germanium substrate fabricated by in situ cycling ozone oxidation method. *Applied Physics Letters*, 105(9), 2–6. doi:10.1063/1.4894631
- Yao, Q., Ma, X., Wang, H., Wang, Y., Wang, G., Zhang, J., ... Wang, W. (2021). Investigate on the Mechanism of HfO₂/Si_{0.7}Ge_{0.3} Interface Passivation Based on Low-Temperature Ozone Oxidation and Si-Cap Methods. *Nanomaterials*, 11(4), 955. doi:10.3390/nano11040955
- Yi, S. H., Chang-Liao, K. S., Hsu, C. W., & Huang, J. (2018). Improved Electrical Characteristics of ~ 0.5 nm EOT Ge pMOSFET with GeON Interfacial Layer Formed by NH₃ Plasma and Microwave Annealing Treatments. *IEEE Electron Device Letters*, 39(9), 1278–1281. doi:10.1109/LED.2018.2853592
- Yoshida, T., Kato, K., Shibayama, S., Sakashita, M., Taoka, N., Takeuchi, W., ... Zaima, S. (2014). Interface properties of Al₂O₃/Ge structures with thin Ge oxide interfacial layer formed by pulsed metal organic chemical vapor deposition. *Japanese Journal* of Applied Physics, 53(8), 66–71. doi:10.7567/JJAP.53.08LD03/META
- Zhang, J. W., He, G., Zhou, L., Chen, H. S., Chen, X. S., Chen, X. F., ... Sun, Z. Q. (2014). Microstructure optimization and optical and interfacial properties modulation of sputtering-derived HfO₂ thin films by TiO₂ incorporation. *Journal of Alloys and Compounds*, 611, 253–259. doi:10.1016/j.jallcom.2014.05.074
- Zhang, L., Gunji, M., Thombare, S., & McIntyre, P. C. (2013). EOT scaling of TiO₂/Al₂O₃ on germanium pMOSFETs and impact of gate metal selection. *IEEE Electron Device Letters*, 34(6), 732–734. doi:10.1109/LED.2013.2259137
- Zhang, R., Huang, P. C., Lin, J. C., Taoka, N., Takenaka, M., & Takagi, S. (2013). Highmobility Ge p- and n-MOSFETs with 0.7-nm EOT using HfO₂/Al₂O₃/GeO_x/Ge gate stacks fabricated by plasma postoxidation. *IEEE Transactions on Electron Devices*, 60(3), 927–934. doi:10.1109/TED.2013.2238942
- Zhang, R., Iwasaki, T., Taoka, N., Takenaka, M., & Takagi, S. (2011). Al₂O₃ / GeO_x /Ge gate stacks with low interface trap density fabricated by electron cyclotron resonance plasma postoxidation. *Applied Physics Letters*, *98*(11), 3–6. doi:10.1063/1.3564902
- Zhang, R., Iwasaki, T., Taoka, N., Takenaka, M., & Takagi, S. (2012). High-mobility Ge pMOSFET With 1-nm EOT Al₂O₃ GeO_x Ge gate stack fabricated by plasma post oxidation. *IEEE Transactions on Electron Devices*, 59(2), 335–341. doi:10.1109/TED.2011.2176495
- Zhao, L., Liu, H., Wang, X., Wang, Y., & Wang, S. (2018). Improvements on the interfacial properties of High-k/Ge MIS structures by inserting a La₂O₃ passivation layer. *Materials*, *11*(11), 2333. doi:10.3390/ma11112333

- Zhao, L., Liu, H., Wang, X., Wang, Y., & Wang, S. (2019). Electrical properties and interfacial issues of hfo₂/ge mis capacitors characterized by the thickness of la2o3 interlayer. *Nanomaterials*, 9(5), 697. doi:10.3390/nano9050697
- Zhu, W. J., Tamagawa, T., M., G., Furukawa, T., & Ma, T. P. (2002). Effect of Al inclusion in HfO₂ on the physical and electrical properties of the dielectrics. *IEEE Electron Device Letters*, 23(11), 649–651. doi:10.1109/LED.2002.805000
- Zimmermann, C., Bethge, O., Winkler, K., Lutzer, B., & Bertagnolli, E. (2016). Improving the ALD-grown Y₂O₃ /Ge interface quality by surface and annealing treatments. *Applied Surface Science*, 369, 377–383. doi:10.1016/j.apsusc.2016.02.066
- Žurauskaitėa, L., Jonesb, L., Dhanakc, V. R., Mitrovicb, I. Z., Hellströma, P. E., And, & Östlinga, M. (2018). Investigation of Tm₂O₃ As a Gate Dielectric for Ge MOS Devices. *Transactions, E C S Society, The Electrochemical*, 86(7), 67–73. doi:10.1149/ma2018-02/31/1020